#### DOCUMENT RESUME

ED 226 111 CE 034 530

AUTHOR
TITLE
INSTITUTION
SPONS AGENCY

Cerf, J. George; And Others Crafts-Artists in the United States.

Mathematica Policy Research, Princeton, N.J. National Endowment for the Arts, Washington, D.C.

PUB DATE Jun 82

NEA-PG-79-23

229p.; For related documents see CE 034 527-528.

PUB TYPE Reports - Research/Technical (143)

EDRS PRICE DESCRIPTORS

GRANT

NOTE

MF01/PC10 Plus Postage. Adults; \*Craft Workers; \*Enrollment; \*Group Membership; \*Handicrafts; National Surveys;

Organizations (Groups); Organization Size (Groups); \*Participant Characteristics; Policy Formation;

Public Policy

#### **ABSTRACT**

Following a preliminary study of crafts membership organizations in the United States, a survey was conducted of members of these organizations in order to determine their number, kind, media of work, personal characteristics, and geographic location. A sample of 5,146 craftspersons who were members of these organizations was mailed a questionnaire. The response rate was 74 percent with 2,637 identifying themselves as crafts-artists; the remainder were dropped from the sample. Analysis of the data gathered showed results such as the following: (1) about 375,000 member crafts-artists live in the United States; (2) the most accomplished and successful craftspersons are members of organizations; (3) crafts-artists are likely to be white females, with a median age of 45 years, married (72 percent), college graduates employed in a crafts, technical, or professional occupation, with incomes about 35 percent higher than the average; (4) fiber is the primary medium of 42 percent of the craftspersons, followed by clay, wood, and metal; (5) about 25 percent of the craftspersons consider their craft work to be full time; and (6) the majority of craftspersons are self taught, but workshops and college or university courses in crafts were also frequent avenues to the crafts. The study leads to public policy implications for training, exhibits and shows, awards and financial support, and the imbalance of supply and demand in the crafts fields.

# CRAFTS-ARTISTS IN THE UNITED STATES

Ъy

J. Georg Cerf Constance F. Citro Matthew Black Audrey McDonald

June, 1982

U.S. DEPARTMENT OF EDUCATION
NATIONAL INSTITUTE OF EDUCATION
EDUCATIONAL RESOURCES INFORMATION
CENTER (ERIC)

This document has been reproduced as received from the person or organization originating it.

Minor changes have been made to improve

Minor changes have been made to improve reproduction quality.

 Points of view or opinions stated in this document do not necessarily represent official NIE position or policy.

## Submitted to:

Division of Research National Endowment for the Arts Washington, DC

## Submitted by:

Mathematica Policy Research Princeton, New Jersey and Washington, DC

## TABLE OF CONTENTS

Chapt	er		Page
	EXE	CUTIVE SUMMARY	i
I.	INT	RODUCTION	1.1
	A.	PRIOR RESEARCH ON CRAFTS BY THE ARTS ENDOWMENT	1.2
	В.	RATIONALE FOR A SURVEY OF CRAFTS-ARTISTS WHO ARE MEMBERS OF CRAFTS ORGANIZATIONS	1.3
	c.	DATA COLLECTION PROCEDURES	1.5
	D.	MAJOR CATEGORIES USED IN TABULATIONS	I.7
	E.	ORGANIZATION OF THE TABLES USED IN THIS REPORT	1.15
II.	EST	IMATES OF THE NUMBER OF MEMBER CRAFTS-ARTISTS	II.1
	A.	ESTIMATES OF CRAFTS-ARTISTS BELONGING TO MEMBERSHIP ORGANIZATIONS IN 1980	II.2
·	В.	COMPARISON TO 1978 ESTIMATES	11.4
III.	PE	SONAL PROFILES OF CRAFTS-ARTISTS	III.1
	A.	DEMOGRAPHIC PROFILE	III.1
	В.	COMPARISON OF CRAFTS-ARTISTS TO THE U.S. POPULATION	111.10
IV.	CRAI	TS-RELATED CHARACTERISTICS	IV.1
	A.	GENERAL CRAFTS-RELATED CHARACTERISTICS	IV.1
	В.	TRAINING IN CRAFTS	IV.8
	c.	INVOLVEMENT IN CRAFTS	įv.16
	D.	ATTITUDES TOWARD CRAFTS: SATISFACTIONS, BARRIERS AND IMPORTANT PROBLEMS	IV.26
	E.	EXHIBITING CRAFTS	IV.3
	F.	SELLING CRAFTS	IV.37
	G.	FINANCIAL SUPPORT AND CRAFTS INCOME AND EXPENSES	· `IV.4
	H.	AN OVERVIEW OF CRAFTS CHARACTERISTICS	



## TABLE OF CONTENTS (continued)

Chapt	er		Page
٧.		IRICAL ANALYSIS OF CRAFTS-ARTISTS ACTIVITY FINANCIAL REWARDS	V.1
	A.	INTRODUCTION	V.1
	В.	CONCEPTUAL OVERVIEW	₹.2
	c.	EMPIRICAL FINDINGS	₹.9
	D.	CONCLUSIONS AND RECOMMENDATION	<b>v.3</b> 7
VI.	CR#J	FTS PROFESSIONALS AND SUBSCRIBERS	<b>VI.1</b>
	A.	EXTENT OF MEMBERSHIP	VI.2
	В.	COMPARISON OF CRAFTS-ARTISTS IN DIFFERENT SAMPLES	VI.4
APPEN	DICE	s	
	<b>A.</b>	QUESTIONNAIRE	A.1
	В.	COMPARISON OF RESULTS OF TELEPHONE FOLLOW-UP OF CRAFTS-ARTISTS	B.1
	c.	COMPARISON OF AMERICAN AND CANADIAN CRAFTS-ARTISTS	c.1
	D.	MODEL OF CRAFTS INVOLVEMENT	D.1

ERIC

U

#### EXECUTIVE SUMMARY

The National Endowment for the Arts, as part of a continuing program of research on crafts-artists and their environment, contracted with Mathematica Policy Research, Inc. (MPR), to conduct a survey of crafts-artists who are members of crafts organizations in the United States. The results of this recently-completed study are presented in the report that follows. The purpose of this summary is to distill the findings for the reader interested in an overview of the results and to present conclusions and implications for policy that flow out of these findings.

#### A. STUDY DESIGN AND PROCEDURES

The basic design for the survey of crafts-artists belonging to crafts membership organizations was to send a detailed questionnaire in the mail to a sample of crafts organization members selected using a two-stage stratified sampling process. In the first stage, crafts organizations responding to an earlier MPR survey were stratified by key characteristics (geographic region, membership size and primary medium in which members worked), and a sample of organizations was selected within each stratum. In the second stage, a sample of crafts-artists was selected from the membership rosters supplied by the organizations responding in the first stage. Mail survey techniques with telephone follow-ups were used in both stages.

The earlier crafts organization survey showed that organizations are distributed fairly evenly by geographic location but are quite unevenly distributed by membership size and primary media categories.

ť



Hence, all large organizations (2,000 or more members) were included in the first-stage sample to ensure that none of these groups, which account for less than 2 percent of all organizations but over 36 percent of the members, was missed. In addition, the less prevalent primary media categories were oversampled to ensure that the resulting sample of crafts-artists permitted reliable estimates of the characteristics of crafts-artists working in each major media type.

The first-stage survey of organizations was completed in the spring of 1980 and the second-stage survey of a sample of crafts-artist members was completed in the fall of 1980. The total sample size for the organizations survey was 281 membership organizations, of which 205 or 73 percent returned usable membership lists. The sample of organization members totalled 5,146 individuals, of whom 3,785 or 74 percent responded. Of this number, 2,637 identified themselves as crafts-artists, while the remainder were not involved in crafts and were dropped from the analysis sample. A detailed discussion of procedures used to collect and tabulate the information provided by crafts-artists, along with copies of fielding forms, is provided in the methodology report for this project. 1/

In addition to the main survey of member crafts-artists, three smaller surveys were conducted to ascertain the extent to which the results of the members survey could be assumed to represent the numbers and characteristics of the total population of crafts-artists. These

<sup>1/</sup>J. Georg Cerf, Constance F. Citro, and Audrey McDonald, Methodology Report: Survey of Crafts-Artists (Princeton, N.J.: Mathematica Policy Research, 1982).

smaller surveys included crafts-artists who sell their crafts at nationally recognized shops and galleries, crafts-artists who exhibit at prominent fairs and crafts exhibitions and subscribers to crafts periodicals not published by crafts organizations. Procedures used for these surveys are also described in the Methodology Report.

The results of the main survey of crafts organization members and the three smaller surveys were tabulated and analyzed in a number of ways. The report which follows this summary presents the study findings under the following headings. The introductory section provides background on the study and describes major characteristics of member craftsartists in terms of primary crafts media, intensity of crafts involvement, net income from crafts, extent of crafts training, region of residence, and years of crafts involvement. Section II develops estimates of the number of crafts-artists in the U.S. based on the results of the members' survey and evaluates these estimates in light of the additional information on the extent of crafts organization membership obtained from the surveys of sellers, exhibitors and subscribers. Section III presents a demographic profile of member crafts-artists and compares their demographic and socioeconomic characteristics to the total adult population. Section IV presents detailed tables on crafts-related characteristics of member crafts-artists. Section V presents the results of a regression analysis directed to determining factors affecting the number of hours member crafts-artists devote to their work and the level of hourly earnings they receive. Finally, Section VI present results of the three smaller surveys of sellers, exhibitors and subscribers and compares these results to the findings for comparable groups of craftsartists, who belong to crafts membership organizations.

#### B. CRAFTS-ARTISTS: HOW MANY ARE THERE?

A primary goal of the research program of the Arts Endowment in the field of crafts is to develop estimates of the number of crafts-artists in the United States. As a national sample survey of the entire U.S. adult population has not appeared a cost-effective means to develop estimates, work has focused on using lists of names believed to represent crafts-artists. A planning study carried out by MPR in 1976 recommended that estimates be developed using membership rosters of organizations of crafts-artists.

Based or data from a survey of crafts membership organizations completed in 1978 and other information, MPR estimated that there were 1,218 crafts membership organizations in the U.S. in 1978 whose membership rolls totalled 458,000. The unduplicated estimate of member crafts-artists in 1978, allowing for individuals belonging to more than one organization, was 375,000. As a result of the recently-completed surveys of organizations and a sample of their members described in this report, it is estimated that in 1980 there were 1,063 to 1,265 crafts membership organizations in the U.S. whose membership rolls totalled 268,000 to 319,000 names. The unduplicated estimate of crafts-artists belonging to these organizations is 150,000 to 180,000. This estimate, for 1980, represents crafts-artists who are currently engaged in one of the crafts media as defined by the Arts Endowment.

The difference between the 1978 estimate of 375,000 member crafts-artists and the 1980 estimate of 150,000 to 180,000 is attributable to several factors. The largest portion of the difference is due to an estimate of membership size of crafts organization for 1980 based on actual counts of membership lists that is about a third lower than the

1978 estimate which was based on broad size intervals checked off by organizations. The next most important factor is the adjustment made in 1980 from member responses, that could not be made in 1978, to subtract out over a quarter of the organization members as not being crafts-artists. Finally, a slightly larger adjustment was made in 1980 compared to 1978 for multiple memberships. Of course, the lower-bound estimate of 150,000 member crafts-artists in 1980 reflects the lower-bound estimate of the number of crafts membership organizations.

One of the research issues here is the extent to which there are crafts-artists with similar characteristics who do not belong to crafts membership organizations. To find out, a series of tests was made which involved selecting populations of crafts-artists without any reference to whether those populations included members of crafts organizations. First, a small number of highly regarded crafts shops, galleries, exhibits and fairs were identified, and a sample of crafts-artists showing their work at these outlets were asked if they were members of crafts organizations and also sent the entire questionnaire previously sent to member crafts-artists. Second, crafts publications not affiliated with crafts membership organizations were identified and a small sample of their subscribers surveyed.

The results of these tests established that virtually all highly accomplished sellers and exhibitors (100 percent and 95 percent, respectively) are members of crafts organizations and that 73 percent of subscribers are also members of crafts organizations. Moreover, the characteristics of seller and exhibitor crafts-artists are very similar to characteristics of crafts-artists responding to the member survey who devote 40 or more hours per week to crafts. The characteristics of

V ...

subscriber crafts-artists are very similar to those of all craftsartists in the member survey. Accordingly, there is a basis for
concluding that the survey of crafts-artists belonging to crafts membership organizations is representative of the characteristics of
highly-skilled crafts-artists and of subscribers as well as members,
and that the estimated population of member crafts-artists overlaps to a
large degree with the total crafts-artist universe.

## C. A PROFILE OF CRAFTS-ARTISTS AND THEIR CRAFTS ENVIRONMENT

As individuals, crafts-artists who are members of crafts organizations are more likely to be female and white than the total adult U.S. population age: 20 or older. (Over 70 percent of crafts-artists are women and almost 98 percent are white, compared to figures of 53 and 85 percent, respectively, for the adult population.) Crafts-artists have close to the same median age as the adult population (45 years compared to 42 years) and a similar distribution by marital status (72 percent of crafts-artists are married compared to 66 percent of the adult population). Similar proportions of crafts-artists and all adults are in the labor force (62 percent versus 65 percent) and, of those who are working or available for work, similar proportions are employed (95 perdent of crafts-artists who are in the labor force are employed and 5 percent unemployed compared to figures of 94 percent employed and 6 percent unemployed for all adults). Crafts-artists differ from the adult population in terms of educational status (57 percent of crafts-artists have completed college compared to 17 percent of the adult population 25 years and older) and occupation (34 percent of crafts-artists list their main occupation as crafts and another 38 percent list professional and technical occupations, while only 16 percent of the adult population aged 18

or older have professional or technical occupations). Annual household income of crafts-artists on average is about 35 percent higher than mean annual household income for the whole U.S. (\$28,900 compared to \$21,400), although the annual income of crafts-artists themselves is lower than per capita annual adult income on average (\$9,500 compared to \$12,200).

Turning to characteristics of crafts-artists replated to their crafts work and environment, about 42 percent of all member crafts-artists consider fiber to be their primary media. The two next most prevalent media are clay and wood with 17 percent each, followed by metal with 13 percent. Leather and paper combined account for 5 percent, while glass, other media and combinations of media are nearly 7 percent.

The level of involvement of crafts-artists in crafts work ranges from occasional activity to crafts as a full-time occupation. About 25 percent of respondents described crafts as their major or primary activity in life; one-third of these crafts-artists spend 40 or more hours per week producing crafts while the remaining two-thirds spend less time. Another 17 percent reported crafts is a secondary activity. Crafts is a leisure activity of about 31 percent of respondents, of whom almost half spend 10 hours or more per week on crafts and the remainder less than ten hours. Almost 14 percent of crafts-artists reported crafts is an occasional activity and the remaining 13 percent indicated another type of involvement, such as primarily teaching crafts or a student of crafts.

Looking at primary media and level of involvement together, fiber is more popular than other media among crafts-artists but as a secondary, leisure or occasional activity rather than a full-time or major activity. Clay and metal are the most prevalent primary media for crafts-artists

who consider crafts their primary activity. Glass is also prevalent as a primary activity. On the other hand, wood is most frequently mentioned as a leisure or occasional primary media.

Current trends in crafts design and work are reported as the most important influence on crafts-artists' work. American folk themes and ethnic/cultural traditions are also important sources of inspiration.

Most crafts-artists produce one-of-a-kind objects and those who engage in crafts as a primary activity accept commissions at an average of 22 per year.

The most popular form of crafts training is "self-taught;" however, the most frequently reported formal approaches are workshops at
crafts schools and college or university courses in crafts. Least
prevalent are apprenticeships. The relationship between amount of
training on the one hand and intensity of involvement on the other is not
straightforward, with some media "requiring" more training than others.
Fiber and wood are more self-taught than other media.

Satisfaction with past training is high and about 15 percent of all crafts-artists are presently receiving some form of training for an average of one hour every week. Two-thirds plan to acquire additional training in crafts. Lack of time and money are the most important barriers for more training.

Training in business skills has been acquired by about 30 percent of crafts-artists. Nearly 60 percent of those who have not received such training indicate a desire for it.

The "average" crafts-artist has been involved in crafts for 16

years and works almost 20 hours per week designing and producing crafts.

An additional 2 hours per week are spent in marketing. About 90 percent

of crafts-artists visited a crafts gallery or museum in the 12 months

prior to the survey and read an average of 2.4 crafts publications. Twothirds of crafts-artists also collect crafts.

Crafts-artists tend to work alone and in their home. About a third report that other family members work in crafts, but often in a different media. One-third also replied that their parents were involved in crafts.

Crafts-artists who own their homes outnumber renters by a ratio of about 4 to 1. About 34 percent live in rural areas, another 37 percent in the suburbs and less than 29 percent in urban areas. Over 57 percent live in communities of less than 50,000 population. Crafts-artists reside in all regions of the U.S., but proportionately more are found in the Northeast (28 percent) and South (29 percent) compared to the North Central region (24 percent) and West (19 percent).

creative or intangible aspects of crafts as a "means of creative expression," "sense of accomplishment," and "life's work" outrank more pecuniary aspects as important satisfactions for crafts-artists. However, the barriers to achieving these satisfactions are non-craft obligations, money, time, lack of public appreciation, marketing opportunities and training. The most frequent goals set by crafts-artists are to devote more time to crafts and to develop artistic competence; increasing income and winning an award or recognition are less important.

Over 82 percent of all crafts-artists exhibited their crafts within the 12 months prior to the survey, with nearly half doing so from two to five times. About 42 percent of the respondents who exhibited received an award or prize within 3 years of the survey. Art/crafts fairs and galleries were each listed by 35 percent as the best places to

exhibit. Crafts shops (either their own or someone else's) were the next most popular. The lack of outlets and poorly informed communities were listed as the major problems by crafts-artists who exhibit.

About two-thirds of all crafts-artists produced goods for sale; of these, 23 percent report receiving their primary income from crafts sales. Art/crafts fairs and galleries are again listed as the best places to sell. Too few outlets, dealing with clients, lack of business skills and a poorly informed public are listed as the major problems.

Relatively small proportions, each less than 10 percent, have received a grant or fellowship, applied for a crafts-related bank loan, purchased equipment or materials in cooperation with other crafts-artists or exchanged crafts for other goods and services.

Crafts provides a relatively small contribution to net household income and is about as likely to be a drain on household income as a contribution. The estimated average annual crafts-related income of crafts-artists is about \$2,800. Nearly 43 percent of crafts-artists teach crafts and earn an average of about \$1,100 per year from teaching. Crafts expenses are estimated on average annually at about \$2,600, leaving a net gain of \$345 (based on data from those who reported both income and expense categories).

Average non-crafts personal income is about \$8,000 per year while other family members earn about \$14,900 (\$14,300 of it from non-crafts sources). Overall, average annual income for crafts-artist households is about \$28,900.

Materials are the most freque t crafts-related expense, amounting to about \$1,000 per year for those who buy them. Dues and publications are the next most prevalent expense at about 70 percent each and costing

an average of \$48 and \$85 per year respectively. The most expensive item--employee salaries at \$6,500--is among the most infrequent cost with only 4 percent of crafts-artists reporting that they have employees.

## D. FACTORS RELATING TO CRAFTS INVOLVEMENT AND FINANCIAL RETURNS

In addition to the extensive tabular analysis of characteristics of member crafts-artists summarized above that was performed for this report, responses of crafts-artists to the survey of crafts organization members were empirically analyzed using regression techniques to attempt to relate underlying causal factors to specific outcome variables. The empirical analysis focused on two aspects of crafts-artists activity: their level of involvement as measured by weekly hours of producing and marketing crafts and the financial returns as measured by gross hourly earnings for those who sell their products.

There is considerable variation in both work hours and hourly earnings, and the estimation yielded a number of interesting insights which help in understanding their determinants. However, there are two caveats that must be stressed in interpreting the results. First, the pursuit of crafts is of course motivated by noneconomic factors such as artistic accomplishment, as well as by economic factors, which suggests that the rewards to crafts activity consist of nonmonetary as well as monetary elements. Hence, an analysis of the financial returns to crafts in terms of hourly earnings will understate the true returns to the extent that nonpecuniary remuneration is present. Using a regression approach in conjunction with the data available, it is not possible to identify the relative mix of these different remunerative elements.

Second, individuals may engage in crafts as an avocational activity, as an income producing activity, or both. Persons with identical observed characteristics may differ in their efforts to sell crafts. This difference in sales orientation, which is not observed in the data, will manifest itself in different hourly earnings measures. It may, in addition, be correlated with observed characteristics such as training and exhibition experience so that the estimated effects of the latter variables may partially incorporate underlying variation in sales orientation. Hence, the empirical findings should be interpreted with some caution.

With regard to weekly hours of crafts involvement, the empirical findings suggest that crafts-artists respond positively to a greater potential wage rate, i.e., they work more hours per week the higher the financial return they can expect to gain. The potential hourly wage rate was imputed to all 1,363 crafts-artists respondents who reported the number of hours worked in crafts and other variables, based on regression estimates for the 959 crafts-artists who reported selling their crafts. The imputed hourly wage rate is intended to proxy expected total remuneration, including both monetary and nonmonetary elements. Thus, it serves as a more inclusive measure of the incentive to engage in crafts activity than monetary income alone.

The number of hours spent in producing crafts does not display a consistent relationship to skills and the acquisition of training, although exhibition experience, apprenticeship, and college teaching are important. But, these factors may in themselves represent measures of involvement rather than causal factors underlying crafts activity.

Crafts do compete with other personal activities and the data offer some

indirect evidence of this competition for one's time: individuals from larger families and who own their own homes spend less time in crafts. Rural residents are estimated to work more intensely in crafts than those living in more populated areas. While males are estimated to receive higher earnings than females with similar crafts-related characteristics, they tend to work fewer hours in producing and marketing crafts. This implies that crafts may be a more important form of leisure for females than males. Finally, crafts involvement is not significantly related to the level of income earned by other household members.

Turning to the analysis of what influences hourly earnings, the estimation finds a positive financial return to training in business-related skills and to certain types of training in crafts production, specifically apprenticeship programs and college courses. The general work experience that crafts-artists accumulate over time also has a significant pecuniary impact. Artistic quality and recognition are found to be significant determinants of earnings. This impact, however, is primarily due to the ability to exhibit one's work rather than to the receipt of awards or financial support. Finally, individuals specializing in fiber and paper media earn significantly less than otherwise similar crafts-artists concentrating in other media.

#### E. POLICY IMPLICATIONS

There are several findings from this analysis which may offer guidance to policymakers and crafts organizations who are concerned with assisting crafts-artists to attain greater financial success. The tentative implications for policy that can be inferred from the empirical analysis are:

- Training Not all types of training lead to an increase in earnings for crafts-artists. More specialized and intensive methods, such as apprenticeships and college courses, appear to offer larger financial rewards. Business-related skills training is significantly related to earnings which makes sense because crafts-artists are similar to self-employed business persons. Greater encouragement and provision of these programs seem to be warranted.
- Exhibitions and Shows As was expected, the analysis found a strong relationship between number of exhibits and increased earnings of crafts-artists. The exposure received by crafts-artists through exhibiting their products is clearly a key element in financial success. Shows enhance the access of customers to artists and facilitate relative comparisons with respect to quality and style. Efforts to increase availability of, and participation in, crafts exhibits should have significant payoff in helping crafts-artists increase financial rewards from crafts production.
- Awards and Financial Support The analysis found no perceptible effect on either crafts activity or crafts earnings of having received a crafts award or prime in the past three years or of ever having received a fellowship, grant, or other form of support from a government agency or foundation. This finding suggests that the resources of the Arts Endowment and other organizations for assisting crafts-artists might be better utilized if they were directed to other modes of assistance.
- o Imbalance of Supply and Demand There is some tangential evidence that there may be some overcrowding (i.e., an imbalance of supply and demand) in fiber-related crafts, in that they are the most popular, yet earn the least income. This conclusion must remain speculative, especially given the unobserved differences in sales orientation, which may be such that fiber artists are less oriented to selling than crafts-artists in other media. Nevertheless, the financial implications of the differences in earnings by media warrant further investigation.

Further implications can be formulated based on some additional research efforts. Further empirical analyses that could be carried out with the data from the 1980 Survey of Member Crafts-Artists include:

ERIC

- Extension of the wage rate analysis to all craftsartists in the sample, using appropriate statistical techniques, so that the unobserved decision to sell or not can be confronted directly.
- o Investigation of the factors prompting crafts-artists to pursue crafts as a primary occupation.
- o Exploration of the relationship between financial success and the barriers to exhibiting and selling crafts. (The Arts Endowment has sponsored research on this topic in four cities. The data from the Survey of Member Crafts-Artists would permit further investigation with a nationally representative sample.)
- o Investigation of the level of financial returns to producing crafts in response to current trends, traditions, inspiration, and demands of the market.
- o Examination of the link between crafts activity and other employment.

It would also be useful to pursue further the question of the extent to which the population of crafts-artists who belong to crafts membership organizations fully represents in number and characteristics the total universe of all crafts-artists. The Arts Endowment currently obtains data on leisure activities of Americans from a sample of persons who are included in the National Crime Survey, totalling about 18,000 cases on an annual basis. Several questions are asked of this sample about participation and training in crafts. Characteristics of persons active in crafts from this sample could usefully be compared to the characteristics of respondents to the Survey of Members.

Another approach to consider is to conduct a larger representative survey of subscribers to crafts periodicals not published by crafts membership organizations. Over 25 percent of the subscribers included in the small survey conducted for this report did not belong

to crafts organizations; however, analysis of this group in any depth is not possible as questionnaires were obtained from only 168 subscriber crafts-artists in all, including both organization members and non-members. A larger survey of subscribers that was designed to be representative of the subscriber universe could provide another useful look at the crafts-artist population and perhaps turn up valuable additional information to supplement the profile obtained from the survey of crafts organization members.

#### I. INTRODUCTION

This report is an endeavor to deepen our understanding of craftsartists and to provide information about their characteristics and
concerns that will be useful to the National Endowment for the Arts.

Accordingly, major purposes of the report are to describe the personal
and crafts-related characteristics of crafts-artists and to explore some
of the important relationships between participation and success in
crafts and these characteristics. Other goals are to prepare an estimate
of the number of crafts-artists in the United States and to determine if
the information provided by crafts-artists included in this sample
applies to all crafts-artists.

Before proceeding to these topics, several preliminary matters are discussed in this introductory section:

- o A brief history of studies conducted for the Arts Endowment concerning crafts-artists
- o The rationale for the approach adopted to collecting information about crafts-artists
- o An outline of the procedures followed to gather the data for this report
- Description of the major characteristics of crafts-artists
- o The organization of the tables used in this report.

Following these topics is the body of the report in five main sections:

- (1) Estimates of the number of crafts-artists (Section II) &
- (2) A description of the personal and household characteristics of crafts-artists (Section III)

ERIC

I.1

- (3) The crafts-related characteristics of crafts-artists as well as their attitudes toward crafts, their satisfactions and goals, and their problems (Section IV)
- (4) An examination of some specific relationships between participation and success in crafts and the characteristics of crafts-artists (Section V)
- (5) Results of tests to determine the extent of crafts organization membership (Section VI).

A copy of the questionnaire used in the survey conducted for this report along with a comparison of survey results by type of interviewing procedure are included as appendices. A third appendix compares selected survey results to a 1978 survey of crafts-artists in Canada and a fourth appendix discusses a model of crafts involvement. All the data sampling, data collection and data tabulation procedures along with exhibits are described in the Methodelegy Report for this project. 1/

## A. PRIOR RESEARCH ON CHAPTS ORGANIZATIONS AND CRAFTS-ARTISTS

In 1976, the Research Division of the National Endowment for the Arts initiated a planning study for the purpose of determining the feasibility and methods for a national survey of crafts-artists. One of the recommendations from this study was that a survey of crafts member—ship organizations should be conducted prior to surveying individual crafts-artists. This recommendation was adopted by the Arts Endowment



<sup>1/</sup>J. Georg Cerf, Constance F. Citro and Audrey McDonald, Methodology Report: Survey of Crafts-Artists (Princeton, N.J.: Mathematica Policy Research, 1982).

<sup>2/</sup>Constance Citro and Diana Zentay, A Planning Study for a National Survey of Crafts Arts (Washington, D.C.: Mathematica Policy Research, January 1977). See also Research Division Report No. 2, To Survey American Crafts: A Planning Study.

and the Survey of Crafts Organizations completed in 1978. The results were also published by the Arts Endowment. Both the planning study and the survey of crafts organizations provided strong evidence that crafts-artists belonging to organizations are a major component of the total crafts-artist population. A mail survey of the membership rosters of the sample of crafts organizations was recommended as a cost-effective means for gathering information about the population of crafts-artists belonging to such organizations.

Based on the results of the 1978 Survey of Organizations, the population of member crafts-artists had been estimated at about 375,000 for the U.S. Based on the survey of members that was conducted for this report, this figure has been revised, so that it is now estimated that there are 150,000 to 180,000 crafts-artists who belong to crafts membership organizations.

B. RATIONALE FOR A SURVEY OF CRAFTS-ARTISTS WHO ARE MEMBERS OF CRAFTS ORGANIZATIONS

During the early phases of the research on crafts, it became clear that a national probability survey to gather specific information about crafts-artists and their work would be prohibitively expensive. As already discussed, the planning study concluded that surveys of crafts membership organizations and their members would be feasible and provide

<sup>1/</sup>Constance Citro, et al., Results from a National Survey of Crafts Membership Organizations (Washington, D.C.: Mathematica Policy Research, June 1978).

<sup>2/</sup>Research Division Report No. 13, Craft Artist Membership Organizations 1978.

useful information about an important component of the crafts world.

Moreover, evidence suggested that there are relatively few crafts-artists who are not members of at least one crafts organization, so that surveying a sample of members would be a cost-effective means of gathering data on the crafts-artist population.

This hypothesis cannot be tested using only information from the sample frame developed as a result of the Planning Study. To determine if membership in crafts organizations is as widespread as some anecdotal information suggests, two separate tests were designed and included in the present study. Other ways can also be found to verify the extent of members, ip. One test that was carried out involved selecting establishments and fairs which sell or exhibit the works of very accomplished crafts-artists and sampling these crafts-artists. There is some basis to believe that "professionals" may not be as likely to join crafts organizations as those with less skill. A second test was to survey subscribers to crafts periodicals which are not published or associated with a crafts membership organization. The argument here is that these periodicals may include many crafts-artists who either are not aware or do not choose to belong to an organization, but still wish to maintain some contact with trends and developments in their praticular media or crafts generally.

Based on the results of these tests, over 95 percent of seller and exhibitor crafts-artists are estimated to belong to at least one of the crafts membership organizations included on the master list from the prior Survey of Crafts Organizations. About three-quarters of all subscribers are estimated to be members of crafts organizations.

A more complete analysis is included as Section VI of this report.

ERIC Full feat Provided by ERIC

#### C. DATA COLLECTION PROCEDURES

The information presented in this report regarding crafts-artists who are members of organizations was obtained from a two-stage stratified cluster sample of members of organizations identified in the 1978 Survey of Crafts Membership Organizations conducted by Mathematica Policy Research, Inc. (MPR). The two-stage nature of the design results from initially selecting organizations and then individual members from those organizations (the latter stage producing a clustered sample). Initially 281 organizations were sampled, of which 205 or 73 percent returned usable membership lists.

The sample was stratified because the earlier work showed most crafts-artists were concentrated in a few media with smaller numbers working in other materials. To adequately represent all crafts-artists, the most popular media (fiber and clay) were undersampled, and the least popular (paper and leather) were oversampled. Adding up respondents from all the strata, 5,146 were sampled and responses were received from 3,785 persons or 74 percent. Of this number, 2,637 identified themselves as crafts-artists, while the remainder were not involved in crafts and were dropped from the analysis sample. 1/

Questions were developed by MPR based on prior research and suggestions by the Research Staff of the Arts Endowment and consultants to the Endowment and the project. A draft questionnaire was pretested prior to its final printing. A copy of the questionnaire is in Appendix A.



 $<sup>\</sup>frac{1}{A}$  small number of questionnaires, 27, were completed by crafts-artists, but could not be processed for analysis because crucial data were missing and could not be obtained.

The fielding procedures used required extensive mailings with telephone follow-up of non-respondents. The sequence of contacts is shown below for the two main surveys (the times in parentheses indicate the elasped time since the last contact):

## First Stage:

## Survey of Crafts Organizations

Initial request for names of members
Telephone follow-up (3 weeks)

#### Second\_Stage:

## Survey of Crafts-artists

Initial letter
First questionnaire (2 weeks)
Follow-up post card (1 week)
Follow-up letter (1 week)
Second questionnaire (3 weeks)
Telephone follow-up (6 weeks)

Use of the telephone resulted in significant additions to the number of membership lists received and questionnaires returned in the mail. In addition, interviewers completed questionnaires over the telephone. Questionnaires returned as a result of the telephone follow-up were examined to see if there were any major differences in characteristics reported by these crafts-artists and those not contacted by telephone. The results are included as Appendix B to this report.

When completed questionnaires were received in the mail, each was edited and coded. As part of this process, 1,471 respondents were called back to clarify an answer or obtain a missing answer to particular questions which were important for the analysis. A computer based file was created and the data verified. Corrections were made when they were necessary.

In order to properly reflect the relative numbers of craftsartists drawn from each stratum in the tables and to prepare the estimates of the overall number of crafts-artists, a weight for each of the 2,637 crafts-artists respondents was developed. These weights consider the media in which crafts-artists work, the number of organizations to which they belong, and the level of response within the groups used to select the sample. The tables included in this report were prepared using these weighted responses.

In order to provide comparability between the survey of member crafts-artists and the surveys of professional sellers and exhibitors and subscribers, the same questionnairs was used and the data gathering procedures were similar.

#### D. MAJOR CATEGORIES USED IN TABULATIONS

The tabulations presented in this report consist of tables showing characteristics and opinions of crafts-artists cross-classified by other characteristics. These cross-tabulations result from creating mutually exclusive groups for each of several different key analytical variables selected from the large number of questionnaire items, and then examining characteristics of respondents in each of these groups on other variables collected in the survey. The purpose is to ascertain differences among the various groups and secondarily, to identify potential relationships between each analytical variable and the characteristic or data item being tabulated. The major variables used in the tabulations in this report are the following:

- o Primary crafts media
- o Intensity of crafts involvement
- o Net income from crafts
- o Extent of crafts-training

- o Region of residence
- o Years of crafts involvement

Definitions of the categories identified for each of these variables are provided in Figure I.1.

The major variables listed above were selected in the belief that each variable relates significantly to other characteristics of crafts-artists and therefore that crafts-artists would exhibit important differences among the categories identified for each. In several instances, this determination was based on expectations rather than previous research. These expectations were not always met and consequently relatively little use is made of certain analytic variables. On the other hand, media, intensity of involvement, and net income categories usually show significant variation and are used in nearly all of the tables.

As an introduction to the body of the report, a summary of the results of the survey for each of the major variables identified in the cross-tabulations follows. The relationships revealed in tabulations between these and other variables cannot be used to draw any conclusions about causality or direction of causality. Variables selected as major stratifiers may, in some instances, influence variables to which they are related and in other instances may themselves be the outcome of those variables; in yet other instances, two apparently related variables may be independently influenced by yet a third variable not included in the tabulation. Analysis presented in Section V uses multiple regression techniques to investigate predictors (independent variables) of crafts involvement and income (dependent variables) in a causal framework.

3

#### FIGURE I.1

#### DEFINITIONS OF CATEGORIES OF MAJOR CROSS-TABULATION VARIABLES

Primary crafts media. Each respondent was asked to indicate one primary media from among those listed or written in by the respondent. The following are guidelines used when assigning "other" responses to broad media categories.

Fiber: All types of weaving, sewing, working with organic or synthetic strands to make material or finished product.

Clay: Pottery and related types of work including glazing and firing, but excluding china painting.

Metal: Includes jewelry making but not lapidary work; metal working involves forging, forming, fabricating and finishing all varieties of metal and alloys.

Wood: Shaping, joining and finishing wood.

Glass includes glass blowing and forming as well as casting and and assembling stained glass constructions. "Other" includes Other: a miscellany such as plastics, natural materials, and ivory, and combinations of media.

Paper Paper is papermaking and finishing, not calligraphy or and painting applied to paper. Leather includes the processing Leather: of hides as well as the fabrication or decoration of the finished article.

Intensity of Crafts Involvement. A combination of type of crafts involvement (Question 1) and number of hours per week spent producing crafts (Question 22):

Major Respondents reporting crafts as their primary activity
Activity: categorized by number of hours per week spent producing

Secondary All respondents indicating this level of activity in Activity: Question 1.

Main Respondents listing this level of activity in Question Leisure 1 divided into components by the number of hours per week Activity: listed in Question 22.

Occa- Respondents indicating "occasional" in Question 1 and signal spending less than 10 hours per week producing crafts. Activity:

Other: Primarily teaching crafts or a student studying crafts plus respondents not included above.

ERIC Full Text Provided by ERIC

1.9<sub>0</sub> 2*j* 

### FIGURE I.1 (continued)

Net Income from Crafts. The sum of the respondent's income from crafts sales and crafts-related sources (sales of materials and tools) less the respondent's crafts-related expenses. Income from teaching crafts, from non-craft sources or from other family members is excluded.

Extent of Crafts Training. Constructed from the list of kinds of training each respondent reported:

Profesone or more months as an apprentice or 2 or more workshops
sional: given by crafts schools or professionals or private lessons
from professionals.

College 5 or more college art or crafts courses. courses, 5 plus:

College 1 to 4 college art or crafts courses. courses, 1 to 4:

Informal: All other training including lessons from family or friends, school art classes, community programs, adult education courses, self-taught and/or one workshop by a crafts school or professional.

Region of Residence. The state to which the questionnaire was mailed classified into regions as defined by the Bureau of the Census, U.S. Department of Commerce.

Northeast: Connecticut, Maine, Massachusetts, New Hampshire, New Jersey, New York, Pennsylvania, Rhode Island and Vermont.

North Illinois, Indiana, Iowa, Kansas, Michigan, Minnesota, Central: Missouri, Nebraska, North Dakota, Ohio, South Dakota, and Wisconsin.

South: Alabama, Arkansas, Delaware, District of Columbia, Florida, Georgia, Kentucky, Louisiana, Maryland, Mississippi, North Carolina, Oklahoma, South Carolina, Tennessee, Texas, Virginia and West Virginia.

Alaska, Arizona, California, Colorado, Hawaii, Idaho, Montana, Nevada, New Mexico, Oregon, Utah, Washington and Wyoming.

Years of Involvement. A categorization of the responses to the question "How many years altogether have you been working in your craft?"

West:

Examining the distribution of responses to major tabulation variables indicates about 42 percent of all crafts-artists who are members of craft organizations consider fiber to be their primary media. The two next most prevalent media are clay and wood with 17 percent each, followed by metal with 13 percent (Table T.1). Leather and paper combined account for 5 percent while glass, other media and combinations of media are nearly 7 percent. Approximately 96 percent of all respondents provided primary media information. The effect of the weighting can be observed by noting that there are more respondents indicating metal as their primary media compared to wood (424 versus 365), although the weighted percentages indicate that wood is a more frequent primary media (16.6 compared to 12.5 percent). Accordingly, the impact of the weighting scheme is to underweight respondents reporting metal as a primary media while overweighting those reporting wood. 1/2

The levels of crafts involvement result from combining the number of hours crafts-artists spend per week producing crafts and how they described their interest in crafts activities. About 25 percent described crafts as their major or primary activity in life; 8 percent spend more than 40 hours per week producing crafts while the remaining 17 percent spend less (Table I.1). Another 17 percent reported crafts is their secondary activity. Crafts involvement is the main leisure activity of about 31 percent of respondents, a slightly larger share (17



<sup>1/</sup>Details on the procedures used to select the sample and weight the results are included in Cerf, et al., Methodology Report: Survey of Crafts-Artists.

TABLE 1.1

DISTRIBUTIONS OF RESPONSES TO VARIABLES USED FOR CROSS-TABULATION CATEGORIES, SURVEY OF CRAFTS-ARTISTS, 1980

	_		
Cross-tabulation	Percentage Distribution	Unweighted Number of Respondents	
Category			
Primary Crafts Media		·	
Fiber	41.8	968	
Clay	17.4	436	
Metal	12.5	424	
Wood	16.6	365	
Glass, Other, Combination	6.8	203	
Leather and Paper	4.9	<u>140</u>	
Total responding	100.0	2,536	
Not responding		101	
Total interviewed		2,637	
Intensity of Crafts Involvement			
Major Activity		,00	
Under 40 hours per week	16.6	439	
40 hours or more per week	8.2	256	
Secondary Activity	16.9	431	
Leisure Activity			
Under 10 hours per week	17.6	371	
10 hours or more per week	13.5	307	
Occasional Activity	13.8	288	
Other	<u>13.4</u>	<u>328</u>	
Total responding	100.0	2,420	
Not responding		217	
Total interviewed		2,637	
Net Income from Crafts			
Net Loss	00.0	459	
\$500 or more	22.2	• • •	
Under \$500	38.4	659	
Net Gain	10 1	202	
Under \$1,000 <sup>1</sup> /	19.1	382	
\$1,000 or more	20.3	<u>476</u> .	
Total responding	100.0	1,,976	
		661	
Not responding		2,637	

(continued)

TABLE I.1	cor	ıtinu	ed)
-----------	-----	-------	-----

Cross-tabulation Category	Percentage Distribution	Unweighted Number of Respondents	
Extent of Crafts Training			•
Professional	51.8	1,395	
College courses			
5 or more	7.2	200	
1 to 4	17.1	417	
None or informal	23.9	<u>621</u>	
Total responding	100.0	2,633	
Not responding		4	
Total interviewed		2,637	
Region of Residence			
Northeast	28.0	932	
South	29.1	679	
North Central	23.6	533	
west	<u>19.3</u>	488	
Total responding	100.0	2,632	
Not responding	•	5	
Total interviewed		2,637	
Years of Involvement			
Under 10 years	33.3	950	
10 to 14 years	23.9	614	
15 to 24 years	20.1	312	
25 years or more	22.7	<u> 532</u>	
Total responding	100.0	2,608	
Not responding		29	
Total interviewed		2,637	

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

 $\frac{1}{2}$  Includes zero net income.

percent) participating under 10 hours per week compared to 10 hours or more (14 percent). The remaining 27 percent are equally divided between those for whom crafts is an occasional activity and those in the "other" involvement category. Nearly 92 percent of all those interviewed, or 2,420 respondents, provided information used to classify their level of crafts involvement.

Net income from crafts can be considered another measure of intensity or level of crafts involvement with the caveat that positive net income may not be the only criterion of accomplishment. About 75 percent of all respondents reported both income and expenses in sufficient detail to permit calculation of net income. Overall, 85 percent reported gross income from crafts while 82 percent reported crafts-related expenses. 1/ Of crafts-artists providing sufficient information to calculate net income, about 60 percent experienced a net loss in the 12 months prior to completing the questionnaire (Table I.1). Of these, nearly half or 38 percent had a loss of between \$1 and \$500. The remainder had larger losses. The 40 percent who had net income gains are about evenly divided between those with less than a \$1,000 gain (including zero net gain) and crafts-artists with larger net income from crafts. Figures presented in Section IV indicate that crafts-artists with net gains from crafts activities are largely those reporting crafts as their major or secondary activity. Average (mean) net income from crafts is \$1,511 for crafts-artists working 40 hours or more per week in their craft; \$1,167 for other crafts-artists reporting crafts as their

 $<sup>\</sup>frac{1}{T}$  These calculations are based on data presented in Section IV of the report.

major activity; and \$467 for crafts-artists listing crafts as a secondary activity. Figures for other categories of crafts-artists are: loss of \$421 on average for leisure crafts-artists working 10 or more hours per week; average loss of \$403 for leisure crafts-artists working under 10 hours per week; average loss of \$136 for occasional crafts-artists; and gain of \$1,449 on average for all other crafts-artists (Table IV.21).

The extent of crafts training is used as a tabulation variable primarily in the examination of training-related characteristics. Over half--52 percent--of crafts-artists received professional level training in their primary media (Table I.1). The remainder are split about equally between receiving some college level courses and no or informal training. Virtually all respondents reported training information.

Slightly larger shares of crafts-artists live in northeastern and southern states compared to the north central and western states.

Again, nearly every respondent could be classified by state of residence.

One-third of crafts-artists have been involved in crafts for under 10 years while from 20 to 25 percent have been engaged in crafts for 10 to 14 years, 15 to 24 years, or more than 25 years. Nearly 99 percent of respondents reported this information (Table I.1).

#### E. ORGANIZATION OF THE TABLES USED IN THIS REPORT

The following conventions were used to develop tables which are compact yet display data in such a way that comparisons can be easily made:

All percentage distributions, percentage of all crafts-artists and averages (means) are based on weighted data.

ERIC

- o All "number of respondents" are unweighted.
- The lack of statistical significance at the .01 or 1 percent level, as measured by the X<sup>2</sup>(chi square) statistic, is shown by an (\*) for all tabulations which include percentage distributions or percentage of crafts-artists. I/ Each percentage distribution which adds to 100.0 percent or percentage of all crafts-artists, is the unit for determining significance. The percentage of all crafts-artists entries were calculated as yes-no variables to allow formulation of an X'statistic; "yes" equalling all those who have the characteristic, attitude or problem, and "no" all those who did not so indicate. The 1-percent level was selected because a higher level, 0.5 percent, would not have resulted in rejecting a much larger number of tabulations as due to chance, and a lower level, 5 percent, is likely to ignore the effect of nonsampling error on the
- o Averages (means) are shown for all subgroups where there are at least 50 unweighted respondents. A base of less than 50 was judged too small to provide a reliable estimate and could result in misleading conclusions. The symbol # is used in place of the average to indicate the base is insufficient.
- o The marginal (or uncross-tabulated) distribution is shown for every question variable and is repeated on the second page of two-page table formats. The number of respondents for the marginal tabulation is normally more than the sum of the number of respondents for the subcategories because not all respondents answered the questions used to formulate the subcategories (for example of the 2,632 tabulated questionnaires specifying the various media in which they worked, 2,536 specified a primary media).

The X statistic is a way of testing to see if the level and pattern of responses in a cross-tabulation is likely to have occurred merely by chance. If the table (or individual tabulation) is significant, given some level of probability—.01 as in this case, then the pattern of data shown would result merely from chance in only 1 out of 100 instances of similarly drawn samples (1/100 = .01). It is important to remember that X indicates only that the pattern in the table can be accounted for by chance 1 out of 100 times; it cannot be used to draw conclusions about any causal relationships between the variables shown in the table.

- o The question variables are listed in reasonably complete detail, but the subcategories across the top of the tables are more cryptic due to space limitations. The categories are defined in Figure 1.1.
- o A "--" in a table indicates a zero percentage.
- o Percentage distributions were rounded so as to add to exactly 100.0 percent; the largest percentage bore the change of .1 or in very few cases .2 percentage points.

The above points cover the general rules for the tables. Deviations and a few special cases are covered in footnotes on the individual tables.

#### II. ESTIMATES OF THE NUMBER OF MEMBER CRAFTS-ARTISTS

A primary goal of the research program of the National Endowment for the Arts in the field of crafts is to develop estimates of the number of practicing crafts-artists in the United States. Impressionistic evidence, such as the space devoted to crafts in the press, suggests that the crafts-artist population is not small and has increased over the past two decades. The challenge is to substitute hard numbers for impressions.

A national sample survey of the entire U.S. adult population has not appeared a cost-effective means to develop estimates of the number of crafts-artists. Hence, the thrust of the Arts Endowment's research effort is to develop estimates by building up lists of names believed to represent crafts-artists.

Earlier research suggested that membership rosters of organizations of crafts-artists, ranging from large national organizations (such as the American Crafts Council) to small local associations of ceramists, weavers, etc., are a particularly good source for developing estimates of the total crafts-artist population. This section first presents the estimates that resulted from the recently-completed sample survey of members of crafts organizations. Then these estimates are compared to earlier figures derived from the 1978 Survey of Organizations.

To summarize the findings, the work performed in 1978 resulted in estimates that there were 1,218 crafts membership organizations in the U.S. whose membership rolls totalled 458,000. The unduplicated estimate



38

of member crafts-artists, allowing for individuals belonging to more than one organization, was 375,000. As a result of the 1980 surveys, it is estimated that there are 1,063 to 1,265 crafts membership organizations in the U.S. whose membership rolls total 268,000 to 319,000 names. The unduplicated estimate of crafts-artists belonging to these organizations is about 150,000 to 180,000.

The difference between the 1978 estimate of 375,000 member crafts-artists and the 1980 estimate of 150,000 to 180,000 is attributable to several factors. The largest portion of the difference results from reducing the estimate of membership size of crafts organizations in 1980 by about a third compared to the 1978 estimate, based on more precise data. The next most important factor is the adjustment made in 1980 based on survey responses, that could not be made in 1978, to subtract out over a quarter of the organization members as not being crafts-artists. Finally, a slightly larger adjustment was made in 1980 compared to 1978 for multiple memberships, and, of course, the lower-bound estimate of 150,000 member crafts-artists in 1980 reflects the lower-bound estimate of the number of crafts membership organizations.

A. ESTIMATES OF CRAFTS-ARTISTS BELONGING TO MEMBERSHIP ORGANIZATIONS IN 1980

As described in detail in Chapter VI of the companion Methodo-dology Report, information obtained from the survey in 1980 of a sample of organizations and the subsequent survey of a sample of their members was used to construct an estimate of crafts-artists belonging to membership groups following careful step-by-step procedures. In overview, these procedures were to:

- Develop counts of names of individuals (excluding libraries, institutions, etc.) on the mailing lists obtained in the 1980 Survey of Organizations from a sample of 204 groups and inflate ("blowup") these counts to represent the total number of individual names on the lists of all crafts membership organizations. This procedure resulted in an estimate of 268,377 names on the lists of 1,063 organizations.
- o Develop estimates of the proportion of the total number of names that represented practicing crafts—artists as opposed to other persons belonging to a crafts organization. These estimates were based on responses to the 1980 Survey of Members where persons either filled out a questionnaire or sent back a post-card excluding themselves from the survey as not being a crafts—artist. Based on these responses, an estimated 72.5 percent of all names on crafts organization membership lists in fact represent crafts—artists. Applying this figure to the estimate of 268,377 total names gives an estimate of 194,451 names of crafts—artists on the rolls of membership organizations.
- o Further adjust the estimates of crafts-artist names on organization mailing lists to account for the fact that some crafts-artists belong to more than one group. In response to a question on the 1980 Survey of Members, 41.4 percent of respondents reported belonging to more than one group, with 23.6 percent belonging to two, 10.8 percent to three, and 7 percent to form or more. Making adjustments for these responses results in an unduplicated estimate of 152,259 practicing crafts-artists belonging to crafts membership organizations.

All of the procedures and adjustments described above took into account the complex sampling scheme used for the first-stage and second-stage 1980 surveys. They also allowed for differing response rates by sample stratum and geographic location.

 $<sup>\</sup>frac{1}{}$  Interestingly, 18 percent of respondent crafts-artists did not report they were a member of any crafts organization, even though their names were on crafts organization membership lists.

As discussed below, evaluation of the estimate of 152,000 member crafts-artists developed in the Methodology Report leads to the conclusion that this estimate represents a lower bound. An upper bound estimate is that there are about 181,000 crafts-artists belonging to crafts membership organizations.

### B. COMPARISON TO 1978 ESTIMATES

Based on the 1978 Survey of Crafts Membership Organizations, MPR developed a preliminary estimate of persons on crafts organization membership rolls of about 458,000 and an unduplicated estimate of crafts-artist members of about 375,000 for the country as a whole. This latter figure is more than double the estimate of about 152,000 crafts-artists belonging to membership organizations in 1980 represented above. Several reasons account for this difference.

The 1978 estimate was based on a larger estimate of the number of active crafts membership groups--1,218 compared to the 1980 estimate of 1,063. The lower estimate for 1980 resulted from dropping responses to the first-stage Organizations Survey indicating that a group was a chapter or duplicate of another group or was really not a crafts membership organization despite having responded to the 1978 Survey. 2/

The question is whether the 1980 estimate is too low for one or both of two reasons:



<sup>2/</sup>Citro et al., Results from a National Survey of Crafts Membership Organizations, Figure II.2. The estimate of 458,000 names was calculated using the first-cut estimates in this figure adjusted for the revised completion rate.

 $<sup>\</sup>frac{2}{\text{All}}$  other responses were included in the 1980 estimate, including refusals and groups that had become defunct since 1978.

- o It excludes groups as not being crafts membership organizations when they should be included
- o It makes no allowance for the formation of new crafts brganizations since 1978.

Of the respondents to the 1980 Organizations Survey, about 6 percent, including some large groups such as the China Painters, stated they were not crafts groups even though they had responded to the earlier survey. There seems no choice but to accept such self definitions, although if all of these respondents had continued to affirm themselves as crafts groups, then the estimate of organizations would have been 1,131 instead of 1,063 and the resulting estimate of crafts-artist members would have been correspondingly higher.

One can reasonably conclude that the 1980 estimate of 1,063 organizations is too low because insufficient allowance is made for the formation of new organizations since the earlier survey. The 1978 Survey found that 16 percent of responding groups had been formed less than four years ago. Assuming that this rate of formation continued since 1978, one would estimate a total of 1,265 groups instead of 1,063. Offsetting this higher estimate is the fact that the estimate of 1,063 organizations includes groups disbanding since 1978, roughly 8 percent of the total. Hence, the estimate of 1,265 organizations should probably be viewed as an upper bound. Of course, using this higher estimate would result in a higher estimate of crafts-artists belonging to organizations.

Taking the more conservative estimate of 1,063 organizations, (the figure that was used in developing the crafts-artist estimate in the Methodology Report) and multiplying by the average membership size

ERIC

11.5" 42

from the original 1978 estimate, gives an estimated number of names on crafts organization mailing lists of 400,000 compared to the 1978 estimate of 458,000 and the 1980 estimate of 268,000. Hence, the impact of the lower estimate of organizations is to reduce the estimate of names in 1978 by 58,000.

The explanation for the difference between 400,000 names and 268,000 names in 1980 has to do with differences in the average membership size attributed to the organizations. In 1978, membership size was estimated using midpoints of the pre-printed size intervals on the questionnaires checked by the organizations responding to the 1978 Survey. These size intervals were broad, for example, 10C-499 members, 500-1,999, and so on. The 1980 surveys provided actual counts of names on membership lists furnished by a sample of organizations. It turned out that most organizations had picked the correct size interval on the 1978 questionnaire, but that their membership counts were lower than the interval midpoints. The differences were enough to subtract 132,000 from the 1978 estimate of 458,000 names.

The adjustments made in 1978 to refine the estimate of 458,000 names on crafts organization mailing lists to an unduplicated estimate of crafts-artist members included no correction to eliminate persons on membership rolls who were not practicing crafts-artists. In 1980, members were personally surveyed and 27.5 percent declared they were not a crafts-artist and therefore should be removed from the sample.

The 1978 estimate of 375,000 crafts-artists did include an allowance for multiple memberships. This was based on very sketchy data, and, on average across geographic areas, the factor used was 18 percent.

ERIC

11.6, **43** 

In 1980, crafts-artists were asked how many crafts groups they belonged to. Adjustments made based on these responses reduced the estimate of crafts-artist names by 22 percent, or slightly more than the analogous adjustment made in 1978.

In summary, four factors are determined to have produced the 1980 estimates of 268,000 names on crafts organization membership rolls and 152,000 unduplicated crafts-artists compared to the 1978 estimates of 458,000 and 375,000, respectively. In order of computation, the proportion of the total difference between the 458,000 names on membership rolls estimated in 1978 and the 152,000 individual crafts-artists estimated in 1980 accounted for by each of the factors discussed above is:

- o 19 percent due to a smaller estimate of the number of organizations in 1980
- o 43 percent due to a reduced estimate of membership size
- o 24 percent due to subtracting members who were not crafts-artists in 1980 compared to no such adjustment in 1978
- o 14 percent due to a larger percentage of multiple memberships factored out in 1980 compared to 1978 so that individual crafts-artists are only counted once.

Clearly, the most important factor in reducing the 1980 estimate of crafts-artists was the revised data on membership size followed by the new information on proportion of members who were not crafts-artists.

The 1980 estimate, although based on much more complete information than was available in 1978, is still an estimate rather than a certainty. It is subject to sampling error in that the data on member-

ERIC

11.7 44

ship size, proportion not crafts-artists, and multiple membership come from samples of organizations and individuals. The 1980 estimate is also a lower bound based on the discussion earlier that one factor, namely the total number of organizations, is itself an estimate that appears to be on the low side. Taking the higher estimate of 1,265 crafts membership organizations, gives an estimate of about 319,060 instead of 268,000 names on crafts organization membership rolls and an unduplicated estimate of about 181,000 instead of 152,000 member crafts-artists.

Finally, as stated in previous reports,  $\frac{1}{t}$  there is no doubt that estimates of crafts-artists belonging to organizations exclude many members of special groups working in ethnic crafts traditions outside the mainstream of professional crafts activity. Such special groups include Native Americans, persons of Spanish origin, the Amish, and the like.

It is also true that estimates of crafts-artists based just on crafts membership organization rosters exclude similar crafts-artists who do not, for one reason or another, belong to organizations. A chief purpose for conducting the three smaller surveys of crafts-artists who sell at well-known galleries, those who exhibit at well-known fairs, and those who subscribe to crafts periodicals not published by a crafts membership organization was to determine the proportions of crafts-artists in these samples who were members of crafts organizations. As reported in Section VI, 100 percent of sellers, 95 percent of exhibitors,

ERIC Full Text Provided by ERIC

11.8 45

<sup>1/</sup>Citro and Zentay, A Planning Study for a National Survey of Crafts Arts; and Citro, et al., Results from a National Survey of Crafts Membership Organizations.

and 73 percent of subscribers are estimated to belong to crafts membership organizations. The nature of the three additional surveys—namely that they are based on small, purposive samples—means that estimates of non-member crafts—artists to add to the estimates based on the members survey cannot be developed (see further discussion in Section VI). However, the membership proportions among sellers, exhibitors, and subscribers are indications that the universe of member crafts—artists overlaps to a large degree with the total universe of crafts—artists.

ERIC

11.90 46

### III. PERSONAL PROFILES OF CRAFTS-ARTISTS

## A. DEMOGRAPHIC PROFILE

Based on all responses to the Survey of Member Crafts-Artists, the population of crafts-artists who belong to crafts organizations is predominantly older, white, married and female. Over half completed 4 years of college with many majoring in fine and applied arts. Forty-six percent of all crafts-artists are employed full-time (30 or more hours per week) and another 13 percent are employed part-time. A third of employed crafts-artists list crafts work as their primary occupation, while another third categorize themselves as professional workers. 1/2

Most live in a house they (or family members) own with an average (mean) of 2.7 persons of which 0.9 are dependents. No one community type (rural, urban, suburban) clearly predominates, although the majority live in a community with less than 50,000 inhabitants.

This profile is a description of all member crafts-artists based on summary characteristics. It masks considerable diversity of characteristics when individual demographic subgroups are examined by the primary media in which they work, their level of crafts involvement and the amount of crafts-related net income they received in the 12 months prior to completing the questionnaire.

For example, women are not predominant among crafts-artists working in wood of whom only 19 percent are female. On the other hand,



<sup>1/</sup>Professions include, but are not limited to, accountants, artists, the clergy, dentists, physicians, registered nurses, engineers, lawyers, librarians, teachers, writers, scientists, social workers, actors or actresses.

97 percent of the fiber crafts-artists are female. Glass or other media and metal (including jewelry) each attract about 50 percent female, participation, while clay and leather or paper work have about 73 percent females (Table III.1). There is less variation by level of involvement, although male crafts-artists outnumber females (52 to 48 percent) among those who report crafts is their major activity and spend at least 40 hours per week doing crafts work. This level of involvement is also reflected in the net crafts income variable; net gains of \$1,000 or more are more frequently reported by males compared to gains of less than \$1,000 or losses of \$500 or more and less than \$500 (Table III.1).

Age also varies widely by media and level of involvement. Crafts-artists engaging in woodworking have a higher median 1/age (53 years) than any other media type. Metal and paper or leather have the youngest median age (about 40 years). Comparing median ages and their underlying age distributions by level of involvement reveals younger median ages (33 to 42 years) among those who listed crafts as their primary or secondary activity (Table III.1). Those who report crafts as a leisure or occasional activity are older, 48 to 58 years. Those reporting \$1,000 or more net income from crafts activities are younger, on the average, than crafts-artists with smaller gains or losses.

Crafts is an activity engaged in predominantly by whites, at least among those who are members of crafts organizations. With the exceptions of leather or paper and glass or other media, at least 98

ERIC Full Text Provided by ERIC

48

 $<sup>\</sup>frac{1}{}$  The median is the number which splits the entire distribution exactly in half with an equal number above and below the median.

TABLE III.1

DEMOGRAPHIC CHARACTERISTICS OF CRAFTS-ARTISTS
SURVEY OF CRAFTS-ARTISTS, 1980

<i>T</i>			Pric	mary Cra	ft Hed	10			Inte	neity of C	refte Inv	<u>Lyesent</u>	:		In	come f	rom Crat	fto_
									Acti yi	ty	Leieure /				Net Lo	o of	Not Go	in of
	All Crafts- Artists	Fiber	Cley	Matal	Wood		Paper, Leather	Under 40 Hrs.	40 Hre. Plue	2nd Activity	10 Hrs. or less	10 Hre. Plus	Occa- mional Activity	Other	\$500 stue	Under \$500	Under \$1000	\$1000 ptus
Percentage of female crafts-artists Number of Respondents	70.2 2632	96.5 967	72.7 <b>436</b>	52.0 423	19.2 364	505 205	73.8 140	71.9 439	47 .8 255	, 71 <b>.3</b> 431	70.6 389	73.1 306	68.8 298	73.7 327	65.1 460	76.8 658	70.8 382	57.0 475
Age distribution Less then 25 years old 25 to 34 years old 35 to 44 years old 45 to 54 years old 55 to 64 years old	3.4 24.8 20.3 17.8 18.4	2.9 22.1 20.9 17.8 18.3	1.5 30.7 25.7 10.2	4.4 34.6 19.5 15.5	2.0 16.0 18.0 16.8 26.4	4.7 28.1 16.0 15.5 22.8	8.3 31.0 -20.5 16.9	1.8 36.4 24.1 14.3	1.9 40.7 - 25.1 13.9 12.7	3.1 29.3 -26.7 16.2 17.3	4.3 21.0 18.9 23.4 17.9 14.5	0.4 13.5 10.1 18.2 31.1 28.7	4.9 19.5 16.3 19.8 21.7	8.6 28.5 24.2 18.8 18.6	2.5 29.9 21.1 18.8 21.7 5.9	5.3 20.1 16.5 17.3 22.2	3.8 20.3 20.4 19.4 15.5 20.8	0.8 38.4 25.1 14.2 13.0
→ 65 and over → TOTAL → Number of respondents → Median age	15.0 100.0 2611 45.2	10.0 100.0 963 47.3	100.0 484 41.9	13.8 100.0 417 40.6	20.8 100.0 358 53.3	13.1 100.0 201 45.8	12.4 100.0 139 40.2	11.1 100.0 438 40.0	5.7 100.0 253 37.9	100.0 427 41.0	100.0 366 47.5	100.0 302 50.2	160.0 297 49.7	100.0 325 48.2		190.0 854 49.7	100.0 378 47.8	100.0 474 39.3
Marital status Never merried Married Widowed/divorced/	13.5 71.6	10.8 72.5	13.7 70.8	18.7 71.4	8.4 81.5	20.5 66.2	24.8 59.2	13.8 70.4	19.3 65.8	12.7 75.1	9.9 77.3	6.0 75.8	19.0 68.8	15.4 85.4	13.9 72.8	12.9 73.1	10.2 74.2	15.0 70.1
esperated TOTAL Number of respondents	14,9 100.0 2622	16.7 100.0 967	15.7 100.0 434	11.9 100.0 418	10.1 100.0 362	13,3 100.0 202	16.0 100.0 140	15.9 100.0 439	14.9 100.0 252	12.2 100.0 428	12.0 100.0 370	19,2 100,0 306	12.2 100.0 298	18.2 100.0 325		14.0 100.0 858*	15.6 100.0 382*	14.7 100.0 474
Rece White Black American indian/	97.8 0.5	98.6 0.7	98.3 0.8	98.1 0.1	98.5	95.0 0.2	91 <b>.</b> 0	98.4	89.1 0.2	97.2 0.2	98.8 0.5	96.5 1.2	98.1 0.7	98.2 0.3	97.1 0.6	98.7 0.6	98.2	98.1 0.2
Aleeken netive Asien/Pecific	1.2	0.2	0.7	0.3	1.5	4.3	8.4	0.3	0.2	2.6	0.7	1,5	0.7	0.9	1.5	0.4	1.2	1.4
islander TOTAL Number of respondents	0.4 100.0 2606	0.3 100.0 960	0,2 100.0 432	1,5 100.0 419	100.0 359	0,5 100.0 202	100.0 138	0,3 100,0 438*	0,5 100.0 252*	100.0 422*	100.0 368*	0.8 100.0 304*	0.5 100.0 291*	0.5 100.0 326*	0.8 100.0 457*	0,3 100.0 655*	0.3 100.0 379*	0.3 100.0 472
Parcentage of crafts- artists of Hispanic origin or background Number of respondents	1.3 2533	0 <b>.</b> 9'	1.8 <sup>4</sup> 418	3.3* 407	2.2° 352	• 1.3°	0.9* 134	1.0* 433	2.4° 242	* 1.3* 410	1.6* 359	1.4* 297	0.2* 284	1.6* 311	⁴1.3⁴ 445	0.4 <sup>4</sup> 638	2.0* 371	1 .a 460

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Besed on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts-artists by count larger than can be accounted for by chance except those percentages or percentage distributions followed by an estarisk (\*).

50

percent of the crafts-artists are white. Over 8 percent of paper or leather workers are American Indians or Alaskan Natives, as are about 4 percent of the crafts-artists in glass or other media (Table III.1). About 1 percent of all crafts-artists are of Hispanic origin or background.

The major variation in education level is for crafts-artists working in wood who exhibit smaller percentages with post-high school formal education than other media groups. The individual components of level of involvement in crafts and net income from crafts activities do not show much variation; there is a pattern toward higher educational attainment levels among those for whom crafts is a major secondary activity, but it is not strongly pronounced (Table III.2). About 7 percent of all crafts-artists used the G.I. Bill to attend school or finish their degree.

Fine and applied arts is the largest education major for crafts-artists attending post-secondary schools (29 to 50 percent) for all media categories of crafts-artists except wood. Crafts-artists with wood as a primary medium list engineering and "other" as most prevalent at about 18 percent each; fine and applied arts is next at 15 percent (Table III.3).

Full-time employment (30 or more hours per week) is the labor force status category with the largest share of crafts-artists for all of the media types, levels of involvement and net income categories (Table III.4). However, among the media categories, fiber and wood engaged significant percentages of full-time homemakers and the retired, 31 and 26 percent respectively. A higher percentage of those who consider crafts their major activity are employed compared to crafts-artists assessing their involvement as secondary, leisurely or occasional. In

TABLE III.2

EDUCATIONAL CHARACTERISTICS OF CRAFTS#ARTISTS
SURVEY OF CRAFTS-ARTISTS, 1980

9				<del></del>	Pri	mary Cra	ft Med	10			<b>A</b> nt∎	neity of C	refte Inv	o L yeman	t		In	come f	rom Cre	ifte
•	4		ALL					- · · · · ·		Under	Activi 40		Leieure	10	Y Occe		Net Lo	<u>ee of</u> Under	Net Ge	in of
			Crefts- Artists	Fiber	CLey	Metel	Wood		Paper, Leather	40 Hre.	Hre. Plue	2nd Activity	10 Hrs. or Less	Hre. Plue	Activity	<u>Other</u>	_			
Highest year o	f school	_																•	,	
ing completes				٠,		4 0	0.3	7 0	5.5	2.2	1.9	3.1	3.5	7.3	3.8	5.1	4.7	4.1	7.7	1.3
Some high		7	4.6	4.6	0.9	1.9	9.3	7.8 10.9	9.3	9.6	7.9	8.8	15.3	19.2	8.1	11.4	7.3		11.6	10.9
High achoo		j	12.0	12.2	5.3	11.9	21.0		22.7	21.5	21.0	13.6	14.3	13.6	13.5	13.0	11.8	13.9	15.4	
Some colleg			1F .1	14.5	14.4	13.0	18.9	18.4	22./	21.0	21.0	13.0	17.5	10.0						•
Two-year c	ollege	4							7.0		4.0	5.3	4.9	6.5	9.0	2.9	5.8	5.5	8.5	4.6
degree		-	5.8	6.6	4.6	4.5	5.1	8.2	7.6	5.5	4.0	0.3	₹.5	0.0	3.0	L.10	0.0	0.0	•••	•
Four-year	college		_ ^~		•			00.0	04.0	28.5	28.0	31.5	21.3	18.8	26.7	23.5	23.4	27.2	25.4	31.3
degree			25.1	27.1	28.4	26.0	15.0	26.3	21.3	28.5	20.0	31.0	E1.5	10.0	20.7			,_		•
Technicel/								4 =		4.0	7.3	4.6	7.0	4.1	2.3	3.9	5.4	3.8	5.3	5.5
tional d	_		4.8	4.0	3.8	5.5	7.9	4.5	5.5	4.8	8.0	11.9	12.5	10.1	15.5	14.7	15.1		9.4	11.
Greduete w	ork		11.8	13.6	14.0	12.5	8.3	6.5	6.1	9.6			19.0	10.7	12.9	22.3	18.5	13.9	13.8	16.
Mestere de			15.8	14.2	25.4	19.3	9.3	12.3	21.2	16.3	16.2	<sub>,</sub> 19.1	15.0	10.7	1240	EE . 0				
Doctorete/	Profes-							4.0			0.0	1.6	6 .1	2.2	7.5	3.2	7.0	3.6	1.7	1:0
eionel d	egree		3.2	1.8		4.6	5.2		0.6	1.1	2.3			1,5	0.7		1.0	0.9	1.2	0.
Other			0,8	1.4	0.6	0,8		1.1	0.2	0,9	2.9 100.0	0,2 100.0	1,2 100.0	100.0	100.0	100.0	100.0		100.0	100
TOTAL			100.0		100.0	100.0	100.0	100.0	100.0	100.0 435	255	430	369	306	298	327	459	656	381	474
Number of resp	ondents		2626	964	435	422	362	203	140	430	€00	430	305	300	£30	<i>0</i> 2/			23.	

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Based on calculation of  $X^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table vary from the percentages for all crefts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exterisk  $\{*\}$ .



TABLE III,3

EDUCATIONAL MAJOR OF CRAFT-ARTISTS
-ATTENDING POST-SECONDARY SCHOOLS
SURVEY OF CRAFTS-ARTISTS, 1980

·				Primery Co	reft Medie		
	All Grafta- Artiata	Fiber	Cley	Hete;	Wood	Gleee, Other	Peper, Leether
Major field of study:		1					
Agriculture	0.7	0.4	0.1	2.0	1.7	1.1	
Archi tecture	1.7	1.2	1.0	2.9	2.8	-	3.6
Aree etudies	0.2	0.3	0.1		0.3	0.4	
Biological actances	1.8	2.7	1.3	1.5	1.1	0.7	1.5
Business and management	8.7	7.2	5.2	3.7	9.4	8.0	8.0
Communications	0.4	0.3	0.2	0.8	1.3	0.7	
Computer end information	- •	•	-				
ectionces	0.1 *	0.1				0.3	
Education	14.8	18.4	13.7	9.3	11.1	13.0	18.8
Engineering	3.8	0.4	1.3	5.5	18.3	5.0	1.5
Fine end applied erts	34.1	28.4	50.4	41.8	14.5	28.8	41.4
Foreign Lenguegee	1.2	1.8	0.2	0.7	1.8	1.8	0.3
Heelth professions	4.7	5.9	2.7	2.5	4.4	4.4	3.3
Home aconomics	8.2	4,1	0.8	1.1			3.3
Law	0.4	<u></u>		0.1	0.5	4.7	_
Letters	3.5	4.9	2.8	2.6	0.8	2.8	2.4
Library eciance	0.3	0.4	0.7			`	0.3
	1.0	1.4		1.8		1.4	3.3
Hathametice	1.0	<u>'</u>					
Physical sciences	1.8	2.0	1.1	1.4	3.6	3.1	1.5
Psychology Public effeire	2.1	2.8	1.0	2.7	2.3	2.7	_
	0.7	0.9		2.3	0.7		0.3
Societ eciences	5.2	5.4	2.8	6.3	3.8	8.5	5.7
TheoLogy	0,2	0	E . D	. 0.0	0.0	0.0	
Interdisciplinary	0.5		***		3.1		
etudiee	0.5			44.4	18.7	-14.5	3.8
Other	_12,0	10,8	13,7	11.4	_10,7	17.0	
TOTAL .	ن. 100	100.0	100.0	100.0	100.0	100.0	100.0
Number of respondents	2027	782	378	337	218	149	104

SOURCE: Survey of Crefts-Artists who are Members of Crefts Organizations, Methametics Policy Research, Inc.

Based on celculation of  $x^2$  etatistics, in 98 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exterisk  $\{*\}$ .

وتعطي

ERIC

TABLE III.4

LABOR FORCE AND EMPLOYMENT CHARACTERISTICS OF CRAFTS-ARTISTS
SURVEY OF CRAFTS-ARTISTS, 1880

			Pri	mery Cr	eft Hed	10			Inte	neity of C	refte Inv	olvenen	<u>t</u>		In	come f	rom Cre	fte
	ALL					Ì		<u>Mejor</u> Under	Activi 40	ty	Leieure	Activit	Y Occa		Net Lo	oe of	Not Ge	in of
	Crefts- Artists	Fiber	Cley	Metel	Wood		Peper, Leether	40	Hrs. Plue	2nd Activity	10 Hre. or Lees	Hre. Plus	eionel Activity	Other	\$500 ptue	Under \$500	Under \$1000	\$1000 ptu
Lebor force etetue:											1							
Employed, full time	45.5	32.1	52.7	53.9	58.1	49.2	56.8	<b>57 .</b> 7	77.2	47 .1	43.7	20.0	44.8	38.3	53.5	33.6	34.8	85.6
Employed, part time	13.4	15.3	14.5	14.8	7.1	10.6	11.7	16.9	5.7	20.2	10.9	11.6	8.7	15 <b>.5</b>	15.1	12.2	15.8	13.8
Full-time homemeker	18.9	31.1	16.7	11.0	3.5	12.6	9.9	10.B	4.7	21.2	18.3	28.0	20.2	18.2	15.9	27.0	22.7	8.0
Full-time etudent	2.8	2.4	2.7	5.7	0.7	4.0	5.0	1.3	1.1	1.0	3.0	0.1	1.1	12.4	4.1	4.0	3.0	1.8
Retired	13.8	13.1	8.4	10.6	26.0	18.9	10.8	9.8	4.5	6.0	16.7	27.4	17.7	9.1	5.2	10.5	20.0	8.3
Unemptoyed	5.8	2.7	3.5	1.7	2.1	1.5	4.8	2.0	3.7	1.2	3.8	2.8	3.9	2.8	5.3	3.5	2.3	2.1
Other, not working	2,7	3,3	3,5	2.2	1.5	3,2	1.0	2.3	3.1	2,8	2.5	2.3	3.6	2.7	3.0	1.2	1,6	2.4
TOTAL	100.0		100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0 326	100.0 460	100.0	100.0 382	100.0 473
Number of respondents	2628	966	436	422	364	201	140	<b>43</b> 8	254	431	370	306	298	320	400	650	302	4/3
Percentage reporting crefts so their main									~				ì				* * *	
occupations	33.8	27.2	47 .4	41 .4	25.5	47.2	21.7	88.2	96.6	13.8	0.5	10.0	3.5	16.B	26 .8	8.6	26.3	69.6
Number of employed																		
respondents	1641	492	314	307	240	128	83	341 .	218	301	505	118	167	197	322	318	208	394
Non-creft occupation of																		
those with full- or																		
part-time employment: Clarical	7.2	10.7	6.6	4.3	2.1	5.0	8.7			8.2	8.2	5.3	8.2	7.3	6.1	9.6	5.5	2.9
Skilled worker	6.6	0.8	2.8	11.8	13.8	11.3	7.8	B.1		7.8	6.8	8.3	4.3	5.6	7.3	5.0	9.6	8.5
Fermer, form meneger	1.3	0.4		1.8	2.7	-	4.3		12.5	0.4	1.4	4.1	0.5		0.4	0.3	1.5	2.7
Laborer	1.3	1.7	1.4	0.6	1.4	_	1.2	7.9		0.8		4.1	1.1	0.3	1.7	2.0		1.1
Managar, administra-			•••	- •			,					-		1				
tor	8.7	8.4	10.4	3.7	8.4	5.3	14.1	3.8		8.7	5.3	11.1	14.6	В.3	8.9	8.8	7.1	13.6
Militory	0.5	_	0.2	_	2.1		9.6			0.3	1.3	0.3	0.5	0.2	0.1	1.3	0.3	0.3
Operative	1.1		_	1.6	4.1	1.5			3.5	1.6	0.6	_		0.5	2.3	0.5	1.2	
Professional	52.2	52.0	62.3	57.1	48 <b>.</b> 8	54.0	48.7	50.8	85.4	44.2	55.B	34.3	57.5	67.8	57.2	52.5	53.5	51.0
Proprietor, owner	6.5	8.0	4.2	8.5	8.2	7.7	2.0	8.0	11.3	10.9	2.3	12.0	3.7	3.1	2.8	4.0	5.8	9.2
Selee	5.1	7.5	3.1	2.8	4.B	8.8	2.4		7.3	3.4	5.8	8.3	4.9	3.8	5.8	4.8	2.8	2.4
Service worker	4.4	6.2	3.2	4.1	4.8		1.8	14.1		1.5	0.8	0.6	0.7	3.1	1.4	4.1	3.4	3.4
Technice L	4.B	2.8	5.7	5.5	5.3	6.3	8.3	7.2		6.8	8,2	5.5	1.1	i.9	4.5 0.5	5.8	7.4 1.8	5.2
Other	0.3	0,4	100.0	400 0	0,6 100.0	100.0	100.0	100.0	100.0	<u>0.4</u> 100.0	0,6 100.0	1.1	100.0	100.0		100.0	100.0	100.0
TOTAL	100.0 910	100.0 311	138	100.0 157	145	100.0 55	100.0 61	100.0 28	100.0	100.0 250	163	97	141	143	188*	243*	140*	100
Number of respondents	ยาบ	311	138	10/	140	00	01	<b>4</b> 0	8	200	103	8/	141	୍ଥ । କଧ	100	240	140	101

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Besed on calculation of  $x^2$  etatistics, in 98 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by wrisk  $\{^*\}$ .

56

over \$1,000 are more likely to be employed than those with smaller gains or losses (Table III.4). It is these same components—those who report crafts as a major activity, and receive \$1,000 plus net income—that also list crafts as their main occupation.

For those crafts-artists who do not consider crafts as their main occupation, the most prevalent occupation reported is professional, ranging from 34 to 68 percent of the crafts-artists in each category. Other occupations with significant representation include skilled workers, managers and administrators, and proprietors and owners (Table III.4). This tendency toward highly skilled occupations reinforces the findings based on the educational attainment characteristics.

Crafts-artists whose homes are owned outnumber renters by a ratio of about 4 to 1 for all components (Table III.5). There is an average of 2.7 persons per household and 0.9 dependents with little variation by type of media, level of crafts involvement or net income from crafts.

There is more variation in the type of area where crafts-artists live. Among the media components, suburban areas \( \frac{1}{2} \) are more prevalent for fiber, clay and paper or leather; rural areas are more popular for those who work in metal, wood and glass and other media (Table III.5). Crafts-artists reporting crafts as a major activity live in rural settings whereas those with less intense involvement lean toward suburban areas. Crafts-artists with net crafts income of \$1,000 or more live



 $<sup>\</sup>frac{1}{2}$  Suburban, rural and urban were defined by respondents.

TABLE III.5

HOUSEHOLD AND LOCATION CHARACTERISTICS OF CRAFTS-ARTISTS
SURVEY OF CRAFTS-ARTISTS, 1980

distribution (the plan she for gauges and paythers. In the consistent			Pri	mary Cr	oft Med	ia			Inte	nsity of (	Crafts Inv	olvemen	ıt		In	come f	rom Cra	fts
									Activi	ty	Leisure				Net Lo	ss of	Net Ga	in of_
	All Crafts- Artists	<u>Fiber</u>	Clay	Metal	Wood		Paper, Leether	Under 40 Hrs.	40 Hrs. Plus	2nd Activity	10 Hrs. or less	10 Hrs. Plus	Occa- sional Activity	Other	\$500 plus	Under \$500	Under \$1000	\$1000 <u>plus</u>
Parcantage of crafts- artista who: Own Rant TOTAL	81.7 18.3 100.0		80.5 19.5 100.0	76.6 23.4 100.0	85.4 14.6 100.0	77.2 22.8 100.0	73.2 26.8 100.0	77.8 22.2 100.0	67.4 32.6 100.0	78.8 21.2 100.0	89.0 11.0 100.0	87.0 13.0 100.0	83.2 16.8 100.0	78.9 21.1 100.0	80.6 19.4 100.0 454*	83.9 16.1 100.0 644*	81.4 18.6 100.0 375*	78.2 21.8 100.0 473*
Number of respondents	2569	952	434	414	356	500	135	434	253	426	366	303	291	320	454*	644*	3/5*	4/3
Average (mean) number of persons in house— hold Number of respondents	2.7 2614	2.7 961	2.6 433	2.6 421	2.7 362	2.7 202	2.5 138	2.7 438	2.6 253	2.8 427	2.9 369	2.6 302	2.5 295	2.5 238	2.7 457	2.6 656	2.7 380	2.8 474
Average (mean) number of dependent persons in household Number of respondents	0.9 2581	0.8 945	0.8 430	1.0 413	1.3 360	1.0 200	0.6 137	0.9 433	1.1 249	1.0 <b>4</b> 18	1.1 365	0.8 303	0.9 291	0.8 236	1.1 453	0.8 653	0.9 377	1.1 467
Type of community: Rural Urban Suburban TOTAL Numbar of respondents	34.2 28.6 37.2 100.0 2618	29.5 30.3 40.2 100.0 962	32.0 29.7 38.3 100.0 433	36.1 29.5 <u>34.4</u> 100.0 421	42.2 24.5 33.3 100.0 363	44.2 25.6 30.2 100.0 202	29.9 26.3 43.8 100.0	45.2 25.4 29.4 100.0 438	51.9 24.4 23.7 100.0 254	32.2 26.6 41.2 100.0 428	27.7 26.2 46.1 100.0 369	35.1 27.2 37.7 100.0 305	27.8 34.4 37.8 100.0 294	26.4 32.8 40.8 100.0 326	29.6 32.2 38.2 100.0 459	25.8 32.4 41.8 100.0 658	35.0 29.0 36.0 100.0 379	42.9 26.7 30.4 100.0 473
Population size of community: Less than 5,000 5,000 to 49,999 50,000 to 499,999 500,000 or more TOTAL	26.8 30.5 10.1 15.6 17.0 100.0 2596	21.8 32.8 10.7 16.5 18.2 100.0 955	27.0 10.9 18.1 18.8	27.8 27.6 9.7 14.5 20.4 100.0 418	35.8 31.9 8.2 13.7 10.4 100.0 363	34.2 29.3 10.9 13.4 12.1 100.0	27.6 28.6 10.4 17.5 15.9 100.0	35.7 26.3 5.2 17.0 15.8 100.0	39.2 27.3 10.6 10.7 12.2 100.0 250	23.5 32.3 11.3 14.5 18.4 100.0	21.3 39.5 10.3 14.9 14.0 100.0 368	27.2 30.9 13.2 15.9 12.8 100.0	27.3 22.8 10.2 17.5 22,2 100.0	18.5 31.7 11.5 17.3 21.0 100.0 322	26.2 25.2 10.7 17.7 20.2 100.0 456	19.5 32.5 11.8 16.9 19.3 100.0 656	29.7 33.7 10.6 14.7 11.3 100.0	34.9 22.9 9.3 15.3 17.6 100.0 470

SOURCE: Survey of Crafts-Artists who are Members of Crefts Organizations, Mathemetica Policy Research, Inc.

Based on calculation of  $X^2$  statistics, in 99 out of 100 cases, the weighted parcentages in this table vary from the parcentages for all crafts-artists by an amount larger than can be accounted for by chance excapt those percentages or percentage distributions followed by an asterisk  $\{*\}$ .



59

in rural areas to a greater extent than do those with lower gains or losses. In all of the component subgroups, urban consistently accounts for a one-quarter to one-third share of crafts-artists (Table III.5).

Categorizing places of residence by population size reinforces the pattern toward low density areas. There is some fluctuation between the relative positions of communities with less than 5,000 population and those with 5,000 to 49,999. One of these two is always the first or second largest percentage and, in all but one case (the "other" level of involvement), whichever is not first is second.

# B. COMPARISON OF CRAFTS-ARTISTS TO THE U.S. POPULATION

crafts-artists who are members of crafts organizations are a small subset of the U.S. population. There were an estimated 150 to 180 thousand of these crafts-artists in 1980, while over 226 million individuals, of whom 154 million were age 20 and older, were enumerated in the 1980 Decennial Census. If Given the small numbers of crafts-artists relative to the total population, one would expect to find some divergence in characteristics. This section compares characteristics of member crafts-artists to corresponding characteristics of the adult population to ascertain the extent of similarities and differences. The adult population (defined as 20 years and over, or, in some instances, 25 years and over) is the basis for comparison because it would be highly

<sup>1/</sup>U.S. Bureau of the Census, <u>Population Profile of the United</u>
States: 1980, Current Population Reports, Series P-20, No. 363
(Washington, D.C., 1981), Table 4.

unlikely that younger persons would belong to a membership organization of practicing crafts-artists.  $\frac{1}{2}$ 

Females are much more prevalent among crafts-artists than they are in the U.S. population aged 20 years or older, comprising over 70 percent of crafts-artists compared to less than 53 percent of all adults. The crafts-artist population has a larger proportion aged 25 to 64 and smaller proportions either under 25 or 65 and over compared to the total adult population, but the median ages for both groups are similar -- about 45 years for crafts-artists and 42 years for all adults. Blacks and Asian and Pacific Islanders are not attracted to crafts work (or at least membership in crafts organizations) to as great an extent as are Whites and American Indians and Alaskan Natives. Persons of Hispanic origin are also not found as crafts-artists in the same proportion as they are in the total adult population. The marital status of crafts-artists is roughly the same as for all adults; married individuals are slightly more prevalent among the former group; those who are widowed, divorced or separated or never married are somewhat more frequent among the latter (Table III.6)

In comparing economic characteristics, there are more revealing differences. Crafts-artists are much better educated than the entire population aged 25 years or older. Over 56 percent of crafts-artists have at least four years of college, whereas only 17 percent of all



61

 $<sup>\</sup>frac{1}{4}$  An even more appropriate comparison group, especially for hose crafts-artists listing crafts as their primary activity, would be the self-employed population. However, the published tabulations from the continuing series of national household surveys that is the basis for this section do not show distributions of other characteristics for the self-employed.

TABLE III.6
CHARACTERISTICS OF CRAFTS-ARTISTS AND THE U.S. POPULATION, 1980

Population Characteristic	1980 Crafts <b>-A</b> rtists	1980 Total Population <u>a</u> /
Sex of the population 20 years old and older:		·
Percentage female	70.2%	52.6%
Age (both sexes) of the population 20 and over	<u>b</u> /	
20 to 24 years old	3.4	13.8
25 to 34 years	24.8	24.1
35 to 44 years	20.3	16.6
45 to 54 years	17.8	14.8
55 to 64 years	18.4	14.1
65 and over	15.0	16.6
o) and over	15.0	10.0
Total	100.0	100.0
Median age	45.2	42.3
Race of the population 20 and over:		
White	97.9	85.2
Black	0.5	10.3
American Indian, Alaskan Native	1.2	0.5
Asian, Pacific Islander	0.4	1.5
Other		2.5
Total	100.0	100.0
10021	10000	20070
Hispanic Origin or Background of the populati 20 and over:	lon	
Percentage	1.3	5.4
Marital Status of the population 20 and over:	:	
Never married	13.5	16.3
Married	71.6	66.1
Widowed, Divorced, Separated	14.9	<u>17.6</u>
Total	100.0	100.0
		(continued)

ERIC

TABLE III.6 (continued)

Population Characteristic	1980 Crafts-Artists	1980 Total Population
Years of School Completed for the population	,	
25 and over:		
Less than 4 years of high school	4.6%	31.3%
Four years of high school	12.0	36.8
One to three years of college	26.7	14.9
Four years or more of college	<u>56.7</u>	<u>17.0</u>
Total	100.0	100.0
Labor Force Status for the civilian non-institutionalized population 20 and over:		
In labor force (percentage	61.7	64.6
of population)	61.7	04.0
Employed (percentage of	(95.4)	93.9
labor force):	•	93.9
Full-time (30 plus hours)	73.7	
Part-time	21.7	6.1
Unemployed	4.5	0.1
Not in labor force (percentage of	(20.2)	35.4
population):	(38.3)	33.4
Student	2.8	
Retired	13.9	
Other including homemaker	21.6	
Occupation of Employed Workers 18 and over:		
Crafts	33.6	NA
Professional, technical	37.8	15.8
Manager, administrator, owner	10.1	10.7
Sales	<b>3.4</b> <sub>1</sub>	6.1
Clerical	4.8	18.6
Skilled	4.4	12.9
Operatives, including transport	0.7_	14.8
Laborers	0.9	6.0
Farm managers	0.9	1.4
Service, others	3.4	<u>13.7</u>
Total	100.0	100.0

(continued)



# TABLE III.6 (continued)

Pepulation Characteristic	1980 Crafts-Artists	1980 Total Population
Income:		
Mean annual per capita income of		1/
population 20 and oyer	(\$9,462)	\$12,217 <u>d</u> /
Net crafts income <sup>C</sup>	\$1,454	NA
Other sources	8.008	NA .
Mean annual household income	28,886	21,366ª/
Estimated bases for characteristics (in thousands)		
Crafts-artists who are members of		
craft organizations	150-180	
Persons 20 and over in March 1980		154,061

SOURCES: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc. 1980; U.S. Bureau of the Census, Profile of the United States: 1980, Current Population Reports, Series P-20, No. 363; U.S. Bureau of the Census, Marital Status and Living Arrangements: March 1980, CPR, Series P-20, No. 365: U.S. Bureau of the Census, Money Income of Families and Persons in the United States: 1979, CPR, Series P-60, No. 129; U.S. Bureau of the Census, Money Income of Households in the United States: 1979, CPR Series P-60, No. 126.

- ( ) = The sum of amounts shown for the more detailed categories.NA = Information not available.
- $\frac{a}{}$  Data are either for April 1, 1980, from the 1980 Decennial Census or for March 1980 from the March 1980 Current Population Survey.
- $\frac{b}{L}$ Less than 1 percent of crafts-artists in the survey reported their age as under 20.
- $\frac{c}{}$  Includes income from sales of crafts work, other crafts-related sales and teaching crafts, less crafts-related expenses, in the 12 months preceding the interview.
- $\frac{d}{R}$  Represents income for calendar year 1979, adjusted for change in Consumer Price Index, all commodities, of 8.9 percent from January to September 1980.

adults have acquired as much formal education. Roughly comparable percentages of crafts-artists and the adult population aged 20 years or older--about 62 percent of the former and 65 percent of the latter--are estimated to be in the labor force (either employed or available for work). About 95 percent of crafts-artists in the labor force listed their status as employed versus 5 percent who were unemployed; the figures for the adult population are 94 percent employed versus 6 percent unemployed (Table III.6).

Of those who are employed, crafts work or professional or technical occupations are much more frequent among crafts-artists than all adults. Over 71 percent of crafts-artists list one of these categories as their occupation (with somewhat more listing professional or technical occupations compared to crafts work), while 16 percent of the employed civilian population aged 18 or older are professional or technical workers (Table III.6).

Average (mean) per capita income of crafts-artists (including net income from crafts and all other income) is about 22 percent lower than mean per capita income for all adults aged 70 or older. (Income for crafts-artists was ascertained for the 12 months preceding the date of interview which occurred in the fall of 1980; income for all adults was measured for calendar year 1979 with an adjustment made for inflation.) Most of the income of crafts-artists is from non-craft sources-\$8,008 out of \$9,426 (Table III.6). The crafts-artist population includes a large proportion who are self-employed (at least the 34 percent of employed crafts-artists who list their main occupation as crafts), and substracting expenses from gross crafts income leaves a net figure on

ERIC Full Text Provided by ERIC

111.15 65

average of only \$345 compared to average crafts gross sales receipts of \$2,493 (Table IV.21). In contrast, less than 3 percent of all employed adults aged 18 or older are self-employed, so that effects of business expenses on average per capita income are quite small.  $\frac{1}{2}$ 

In contrast, mean household income for crafts-artists is about 35 percent higher than mean household income for the total U.S.--\$28,886 compared to \$21,366 (Table III.6). The two sets of findings for personal and household income suggest that, while crafts-artists may make a smaller contribution to household income than adults on average, they are resident in households with higher than average income consistent with the higher educational and skill levels exhibited by crafts-artists compared to the general adult population.

In summary, crafts-artists in the U.S. differ from the total adult population on several dimensions, most notably in their higher proportions of females, Whites, persons with a college education, professional and technical workers, and high-income households. They also look remarkably similar to the total adult population in many respects, including age, marital status, and labor force participation. In some sense, then, there may be a little crafts-artist in all Americans, and the more successful crafts-artists are those who have made the effort to develop specific skills as a result of training or who have been endowed with a particular talent.

<sup>1/</sup>U.S. Bureau of the Census, Money Income of Families and Persons in the United States: 1979, Current Population Reports, Series P-60, No. 129, Table 59.

### IV. CRAFTS-RELATED CHARACTERISTICS

The media crafts-artists engage in, their preparation, their level of involvement, what they view as problems, barriers and satisfactions, the extent to which they exhibit and sell their work and the financial result are the attributes examined in this section to increase understanding of the people who engage in crafts. The emphasis is on exploring the major patterns adopted by crafts-artists in general and then highlighting subgroups which illustrate divergence in characteristics or attitudes. Accordingly, the section is descriptive, but with forays into more comprehensive understandings of crafts activities when motives and attitudes appear relatively unambiguous. Further examination of the data in Section V concentrates more exclusively on analytic concerns; this initial investigation seeks to get more acquainted with crafts-artists.

# A. GENERAL CRAFTS-RELATED CHARACTERISTICS

One of the characteristics which differentiates groups of craftsartists is the media in which they work. The range of work is very
broad—between arts on the one hard and activities that lack elements
of individual creativity on the other. It is the element of creativity
applied to each work that differentiates crafts from manufacturing.

Media combined with creativity provides a convenient operational definition. Those who "create" with certain materials are crafts—artists.

The media considered are fiber, clay, leather, paper, glass, metal, wood
and all others.



67

Fiber is the predominant craft listed as the primary media for 42 percent of all crafts-artists. Clay and wood (17 percent each), followed by metal (13 percent), are next most popular with glass, leather, paper and other media and combinations each accounting for less than 4 percent of crafts-artists belonging to crafts membership organizations (Table IV.1).

The training received in crafts by crafts-artists working in the various media is examined in more detail in a subsequent section. However, from the perspective of primary media, training by professionals or 5 or more college courses in crafts do not appear to be as necessary for fiber crafts-artists as for workers in clay or perhaps metal. Crafts-artists working in wood received informal or no crafts training while the training received by leather, paper, glass or other media crafts-artists shows no clear pattern.

Place of residence by region of the country is not a major distinguishing factor with the exception of a concentration of clay workers in the West (west of a line from eastern Montana through eastern New Mexico).

The level of involvement in crafts displays more interesting patterns. Fiber is much more of a secondary-leisure-occasional activity than are clay or metal which tend to be more important as primary (major) activities. Wood is a very prevalent activity among crafts-artists in all intensity of involvement categories; it is second after fiber in popularity among the major activity (40 or more hours per week) and the leisure and occasional subgroups. Metal work accounts for 17 percent for each of the two primary activity subgroups. Paper tends to be a secondary, leisure (10 hours or less per week) or occasional activity,

TABLE IV.1 PRIMARY CRAFTWORK MEDIA SURVEY OF CRAFTS-ARTISTS, 1980

		Ext	ent of Cr	<u>eft Traini</u>	<u> </u>		Region of	Residence	
	All Crefte-	Profee-	-	• Courses	None or	North-	South	North Central	Wee t
	Artists	eionel	5 Ptue	1 to 4	Informat		004411		
rimary media of crafts- artiats					. •			P	
Fiber	41.8	48.7	30.5	39.1	32.0	40.5	41.3	47.9	37.1
Clay	17.4	20.6	35.4	15.7	6.4	16.3	18.1	13.1	23.5
Leether	1.0	G.8	0.8	0.2	1.9	1.0	0.7	0.5 3.4	2.0 3.8
Paper	3.9	3.5	4.7	7.5	2.2	5.1 3.9	3.3 4.3	2.0	2.9
Glass	3.3	2.7	5.2 15.3	3.0 11.2	4.3 9.1	15 <b>.</b> 6	10.9	10.1	13.3
Hetel	12.5	14.1 7.9	5.3	18.9	37.2	14.8	16.4	20.2	14.7
Wood	16.6 3.5	1 7			6.9		5,2	2.8	
Other and combination OTAL	3,5 100.0	$\frac{1.7}{100.0}$	2.8 100.0	4,4 100.0	6.9 100.0	2.8 100.0	100.0	2,8 100.0	2.7 100.0
umber of respondents	2536	1348	191	393	602	900	656	509	466

			ntenel	ty of Cref	ta Involv	ement							<u>Ir</u>	1COMB 1	rom <u>Cr</u>	efte
		Helor	Activi	ty	Leteure	Activit	A		Ysar	of Inv			Net Lo	ose of	Not G	ain of
	All Smafts- Antista	Under 40	40 Hre. Pius	2nd Activity	10 Hre. or less	10 Hre. Plue	Occa- aional Activity	Other	Under / 10 Yrs,	10 to 14 Yre,	15 to 24 Yra,	25 Yra. Plua	\$500 ptue	Under \$500	Under \$1000	\$10 plus
Primary media of crafts- artists																
Fiber	41.8 17.4	36.8 22.5	16.4 31.6		48.7 7.1	52 10.	44.9 3.4	38.3 28.0	34.3 18.4	44.6 18.5	40.0 20.7	50.6 12.7	37.8 15.4		40.5 16.1	
Cley Leather	1,0	1.4	0.7	0.7	0.6	0.4	0.5	1.2	0.8	1.2	0.7	1.3	2.2	0.5	0.4	1.6
Peper	3.9	2.4 5.3	2.7 8.3	4.5 1.8	5.9 2.9	1.4 1.2	6.9 3.1	3.7 3.2	2.5 4.7	5.2 4.0	5.5 2.5	3.0 1.8	2.8 4.8	4.4 1.4	4.6 5.2	
Gles⊆ Motel	3.3 12.5	16.7	16.7		9.7	7.2	8.9	12.8	16.4	12.6	9.7	8.9	18.2		8.9	16.6
Wood	16.6	12.8	20.5		23.1	21.3	19.0	10.6	19.1	11.7	17.3	17.8	17.4	15.7	18.5	
Other and combination	3,5 100.0	2,1	3,1	3,9	<u>2,0</u> 100.0	5,4 100.0	4,2	2,2 100.0	3,8 100.0	2, <u>2</u> 100.0	3,6	4.1	1,4		4,8 100.0	
TOTAL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Number of respondents	2536	428	249	<b>4</b> 2C	362 <sup>1</sup>	296	276	310	. 922	598	491	504	444	645	372	469

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Based on calculation of X atatistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all cray'ts-artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by a saterisk [\*].

V

while leather has no predominant concentration and glass tends to be a primary activity for those working 40 or more hours a week (Table IV.1).

Some impression of the trends in popularity of individual crafts can be garnered from the years of involvement in crafts. Those who have been in crafts for less than 10 years are more likely to work in wood and metal than if they had a longer involvement. However, wood's popularity increases when the years of involvement exceed 15 while metal's continues to drop. Fiber is most popular, at 51 percent, among crafts-artists who have 25 or more years of crafts work; it is a 34 percent share for those with under 10 years of involvement.

Net income from crafts (positive as well as negative) has no predominant pattern (Table IV.1). The largest share of net losses is among fiber workers, while potters are the largest fraction of crafts-artists reporting net gains of over \$1,000. However, metal and wood are more evenly matched between net gains and losses and there is no pattern among the other crafts.

Clafts-artists were asked to categorize the primary influences on their work using responses provided in the questionnaire. Current trends in crafts is the single most prevalent response for all media, intensity of involvement and net income subgroups; it is mentioned by 33 percent of all crafts-artists. However, the two response categories, American folk themes and ethnic or cultural traditions combined, attracted a larger response, 40 percent overall, although the response to each individually is lower than for "current trends" (Table IV.2). These inherently opposed forces contribute to the variety of objects and styles produced. At the same time, other art forms and the emotions and



TABLE IV.2

INFLUENCES ON CRAFTS WORK
SURVEY OF CRAFTS-ARTISTS, 1980

			Prima	ry Craft	, Medie			1	ntensi	ty of Cre	fts Invo	Lvemen	<u>t</u>		Ir	come f	rom Cre	fte
	ALL							Major /	ctivit	<u>¥</u>	Leteur	Activ	ity Occa-	_	Not !	oss of	Not G	inin of
	Crefts-					Glass,	Papar,	40	Hrs.	2nd	10 Hrs.	Hrs.	et on	• l	\$500	Under	•	\$1000
	Artists	Fiber	Clay	Matal	Wood	Other	Leether	Hrs.	Plus	Activity	or Less	Plus	Activi	ty Other	plus	\$500	\$1000	plus
Parcentage whose work is					i i													
primarily influenced by:					1													
American folk themse	17,2	19.8	9.3	9.6	24.9	16.7	14.4	11.5	12.5	18.9	21.1	24.4	17.6	13.2	10.8	21.5	21.0	14.4
Ethnic/cultural tradi-						+												
sione	22.8	26.2	26.6	14.9	15.5	21.3	23.6	25.1*	19.1*	25.1*	24.5*	23.8*	18.9*	20.3*	21.8*	21.1*	21.9*	23.2
Current trende in																		
creft	32.5	36.4*	28.9*	26.3*	28.2*	31.7*	31 .3*	31.2*	28.9*	28.4*	33.2*	36.4*	34.4*	35.8*	32.9*	35.9*	27.3*	27.8
Other ert forme	18.6	16.3	21.6	26.7	14.5	19.5	18.1	22.1	20.7	23.8	12.5	11.5	21.9	21.6	25.3	16.5	17.2	21.1
A movement or school																		
in the ert world	4.4	4.0*	5.1*	6.1*	3.0*	5.9*	4.6*	7.5*	5.5*	5.6*	3.6*	1.9*	3.2*	4.9*	7.4*	3.2*	4.4*	4.0
Emotions, parsonal																		
experience	13.7	10.1*	20.5*	20.4*	9.9*	17.4*	18.3*	19.7*	18.4*	13 .2*	12.0*	6.0*	11.3*	14.1*	16.3*	12.0*	8.2*	16.5
Other	8.8	5.7*	10.1*	12.5*	12.7*	9.9*	10.8*	9.4*	13.8*	9.80	7.4*	8.1*	5.2*	8.5*	9.0*	6.4*	11.3*	11.8
Number of respondents	2560	942	426	349	414	197	135	433	250	425	355	300	282	322	453	635	373	469
Parcentage reporting a																		
epecific influence by:																		
Medie	5.0	3.9	4.7	4.4	6.2	12.7	4.0	4.1	7.7	4.4	6.2	1.7	5.9	4.6	5.0	5.6	2.2	5.2
Traditional mathods	28.4	32.8	25.3	20.5	26.0	24.6	28.9	21.2	21.8	29.0	35.5	36.3	26.1	26.7	20.7	31.5	33.2	24.8
Contemporary tranda	19.3	23.0	14.9	20.2	13.6	16.4	22.4	21.9	16.3	21.9	17.0	23.1	18.4	17.6	16.7	19.4	19.4	19.3
Patterns, forms, colore	16.6	16.0	14.7	21.9	20.5	13.5	10.6	18.9	17.6	17.7	12.2	15.6	17.2	16.9	21.1	14.8	17.6	18.7
Particular geographic							1											
region	2.3	2.5	2.9	1.9	2.2	1.7	2.8	3.9	0.2	3.9	2.0	2.8	1.0	1.2	3.6	1.5	2.2	2.8
Particular parson	6.8	5.9	9.1	4.7	9.0	7.6	3.5	4.9	4.9	5.3	9.5	5.4	8.8	10.1	6.3	10.0	4.7	5.7
Own ideas	21,6	15.9	28,4	26.4	22.5	23.5	27.8	25,2	31.5	17.8	17,6	15.0	22,6	22.9	26.6	17.2	20,7	23,5
TOTAL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0			00.0	100.0	100.0	100.0	100.0	100.0	100.0
Number of respondents	2129	793	356	365	277	163	112	378	213	367	287	248	220	264	386	519	320	405

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Based on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts-artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an estarisk (\*).



personal experiences of the crafts-artists increase the diversity of crafts produced, but are mentioned less often by crafts-artists.

The influence of other art forms is more important among metal crafts-artists, those who engage in crafts as a secondary activity and for those with losses in net income of more than \$500; overall it is mentioned by 19 percent of respondents. Emotions and personal experiences was reported by 14 percent of all crafts-artists and the percentage does not vary significantly by subgroup (Table IV.2).

As implied by the definition of crafts and some of the early findings, most crafts-artists (74 percent) concentrate on producing one-of-a-kind objects or prototypes (Table IV.3). However, there is some tendency for those working in clay, wood and glass or other media to make multiples of the same work. This is noticeable among crafts-artists earning net incomes in excess of \$1,000. Is it that crafts-artists whose primary activity is crafts make more money from less creative work (copying the same saleable works), or is one of the hallmarks of a successful crafts-artist the ability to create a marketable original design and then to produce multiple3 of that work? This is difficult to resolve from the tabulations at hand, but a more detailed investigation of the characteristics of crafts-artists making one-of-a-kind objects compared to those who produce multiple3 should prove very interesting.

Commissioned work is more prevalent among crafts-artists whose primary activity is crafts and who have positive net incomes--especially above \$1,000--from crafts. (As noted in Section I, the group of crafts-artists whose major activity is crafts and the group averaging net income above \$1,000 overlap to a large extent.) Crafts-artists working in clay are slightly more likely to undertake commissions (Table IV.3). Crafts-

TABLE IV.3

TYPE OF CRAFTS WORKS PRODUCED AND EXTENT OF COMMISSIONS SURVEY OF CRAFTS-ARTISTS, 1980

			, Pr	imary C	eft Me	dia		I	ncome_fr	om Creft	
								Nat L	000 Of '	Net Ge	in of
	All Crefts- Artiete	Fiber	Cley	Metal		Glees, Other	Paper, Leether	\$500 plue	Under \$500	Under \$1000	\$1000 plue
Type of worke produced: One-of-e-kind prototyped	74.2	77.3	66.7	78.1	70.2	89.5	8.08	81 .1	82.0	68.8	55.0
Multiples of the eams work TOTAL	25.8 100.0	22.7 100.0	33.3 100.0	21.9 100.0	29.8 100.0	30.4 100.0	19,2 100.0	18.9 100.0	18.0 100.0	31,2 100.0	45.0 100.0
Number of respondents	2585	952	431	418	352	500	137	454	644	380	473
Percentage doing commis- eioned creft work Number of respondents	45.8 2628	39.1 965	51.4 435	39.7 <b>424</b>	39.9 364	46.2 203	45.3 140	<b>45.4</b> <b>46</b> 0	19.6 658	60.5 382	73.7 475
Commissioned work for: Architects Interior decorators Fashion design The public Others Number of respondents	5.1 10.3 1.2 38.5 4.9 2615	4.2* 11.8 1.5* 29.5 4.1 962	14.6	4.1	14.6	6.5	4.8 • 0.7	4.6 11.5 0.5 37.0 3.6 458	0.8 2.4 0.2 16.8 1.9 655	3.8 11.5 1.4 47.9 6.9 381	13.9 20.0 3.5 57.5 7.8 472
Average [meen] number of commissioned works in lest 12 months Number of respondents	22.0 1274	49.7 333	12.1 201	17.2 246	12.5 129	17 <b>.4</b> 93	8.6 51	8.5 191	10.3 100	8.2 192	63.7 327
Average [meen] selling prices of commissioned works in lest 12 months Lowest Highest Number of respondents	\$ 163 \$ 931 1092	\$ 155 \$ 750 320	\$ 150 \$ 369 204	\$ 277 \$1420 248	\$ 110 \$1489 126	\$ 218 \$1816 92	\$ 81 \$ 382 52	\$ 73 \$ 84 89	\$ 60 \$ 184 90	\$ 68 \$ 262 191	\$ 24 \$ 138 33

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Beend on calculation of  $x^2$  etatistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esteriak  $\{*\}$ .

artists report doing commissioned work most often for the "public" or individual customers and for interior designers (37 and 10 percent of crafts-artists respectively). An average of 22 commissioned works were produced by those crafts-artists doing commissions in the 12 months prior to completing the questionnaire; fiber workers produced nearly 50 works, workers in other media subgroups produced from 9 to 17. As anticipated, those with high net incomes produce the most commissioned works and the most expensive ones.

By media type, commissions in glass or other media, wood and metal received the highest average  $\frac{1}{}$  prices (ranging from \$1,400 to \$1,800). Leather or paper followed by wood received the lowest average price (\$81 and \$110 respectively).

## B. TRAINING IN CRAFTS

To the extent crafts is an occupation, success can be hypothesized based on the quality and quantity of training received.

However, to the extent crafts involves synthesizing material and form, idea and technique, it becomes more difficult to predict success or achievement. From another perspective, the significance of training can be gauged by crafts-arrists own appreciation of its influence and whether they intend to seek additional instruction. All of these aspects of training can be examined using the information now available.

One of the ways of identifying the influence of training on crafts-artists who are members of crafts organizations is to detail the specific training crafts-artists have experienced. The possibilities

 $<sup>\</sup>frac{1}{W}$  Whenever "average" is used without modification it refers to a mean.

range from self-taught to an apprenticeship. These two are in fact the extremes, 56 percent of all crafts-artists report they are "self-taught" and 8 percent were an apprentice at some point in their crafts career (Table IV.4). 1/2 The next most frequent type of training is workshops at crafts schools or organizations, 40 percent of crafts-artists, followed by college or university courses in arts or crafts, 33 percent. From 27 to 21 percent of all crafts-artists have completed school art classes, professional workshops, lessons from family or friends and adult education courses. Private lessons from individuals (18 percent) and community or recreation programs (13 percent) complete the list of types of training (Table IV.4).

Certain media involve more formal education than others. Craftsartists working in clay, for example, depend less on their own experiences and more on apprenticeships, professional workshops and college
and university courses than workers in other media. The pattern is
somewhat similar for fiber, but in this case over half indicate they
are self-taught as well as participants in professional workshops.

Metal, glass or other media, wood and leather or paper also require some
training from more experienced crafts-artists, but a leavening of selfinstruction is evident. Wood is mostly a self-taught skill with relatively little dependence on professional level instruction.

For all crafts-artists, the average period of self-teaching is 13 years. Lessons from family or friends are reported at 8 years.

Apprenticeships lasted for about 14 months; private lessons from

 $<sup>\</sup>frac{1}{\text{Crafts-artists}}$  could report receiving training from many or even all of the sources listed in the question and the table based on that question.

TABLE IV.4

TRAINING CHARACTERISTICS OF CRAFT-ARTISTS SURVEY OF CRAFTS-ARTISTS, 1980

		Primary Craft Madia						Years of Involvement			
	All Crefts- Artists							Under	10 to	15 to	25
		Fiber	Clay	Metel	Wood	Gless, Other	Peper, Leether	10 Yrs,	14 Yrs.	24 Yre,	Yre. Plus
r <sub>c</sub>											
د	,										
ercentage of respondents a/ with past craft training									•	**	1
Lessons from family or											
friends	21.8	28.8	12.7	13.8	17.3	20.7	20.6	15.6	18.3	25.4	28.9
Self-taught	56.2	53.0	36.1	52.2	72.9	63.7	38.1	52.0	59.0	53.5	51.0
School ert cleases	26.8	27.0	36.1	25.5	15.8	20.0	36.0	22.7	26.7	33.2	29.7
Community/recreation									· •		
program	12.3	14.6	14.4	9.2	9.6	9.7	6.0	9.7	11.3	13.0	15.7
College/University	32.9	31.3	43.6	39.7	13.3	24.4	29.9	28.8	39.9	29.8	29.6
Apprenticeship	7.6	4.8	11.6	7.2	5.5	6.7	. 8.2	9.4	8.0	6.6	5.4
Adult education courses	21.2	24.6	18.2	19.4	14.0		21 .7	16.5	17.7	23.2	27.4
Workshops at craft achools/			• -				U'				
organizations	39.7	48.0	41.6	43.3	16.7	23.7	24.7	36.9	42.2	36.7	37.8
Private Lessons from pro-	00.,				•		-				
fassional	17.6	20.5	13.6	19.4	11.9	14.8	22.7	13.0	16.5	20.0	22.7
Professional workshops	24.8	26.5	32.4	29.3	9.4	17.8	19.6	21.0	29.6	26.9	21.2
lumber of respondents	2637	969	436	424	365	203	140	950	614	512	532
number of respondence	2007	000	400	76. 7							
lyerege (meen) emount of											•
training received:											t
Lessons from family or			5.0	6.1	7.7		4	3.2	5.2	8.8	14.5
friends (years)	9.2	9.5	10.9	10.9	14.9	12.6	9.9	6.1	9.4	14.8	27.1
Self-taught (years)	13.2	13.9	4.6	4.2	4.4	12.0	**************************************	3.1	4.6	5.0	6.0
School ert clesses (years)	4.6	4.6	4.0	4.2	7.7	•	. •	٠.١	7.0	-	
Community/recreation			2.0					2.9	3.0	4.4	5.4
program (courses)	3,,9	4.3	3.3	9.2	7	9.0	· ·	7.9	8.3	8.9	7.2
College/University (courses)	8.0	8.3	10.4		7	8.U	7	11.0	13.5	*	<i>'</i>
Apprenticeship (months)	13.7		12.5	15-0	*		7	2.8	3.1	3.9	5.5
Adult education (courses)	3.9	4.1	4.1	3.3	•	•	•	2.0	3.1		
Workshops at craft achools/					- 0 -			3.8	5.7	7.2	23.5
organizations	9.3	11.8	6.5	4.2	3.8	3.6	*	3.5	8.7	, . 2	EU.U
Private Lessons from pro-									2.0	2.5	3.6
fessional (years)	2,9	2.3	2.2	3.8		•	Ŧ.	1.8	2.9	7.7	12.7
Professional workshops	6.6	5.7	5.3	4.6	#	#	#	3.2	4.8	/./	12./

ERIC 78

	_		<u>intenei</u>	ty of Cref	te Involv	ement	_ <del></del>			ncome 1	from Cre	ofto
		Mejor	Activi	ty	Leieure	<u>Activit</u>	Y.		Not L	ose of	Not B	ein of
	All Crefts- Artiets	Under 40 Hre,	40 Hra. Plue	2nd Activity	10 Hre.	10 Hre. Plue	Occe- eionel Activity	Other _	\$500 plue	Under \$500	Under \$1000	\$1000 plue
Percentege of reepondente												-
with poot croft training												
Legeons from femily or	01.0	17.4	12.2	17.4	28.1	22.7	25.8	. 13.0	13.9	26.2	19.3	15.5
friends	21.8 56.2	54.8	66.U		57.4	53.7	52.7	29.2	57.0	50.5	57.2	56.8
Self-taught	26.8	27.3	24.3	30.8~~		24.7	29.3	19.8	30.0	30.9	21.7	30.0
School ert cleeses	12.3	11.4	7.7	12.4	13.1	13.3	12.2	6.7	9.8	16.0	9.0	11.0
Community/recreation program	32.8	35.6	31.4	36.4	26.6	25.5	24.0	32.0	38.0		26.2	32.4
College/University	7.6	15.9	16.6	9.6	1.7	3.5	2.6	5.1	9.4	3.6	5.5	18.1
Apprenticeehip	21.2	14.8	10.3	21.8	29.8	21.2	23.3	9.5	24.9	26.1	16.2	14.9
Adult education courses	21.2	14.8	10.3	21.0	28.0	E1.E	20.0	3.0	24.0			
Workshops at craft	39.7	38.0	34.6	45.5	36.8	37.8	29.4	28.9	47.2	40.8	33.4	37.5
echoole/organizations	38.7	36.0	34.0	40.0	٠,٠	0,.0	LU,4	20.0	47.62	*****		J
Private lessons from pro-	47.0	21.3	14.1	15.6	18.6	24.3	12.2	8.7	18.1	17.0	18.8	15.9
fessionel	17≥8 24.8	31.7	28.2	34.9	17.3	20.0	13.0	20.9	30.0	20.6	22.0	29.8
Professional workshops		439	256	431	371	307	298	328	459	659	379	478
lumber of respondents	2637	438	200	401	371	307	200	JEU	400		J, Ç	4,0
Averege [meen] amount of												
training received:												
Leasone from family or								_				
friende (yeare)	8.2	6.4	•	10.2	7.8	10.4	6.4		6.7	8.5	9.0	6.8
Self-tought (years)	13.2	10.6	12.6	11.8	12.6	18.4	16.0	10.7	12.0	14.7	13.4	11.6
School ere cleeses (years)	4.8	3.9	3.6	5.5	4.5	4.8	3.7	5.2	4.2	4.7	4.8	4.0
Community/recreetion	_					_	_	_				,
program (courses)	3.8	•	ř	3.5	3.7		•		3.9	3.8	•	-
College/University (courses)	8.0	8.0	9.7	8.7	5.0	5.3	6.8	13.9	8.8	7.4	8.9	B.9
Apprenticeship (months)	13.7 🔧	13.5		• .	_#_		•	•	13.5	*		15.2
Adult education (courses)	3.9	•	4.4	3.8	3.9	5.7	2.7	•	4.2	3.7	4.5	3.0
Workshops at craft						_					04.0	
echoole/organizations	9.3	6.0	6.5	6.5	5.0	8.5	4.7	8.1	7.3	5.0	31.0	5.7

SCHRCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Methametics Policy Research, Inc.

5.3

2.8

3.0

2.2

2.4

6.6

2.8

Private Lessons from pro-

feesional (years)

Professional workshops

Based on calculation of  $x^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts-artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exteriok  $\{*\}$ .



2.7

13.4

3.7

<sup>&</sup>quot;Includes respondents reporting a specific amount of training, such as years of lessons, number of courses.

<sup>#</sup> Average to not shown because there are fewer than 50 respondents in category.

MA Not applicable.

professionals extend for an average of nearly 3 years. Art or crafts courses taken through community/recreation programs and adult education number about four, about half the number taken in college or university settings. For these, and other types of training, there is little variation by type of media.

Much more diversity is noticeable in the average number of training experiences when crafts-artists are examined by intensity of crafts involvement. Surprisingly, this is not the case with the percentage of crafts-artists engaging in the various types of training. The differences are statistically significant, but with a few exceptions there is much less variation than when media are used as the classifying variable (Table IV.4). The exceptions are the prevalence of apprenticeship and professional workshops for those for whom crafts is a major (primary) activity.

There is a strong relationship between years of involvement in crafts and types of training; the number of training experiences rises as years of participation increase.

The training that appears to contribute most to financial success in crafts is apprenticeship, although professional workshops also help. All of the other types of training do not appear to lead to higher net income from crafts. In fact, the opposite is true; the next greatest share of crafts-artists have net losses of \$500 or more. This seems to indicate that either financial success is not a goal of all crafts-artists or that training is not a reliable predictor of financial success.

Regarding past training received, crafts-artists report almost overwhelming satisfaction; 90 percent indicated they are at least

satisfied with their crafts training. There is minimal variation in level of satisfaction by media or intensity of crafts involvement and the variation for net income is not statistically significant (Table IV.5).

The extent to which crafts-artists are currently receiving training is still another indication of its perceived value. Nearly 15 percent of all crafts-artists report receiving an hour of training per week. The proportion of crafts-artists currently training is higher for the paper or leather media and those who are involved in crafts as a leisure or occasional activity. Also, it would appear crafts-artists who are currently experiencing net losses in income from crafts are training more intensively than those with a positive net income.

A contributing factor to financial success can be the extent of training in business skills (accounting, contracting, advertising, merchandising). About 30 percent of all crafts-artists report having such training; about 58 percent of those who have not experienced such training desire it (Table IV.5). Moreover, 74 percent of the crafts-artists with net losses of over \$500 desire to receive more training in the business aspects of crafts.

Two-thirds of all crafts-artists plan to seek additional training in crafts (Table IV.6). Smaller shares of wood and glass, other media or media combinations crafts-artists seek training (51 and 47 percent respectively), while larger proportions of workers in the fiber, clay and paper or leather media plan additional training (about 73 for each of the three categories). There is no dominating pattern to the relationship between plans to acquire additional training and level of involvement in crafts.

TABLE IV.5

# SATISFACTION WITH TRAINING, CURRENT INVOLVEMENT AND BUSINESS SKILL TRAINING SURVEY OF CRAFTS-ARTISTS, 1980

<del></del>	*		Pris	ary Cre	ft Hed	i a			Inte	naity of	Crafts In	volve	en t		Inc	oma f	rom Cra	fta
		•						Mejor / Under	Activit 40	YX	Laisura	Activi 10	Occa-			4		/
	Crefte- Artists	raber	Clay	Metel	Wood		Paper, Leather	40 Hra.	Hrs. Plus	2nd Activity	10 Hra. or Lasa			Other	\$500 plus	Under \$500		\$1000 plum
Setiefaction with training received				T												) !!   		
Vary satisfied or satisfied Very unsatisfied	89.6	80.9	92.4	88.1	84.3	94:1	83.8	90.4	82.8	91.9	87.2	90.9	82.5	91 .3	87 .1	88.6	88.1	90.3
or unsattefied	10.4 100.0	9.1 100.0	7.6 100.0	11.9 100.0 423	15,7 100.0 356	5.9 100.0 196	16.2 100.0 138	9,6 100.0 432	7,2 100.0 254	8,1 100.0 426	12.8 100.0 363	9,1 100,0 297	17.5 100.0 298	8.7 100.0 321	12,9 100.0 454*			100.0
Number of Respondents	2593	958	421	423	336	196	136	432	204	420	303	287	230	J2-1 	404	00/	3 3/4	4/0
Percentage now receiving craft training Number of Respondents	14.5 2586	15.0 949	14.3 <b>4</b> 29	11.2 416	5.6 361	6.9 198	22.2 139	11.6 431	3.5 256	11.6 <b>4</b> 22	20. <b>4</b> 367	20.8 300	12.5 297	16.3 320	19.0 4 <b>4</b> 6	19.1 655	9. <u>3</u>	6.7 472
Average (mean) hours per week of training Number of Respondents	1.0 2586	0.7 949	0.9 429	1.7 416	0.8 361	0.6 198	1.4 139	1.3 430	1.4 256	0.6 421	0.8 367	0.8 300	0. <b>4</b> 297	2.0 231	1.3 446	0.9 655	0.5 382	0.8 472
Percentage with training in business akilla Number of respondents	દેક.2 1919	25.5* 603	28.4* 376	30.5* 354	33.3* 251	28.9* 177	30.7* 91	25.7 <b>*</b> 427	30.2* 251	32.4* 408	34.8* 183	28.8° 21 <b>4</b>	28.8* 110	22.4* 211	31 .1' 0' 7	27.8 270	23.3°	* 35.1° 459
Percentage with no busi- nass training with a desire for training in	ı		04.0	cc. a*	40.4	40.0*	00 F#	50.0	50.0	<b>67.</b> 5	45.0	43.9	43.9	<b>50.3</b> .	, 73 <b>.</b> 7	51.4	51 <b>.</b> 5	58.2
business skills  Number of respondents	57.4 1304	57.7* 419	61.6* 266	66.9* 239	48.4* 163	49.2* 117	62.5* 59	69.9 298	58.8 176	256	124	145	74	153	208	194	255	294

SOURCE: Survey of Coafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Based on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted parcentages in this table vary from the parcentages for all crafts-artists by an amount larger than can be accounted for by chance except those parcentages or parcentage distributions followed by an esteriak (\*).



TABLE IV.6

ATTITUDES TOWARD ADDITIONAL TRAINING IN CRAFTS SURVEY OF CRAFTS-ARTISTS, 1980

			P	rimery C	raft He	die			<u></u>	ntensity of	Crafte Invo	Lyamant		
	All Crefts- Artiste	Fiber	Cley	<u> Hetel</u>	Wood	Glees, Other	Paper, Leether	Major / Under 40 Hre,	Activity 40 Hre. Plue	.2nd Activity	Leisure A 19 Hre. or lees	10 Hre. Plue	Occa- aional Activity	Other
Percentege of crafts-artists planning to acquire additional training	67.2	72 <b>.4</b>	72.4	70.1	51.2	46.5	73.3	67 <b>.</b> 6	57.5	71.9	76.9	64.1	57.9	71.5
Number of respondents	2603	956	427	423	360	200	139	<b>435</b>	254	428	369	303	295	322
Barriara pravanting a/ additional training Lack of time unapacified	18.8	18.5	18.1	19.5	17.5	18.0	21 <b>.</b> 6 "	1647	24.0	21 .2	19.1	16.7	18.8	17.1
Lack of time because of full-time noncreft work	6.2	7.3	7.3	5.7	4.6	3.0	4.1	1.3	1.9	8.7	9.7	5.6	6.9	9.2
Lack of time because of family responsibilities	7.8 18.9	11.8 16.6	8.8 21.6	8.1 24.9	1.8 13.8	3.0 18.0	3.1 28.9	7.4 26.4	4.5 633.1	8.1 22.1	12.4 14.5	4.8 14.3	5.4 1J.4	10.0 19.1
Lack of money No training evailable close by	2.6	2.4	2.0	2.4	1.8	, 3.0	7.2	4.5		4.0	1.2		2.3	2.8
No courses or training in specific craft Lack of transportation	5.7 0.7	6.3 0.7	5.3 0.9	5.7 0.4	7.7 0.3	5.3	7.2 2.1	6.1	5.2 1.9	7.8 0.6	5.8 0.6	6.4 1 5	1.9	4.4
No courses or quatified teachers evailable	6.2	4.6	5.8	7.3	10.5	7.5	5.2	7.1	8.4	7.2 2.5	5.2 0.9	8.4 8.8	5.4 3.5	5.2 1.6
Health problems Other reasons	3.4 2.3	<b>4.</b> 9 1.9	0.9	3.7 3.2	3.1 1.8	2.3 4.5	4.1	2.9 1.6	0.6 1.9	2.2	1.8.	.1.2	1.9	4.4
Number of respondents	2615	961	433	421	362	201	139	434	253	431	369	. 304	297	325

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Methematics Policy Research, Inc.

Based on calculation of  $x^2$  statistics, in 98 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esteriak  $\{*\}$ .

ERIC Full Text Provided by ERIC

Percentage of all respondents who listed up to three berriers.

Lack of time, work and family responsibilities and money were the most frequently listed barriers to acquiring additional training.

Lack of time was mentioned by 33 percent (combining the three time-related reasons) and money by 19 percent of respondents (Table IV.6).

Other barriers included no courses or qualified teachers or no courses in the specific media sought by the respondent (about 6 percent each).

Barriers to acquiring additional training seem most severe for paper or leather crafts-artists; they point out lack of time and money, and that no training is available close by. Fewer crafts-artists working in wood note time and money barriers but they do mention lack of courses or qualified teachers. When barriers to additional training are examined in regard to level of involvement, lack of money appears as the most severe barrier for crafts-artists listing crafts as a major (primary) or secondary activity (Table IV.6).

#### C. INVOLVEMENT IN CRAFTS

The amount of time devoted to crafts by members of crafts organizations is a function not only of their interest in crafts, but other competing endeavors such as employment and home-based activities. Another facet of involvement is the degree to which crafts-artists immerse themselves in crafts-work. Some of the manifestations of a crafts commitment include years spent in crafts work, hours per week spent on crafts, association with other crafts-artists inside and outside their families and changes in primary media. Involvement can also be traced through the working arrangements of crafts-artists.

Surprisingly, there is relatively little variation in the Tevel of crafts involvement (primary or secondary occupation, full-time teacher

or student, main leisure or occasional activity) when crafts artists are segregated by media, extent of crafts training, region of residence, number of years worked in crafts and net income from crafts. None of the distributions are statistically significant. Crafts as the main leisure (32 percent) and as an occasional activity (22 percent) together account for over half of all crafts-artists. Crafts is reported as a primary occupation for 24 percent of respondents and as a secondary occupation by 16 percent (Table IV.7).

This consistency is almost as evident for years of involvement in crafts; the average for all crafts-artists is 16 years. Fiber tends to hold the interest of crafts-artists for longer periods of time, but fiber crafts-artists tend to be older than those working in the other media, so the finding may reflect the age of the crafts-artists rather than anything intrinsic to the media. Crafts-artists who list crafts as their primary or secondary activity have not been working in crafts for as long, about 13-14 years on the average, as those reporting crafts as a leisure or occasional activity (Table IV.7). There is no clear association of years of involvement and net income from crafts.

Crafts-artists who have high net incomes from crafts spend longer hours creating and marketing their crafts on average than those making less money from crafts or experiencing losses (Table IV.7). Crafts-artists working in crafts for less than 10 years also devote somewhat more hours to producing their craft than those involved for longer periods. In addition, workers in clay and glass or other media or combinations spend a longer number of hours per week creating and producing than do their counterparts in other media.

TABLE IV.7

INVOLVEMENT IN CRAFTS
SURVEY OF CRAFTS-ARTISTS, 1980

			Prime	. y Cref	t Medi	•		Ext	nt of Cr	aft Trein	ing	<u>R</u>	egion of	Residence	
	All (rafts-						Peper,	Profee-		Coures	None or	North-	_	North	
	Artists	Fiber	CLey	Motel	Wood	Other	Leether	eionel	5 Plus	1 to 4	Informal	600 t	South	Centrel	West
Level of involvement in								v							
crefts:								à			*				
Primery occupation	23.5	17.2	34.5	31.4	21.9	29.9	20.9 *	26.9 (	30.2	13.0	21.6	24.1	23.8	15.6	31.6
Secondary occupation	16.2	16.4	18.9	20.5	11.6	13.9	17.4	18.1	23.4	12.9	12.3	16.4	15.0	14.9	19.3
Full-time teacher	3.8	1.4	10.8	5.1	2.3	1.3	3.1	4.6	12.0	1.8	1.2	4.7	3.2	4.2	2.9
Full-time student	1.2	0.8	2.3	2.7	0.1	0.8	1.9	1.7	1.2	0.9	0.4	0.9	0.3	1.5	2.7
Mein leieure estivity	32,1	38.9	15.1	25.2	41.6	28.6	2t "5	30.5	10.4	38.4	37.3	31.3	31.6	39.2	25.6
Occesional activity	22.3	24.1	17.2	14.4	22.5	25.5 <sub>.</sub>	27.1	17.0	22.8	31.9	26.3	22.0	25.0	23.2	17.7
Other involvement	0,9	1,2	1,2	0,7			<u> 1,1</u>	1,2		0,1	0,9	0,6	1,1	1,4	0,2
TOTAL	100.0	100.0	100.0	100.0	190.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Number of respondents	2536	962*	430*	420*	365*	202*	138*	1382*	197*	417*	618*	926*	675*	530 <b>*</b>	481*
Averege (mean) number of															
yeere spent working . 1			44								40.0		40.0	47.5	40 5
crefts	16.3	17.9	14.3	14.1	16.4	14.4	16.9	16.8	14.3	15.9	13.9	15.5	16.0	17.5	16.5 487
Number of respondents	2608	957	434	419	363	202	136	1276	110	426	529	917	671	526	487
Averege (mean) number of															
hours per week in lest		•													
yeer epent producing										45 3	00.7	20.6	19.4	18.9	23.4
crefte	19.5	17,5	26.0	23.4		24.7	15.0	22.0	24.6	15.3 384	20.7 481	835	612	463	432
Number of respondents	2444	<b>85</b> 0	397	377	334	177	124	1154	96	384	481	830	012	403	432
Average (meen) number of hours per week in last year epent marketing							4		_		, s				
crefte	2.3	1.6	3.3	3.0		-	1.8	2.3	2.5	1 <sub>9</sub> 7	2.8	2.3	2.2	2.2	2.8
Number of respondents	2375	891	394	383	318	178	132	1177	103	394	468	823	620	480	447

ERIC

TABLE IV.7 [continued]

		I	ntensity	of Cref	ts Involv	ement.							I	come 1	rom Cre	ofts
¥		Major	Activi <u>t</u> y	<u>.</u>	Leisure	Activit	Y				olvement		Net Lo	oe of	Nat G	ain of
	All Crefts- Artists	Under 40 Hrs.	40 Hrs. Plus A	2nd Activity	10 Hrs. or less	10 Hrs. Plus	Occe- sionel Activity	Other	Under 10 Yrs,	10 to 14 Yrs,	15 to 24 Yrs,	25 Yrs. Plus	\$500 plue	Under \$500	ปกder \$1000	\$1000 plue
Level of involvement in																
crefts:										00.4	00.0	45 4	00.0	4 0	40.0	63.6
Primary occupation	23.5								28.7	26.1	22.0	15.1	23.8	4.8	16.3	
Secondary occupation	16.2								18.4	17.7	15.9	12.3	16.2	7.5	26.4	
Full-time teacher	3.8								1.2	5.7	7.4	2.6	8.3	0.8	2.4	
Full-time Student	1.2			TABULATE	ED .		NOT TABUL		2.4	1.4	0.2	0.3	2.9	1.7	0.4	
Main telsure ectivity	32.1		(Se	e note)			(See not	: <b>•</b> }	27.4	28.0	30.4	42.9	35.9	42.7	33.5	
Occesional activity	22.3		*						21.4	20.4	24.0	24.7	10.9	41.8	21.0	
Other involvement	0,9							-	0.5	0.7	0,1	2,1	5.0		400.0	0,5
TOTAL	100.0								100.0	100.0	100.0	100.0		100.0	100.0	
Number of respondents	2536								945*	611*	506*	529*	460*	653*	382*	471*
Average (mean) number of years apant working in crafts Number of respondents	16.3 2608	13.7 438	13.2 254	14.3 429	16.3 3 <b>6</b> 5	21.5 303	17.3 296	16.7 236	5.8 948	11.0 614	17.7 512	36.1 532	15.2 458	17.7 652	16.2 379	14.0 474
Averege (meen) number of hours per week in last year spent producing crefts Number of respondence	19.5 2*44	29.7 <b>438</b>	54.9 256	20.7 430	6.8 371	20 <b>.</b> 5 307	5.0 298	22 <b>.</b> 0	22 <b>.</b> 5 863	20.7 550	19.4 451	17.9 465	21.5 434	11.1 557	17.5 336	34.1 464
Averege (meen) number of hours per week in lest year spent merketing crefte Number of respondents	2.3 2375	6.7 398	7.2 228	2.7 386	0.2 346	1.0 280	0.3 290	1.6 218	3.0 867	2.1 549	3.0 447	1.1 486	2.0 425	0.4 635	2.2 342	5.5 442

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Besed on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than cen be accounted for by chance except those percentages or percentage distributions followed by an esteriak (\*).

NOTE: Level of involvement in crafts is not tabulated by intensity of crafts involvement because the former is one of the two variables used to define the latter catagories.



30

Some other characteristics of crafts involvement indicate the extent of other craft-related activities. For example:

- o 91 percent of all crafts-artists visited crafts galleries and museums during the 12 months prior to completing the questionnaire
- o 88 percent read an average of 2.4 different crafts publications
- o 82 percent belong to an average of 1.5 craft organizations
- o 68 percent collect crafts.

Workers in clay and metal tend to be more museum-, gallery- and publication-oriented; paper or leather crafts-artists visit museums and galleries but do not read publications to the same extent as those in other media. Crafts-artists working in clay collect crafts to a greater extent than their counterparts (Table IV.8).

Crafts-artists tend not to change media frequently; nearly half, 45 percent, have worked in the same materials for more than 10 years. For those who changed media within the three years prior to the survey, personal factors such as discovery of a new theme or a change in interests are the most important reasons cited (Table IV.9).

A different perspective on crafts involvement comes from a consideration of intergenerational involvement and whether other household members share the respondent's interest in crafts. Crafts

<sup>1/</sup>Because all respondents were selected through their association with a crafts membership organization, this percentage should be 100.

However, apparently not all crafts-artists considered the organizations to which they belong as "crafts" organizations. Support for this hypothesis is found in the level of effort that had to be devoted to convincing some potential respondents that they met the definition of a crafts-crtist. Some did not consider their work "worthy" of consideration while others insisted their work was art, not crafts.

TABLE IV.8

### PARTICIPATION IN CRAFTS-RELATED ACTIVITIES SURVEY OF CRAFTS-ARTISTS, 1980

1	v			Primery Cr	eft Hedie		
	All Crefts- Artiete	Fiber	Ctey	Hatal	Wood	Gless, Other	Peper, Leether
	٧	<b>5</b>					
ercentege of crefte- ertists involved in the following ectivities in the lest 12 months:		N mer			•		*
Visiting crafts galleries/	90.8	<b>81.5</b>	97.1	94.4	79.7	87.4	95.6
Number of respondents	2608	954	439 "	419 ~	362	202	140
Reading crafts publications Number of respondents	87.6 2605	96.4 S	94.3 495	90.6 419	89.5 362	76 <b>.</b> 2 200	<b>81 .</b> 3 137
Average [meen] number of publications read Number of respondents	2.4 2292	- 2 <u>.7</u> 939	2.5 413	2.1 386	1.8 310	2.0 158 <sub>3</sub>	1.9 113
Belonging to crefte organi- zetione Number of respondents	91.6 2610	61.9 7 <b>9</b> 6	83.2 3 <b>9</b> 6	84.2 377	300 80.9	77.0 170	59.8 <sup>4</sup> 102
Average (meen) memberships Number of respondents	1.5 2610	1.7 958	1.6 433	4.6 421	1.3 363	1.3 198	0.8 139
Collecting crafts Number of respondents	6 <b>8.</b> 0 2552	70.8 936	79.5 431	65.3 40%	52.6 353	59.8 197	66.6 135

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Besed on calculation of  $x^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exterior  $\{^*\}$ .

TABLE IV.9

EXTENT OF CRAFTS INVOLVEMENT SURVEY OF CRAFTS-ARTISTS, 1980

•			Prime	ry Cre	ft Hed	ii e			Inter	eity of Cr	ofts Invo	lvement						
								Melor	Activit	v	Leieure Ad	ctivity	4.			re of		
	All Crafta-			1.		Glees,	Papar,	Under 40	40 Hre.	2nd	10 Hre.	10 Hra.	Occe- eionel		10	10 to	24	Yre.
	Artists	Fiber	Clay	Metel	Wood	Other	Leether	Hre,	Plue	Activity	or less	Plue	Activity	<u> Uther</u>	Yra.	Yra.	Yra,	Plue
Langth of time worked																		
in primary madium:																		
Less then 5 years	26.5	24.9	21.3	28.6	25.0	40.4	30.8	22.7	20.1	22.8	30.4	21.2	36.2	31.5	55.1	16.0	10.4	10.5
6 to 10 years	28.9	28.1	34.4	35.3	26.5	27.3	21.5	37.B	32.0	35.6	29.2	20.7	22.7	25.8	41.3	43.6	12.8	10.3
More then 10 years		47,0	44.3	36.1	48,5	32,3	47,7	39.5	47,8	41,6	40,4	58,1	41.1	42,7	3.8	40,4	76.8	79,2
TOTAL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.D	100.0	100.D	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Number of respendence	2627	967	436	422	365	201	1 40	438	256	431	370	307	'296	326	947	613	512	527
Descented who changed		*			**								*					
Parcentage who changed andium in Lest 3 year																		
		8.2	5.5	5.5	5.2	6.9	19.1	5.74	5.7*	8.6*	8.6*	6.0*	5.4*	11.9*	3.8	9.1	8.7	10.1
Discovered new the		0.2	0.0	0.00	U.L	0.0		.,	•••				- •		-			
Increasing cost of		0.9*	0 58	1.0*	0.94	1.1*	1.4*	0.84	3.3*	1.4*	1.2*		1.3*	2.8*	0.6*	1.8*	1.8*	2.0*
energy	1.4	0.8	2.5*	1.0	U.8	. 1.	1.4	0.0	0.0	1.4			3	_,_			• • •	
Few teaching post-			4.0*	0.0		•		0.2	3.1	0.4	0.2	-	0.4	0.1	0.2*	0.3*	0.5*	0.9*
tione eveileble	0.4	0.1*						7.3	3.5	10.5	10.5	11.7	14.8	16.7	6.9	13.6	11.9	14.4
Change of interest		12.7	9.0	7.1	5.6	14.2	24.6	3.74		4.8*	1.9*	2.0*		2.6*	1.7*	4.6*	1.8*	3.5
, Other	2.9	1.6	4.0	3.7	2.1	3.7	9.9					305	295	327	842	612	508	528
Number of respondents	2619	962	433	421	363	503	1 40	436	255	427	388	300	200	327	0 <b>4</b> 6.	OIE	000	JEU
Perents involvement in crefts													d d					
Yes, es primary													Ĉ P					
source of income	2.8	2.7	2.7	2.9	3.3	2.1	5.1	2.7	8.7	2.2	1.6	2.8	0.9	1.4	1.5	2.5	2.2	4.9
Yes, ac secondary																		
acurca-of income	2.9	3.4	4.7	2.2	3.5	1.9	1.6	4.5	4.3	4.9	1.1	2.7	1.1	1.6	1.8	2.9	4.5	3.3
. Yee, so a leisure		- •	•		•	-												
* octivity	29.7	38.9	20.2	20.5	25.9	28.8	24.7	23.9	19.1	28.7	38.6	37.3	27.9	24.8	23.9	25.7	31.5	40.2
No	64.6	55.0	75.4	74.4	67.3	67,2	68.6	69.0	69.9	64.2	58.7	57.2	70.1	72,2	72,8	68,9	61.8	51,8
TOTAL	100.0	100.C		100.0		100.0	100.0	100.0		100.0	100.0	100.0	100.0	100.0	100.0	100.D	100.0	100.0
Number of responderits	2634	967	436	424	364	203	139	438	255	429	370	307	298	239	947	613	512	531
Other embers of household involved						*						5				-		
in crafta:																	4= 4	40.0
Yee, eege medium	14.3	11.3	13.1	14.7	19.4	20.7	17.8	16.7	22.1	11.1	13 . 1	16.8	11.8	14.5	14.2	12.5	15.1	16.0
Yee, different																		
madium .	19.1	19.0	13.8	18.9	24.3	18.2	17.4	18.7	16.6	19.2	21.8	18.1	" 18 <b>.</b> 7	19.3	15.9	20.5	19.2	22.4
No	66.6	69.7	73.1	66.4	56.3	61.1	65.0	3, 39	61,3	69.7	65.1	65.3	89,5	86,2	69.9	<u>67.0</u>	85,7	61,6
TOTAL	100.0	100.0	100.0				100.0	100.0	100 0	100.0	100.D	100.0	100.0	100.0	100.0	100.0	100.D	100.0
Number of respondents	2627	965	435	423	361	203	140	438*	255*	430*	370*	304*	296*	328*	945*	614*	510 <b>*</b>	529*

SOURCE: Survey of Crefts-Artists who are Members of Crefts Organizations, Mathematics Folicy Research, Inc.

Ric on calculation of X<sup>2</sup> atatistics, in 88 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an accounted for by chance except those percentages or percentage distributions followed by an esteriak (\*).

provided a primary or secondary source of income for about 6 percent of crafts-artists parents. A much larger share, 30 percent, report their parents participated in crafts as a leisure activity, especially for crafts-artists who engage as a leisure/occasional activity. About a third of all members of crafts-artists households also engage in crafts. Wood and glass or other media or combinations of media seem to appeal to other household members as well as to the crafts-artist respondent (Table IV.9).

Over 80 percent of all crafts-artists worked alone without a partner, paid or unpaid employees. However, 30 percent report having had one or more apprentices, 13 percent with one, 17 percent with 2 or more. Nearly 70 percent of the apprentices were paid and 36 percent worked full-time. There is not a statistically significant variation for these characteristics by media (Table IV.10).

A related characteristic, the specific place where crafts are made, indicates 70 percent of all crafts-artists work in their home, with the largest percentages reported by the fiber, leisure and occasional activity and net loss of less than \$500 subgroups. The next most prevalent locale is a private studio, about 20 percent overall. Studios are reported more frequently by clay and metal crafts-artists, those for whom crafts is a major activity and those who earn at least \$1,000 in net income from crafts. Work space provided by schools is more common for clay and metal workers--about 10 percent for each (Table IV.11).

Nearly a sixth (17 percent) share their workspace; crafts-artists working in clay and as a major activity do so more than the other subgroups.

The work of very few crafts-artists has been affected by local zoning or building regulations; about four percent reported any problems,

TABLE IV.10

USE OF PARTNERS, EMPLOYEES, HELPERS AND APPRENTICES
SURVEY OF CRAFTS-ARTISTS, 1980

				Primary Cr	eft Medie		
	All Grafts- Artista	Fiber	Cley	Metel	Wood	Gless, Other	Paper, Leather
Percentage of crafts- artists working:							
Atons	82.5	86.9	78.3	81.8	80.0	72.1	85.1
With a partner	8.1	4.3	13.4	9.3	8.9	12.1	6.0
With paid employees	6.2	5.1	5.4	7.1	8.7	12.3	1.9
With unpeid helpers	3,2	3.7	2.9	1.8	2.4	- 3 <u>.5</u>	7.0
TOTAL	100.0	100.0	100.0	100.0	100.0	100.0	7.0 100.0
Number of respondents	2612	962*	434*	419*	360*	505*	140*
Percentage of crefts-artists with epprentices in lest 3 years: One More than one None TOTAL Number of respondents	13.4 17.1 <u>69,5</u> 100.0 508	11.0 16.9 <u>72.1</u> 100.0 136*	16.8 17.9 <u>65,3</u> 100.0 95*	15.1 16.2 <u>68.7</u> 100.0 93*	15.3 15.3 <u>69.4</u> 100.0 67*	11.9 20.9 67.2 100.0 67*	6.7 13.4 <u>79.9</u> 100.0 30*
Percentage of crafts-ertists							
with paid apprentices:	67.8	53 <b>.0</b> *	58.2*	68.5*	83.1*	88.4*	14.0*
Number of respondents	168	43	34	30	24	24	6
Percentage of crafts-artists	_	*	*	40.4		04.04	44.0
with full-time apprentices	35.5	27.9*	32.7*	40.1*	50.1*	24.6*	14,0*
Number of respondents	167	43	34	30	23	24	6

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Based on calculation of  $X^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esteriak  $\{*\}$ .



TABLE IV.11

LOCATION OF CRAFTS WORK AND EFFECT OF LOCAL ORDINANCES ON LOCATION SURVEY OF CRAFTS-ARTISTS, 1980

			Pri	mery Cr	oft Med	18			Inte	neity of C	refts Inv	<u>o Lveme</u> n	t		In	come f	rom Cre	fte
				<u> </u>				<u>Mejor</u> Under	Activi 40	ty	Leteure	Activit	Y Occa-		Not Lo	ee of	Net Ge	in of
	Ali Srefte- Artists	Fiber	Cley	Ketal	Wood		Paper, Leather	40	Hrs. Plus	2nd Activity	10 Hrs. or Less	Hre. Plue	sional Activity	Other	\$500 plus	Under \$500	Under \$1000	\$1000 plus
Primary locals of crafts			•															
work:			40.4	E0 E	67.4	66 .0	70.7	52.6	33.0	68.0	86.3	86.1	79.3	62.8	84.9	80.8	76.0	49.9
In home	70.0	B7.4	46.4 36.3	50.5 34.6	67.4 24.6	26.7	16.7	38.0	59.8	29.9	7.3	10.7	8.4	15.7	24.1	7.0	16.4	42.2
Privata atudios	20.5	7.7		1.3	2.7	0.9	1.4	4.8	3.1	3.5	0.4	0.1	2.2	0.1	2.7	1.2	1.5	3.7
Cooperative studio	2.0	0.7 0.7	5.4 2.3	1.4	0.9	1.3	4.9	1.3	5.0	0.5	2.5	0.4	5.0	2.0	1.1	2.2	2.0	1.2
Community studio	1.4	U./	2.3	1.4	U.8	1,5	7.0	1 .0	F .0	0,0		-•.						
Space provided by	<b>5</b> 2	2 2	9.4	9.9	4.2	4.7	4.8	1.8	1.4	1.5	3.5	1.9	7.8	17.5	8.3	8.3	3.5	1.5
echoot	5.2	2.3 1.2	0.2	2,3	0.2	0.4	1.5	1.4	0.7	1,6		0.8	0.5	1.9	0,9	0,5	0,6	1.5
Other TOTAL:	0,8 100.0	100.0	100 0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
	2618	963	434	421	362	202	140	438	255	428	371	308	295	324	457	657	381	473
Number of respondents	2010	800	404	-74-1														
Crafts-artists whose wor	k																	
space is:													70.4	60.0	70.0	70.6	74.1	69.7
Dwned	71,9	76.4		66 .1	79.2	66.6	61.7	66.6	60.5	69.6	80.3	81.1	72.1	63.3	70.2 21.3	72.6 12.8	16.3	22.5
Rented	17.3	14.7	23.1	10.0	11.3	23.8	21.8	26.7	32.4	21.9	10.0	11.6	13.6	14.1			9,6	7.8
Free of cost	10.8	8,9	12,8	15,1	9,5	9,6 100.0	16.4	6,7	7,1	8.5	9.7	7.3	14.3 100.0	22.6 100.0	8,5	100.0	100.0	100.0
TOTAL:	100.0	100.0		100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0 307	268	316	451	644	3/8	474
Number of respondants	2599	961	428	414	360	199	137	437	256	428	362	307	600	310	401	U-T-		7, 7
Descentane who obser																		1
Percentage who share																		- 1
workspace with other crafts-artists	16.9	10.4	27.2	19.3	19.5	17.8	20.0	22.3	22.0	14.4	13.4	8.5	15.7	25.6	18.7	17.9	12.2	18.≇
Number of respondents	2584	955	429	417	357	200	137	437	255	428	363	301	291	317	455	645	375	471

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Methanstics Policy Research, Inc.

Based on calculation of  $x^2$  statistics, in 88 out of 100 cases, the weighted percentages in this table very from the percentages for all crefts-ertists by an amount larger than can be accounted for by clance except those percentages or percentage distributions followed by an esterisk  $\{*\}$ .



nearly evenly split between problems when crafts-artists relocated within the same area and when they moved to a different area.

D. ATTITUDES TOWARD CRAFTS: SATISFACTIONS, BARRIERS AND IMPORTANT PROBLEMS

why do crafts-artists who are members of crafts organizations engage in crafts? Income is one motive, satisfaction in personal accomplishment and as a diversion are others. Actually, in view of the diversity in crafts, it is reasonable that there is a combination of satisfactions that shape individual experience. There are also barriers which obstruct satisfaction and problems which hinder participation or satisfaction.

when crafts-artists were asked to rank their satisfactions, the one most often listed was crafts as a "means of creative expression" (42 percent). Listed second most often was a "sense of accomplishment" (17 percent). These, along with "life's work", summarize the intangible or creative benefits of crafts compared to satisfactions such as producing works for purchase, as a significant source of income or a diversion from daily routines (Table IV.12).

Second-ranked satisfactions still are intangible, but the order is reversed; "life work" is the most important (32 percent), while "means of creative expression" (16 percent) is next most frequently mentioned. The third most important satisfaction is "producing crafts for purchase" (13 percent); then comes as a "diversion from daily routine" (15 percent).

The responses are fairly convincing evidence for the primacy of creative satisfactions from crafts with financial or diversionary motives clearly secondary. It would appear crafts-artists look to



TABLE IV.12

TYPES OF SATISFACTIONS FROM CRAFT WORK SURVEY OF CRAFTS-ARTISTS, 1980

				Primary Cre	ft Hedie		
	All Crafte- Artiete	Fiber	Cley	Hetel	Wood	Glees, Other	Peper, Leether
Percentage of crafts-artists with types of satisfactions derived from craft work							-
Most important					22.2	<b>50.4</b>	40.0
Heene of creative expression	42.4	59.4	58.0	50.6	39.8	50.4	40.2
Life's work	6.0	3.5	12.4	10.1	6.4	9.6	18.6
Sense of accomplishment	17.1	20.8	16.5	22.7	26.4	14.8	27.8
Produce works for purchase	3.5	3.⊍*	2.3*	5.7*	6.4*	9.6*	5.2*
Darive eignificent income	2.0	2.2*	2.3*	3.2*	2.0*	1.1*	3.1*
Diversion from daily routine	8.0	10.4*	6.1*	7.7*	16.4*	10.4*	4.1*
Other	0.6	0.5*	1.5*	0.4*	1.5*	0.7*	
Second most important						40.5	44.0
Means of creative expression	16.3	17.9	19.0	21.5	21.3	18.5	41.2
Life's work	5.9	6.5	14.5	9.7	3.4	3.0	4.1
Sense of accomplishment	31.9	43.3	37.3	38.9	39.8	39.3	28.9 6.2*
Produce worke for purchase	6.3	6.8*	7.8*	8.9*	7.6*	12.6*	3.1*
Derive eignificent income	2.8	3.4*	5.2*	4.D*	2.7*	5.2*	
Diversion from daily routing	7.2	11 .0*	5.5*	5.7*	10.6*	<b>0.9</b> *	7.2*
Other	0.6	0.4*	2.0*	0.4*	1.2*		
Third most important		_		=	40.0		7.0
Means of creative expression	7.4	7.6	10.4	11.7	10.3	8.2	7.2 8.3
Life's work	6.3	6.5	10.1	11.3	4.8	8.2	18.6
Sense of eccomplishment	13 <b>.4</b>	16.2	22.3	15.4	11.9	18.5	7.2*
Produce worke for purchase	12.7	14.4*	21.4*	17.4*	16.4*	16.3*	10.3*
Derive eignificent income	4.4	3.5*	5.8*	8.1*	5.5*	8.2*	25 <b>.8</b> *
Diversion from daily routine	14.6	22.7*	11.6*	14.2*	17.6*	11.9*	•
Other	0.7	1.7*	0.3*	0.4*		0.7*	
Number of respondents	2587	968	436	140	203	424	365

60URCÉ: Survey of Crefte-Artists who are Members of Crafts Organizations, Mathematics Policy Russerch, Inc.

Besed on calculation of  $\chi^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an estarisk  $\{^{+}\}$ .



receive satisfaction from their own involvement in the creative process. This is re-inforced by the general lack of variability in the order of satisfactions when examined by type of media (Table IV.12). However, somewhat larger shares of crafts-artists who work primarily in fiber, wood and glass or other media or media combinations do report diversion as the most important satisfaction. Clay and metal seem more oriented toward creative satisfactions then the other media; paper and leather conform most closely to the pattern for all crafts-artists (Table IV.12).

The barriers to achieving these satisfactions are numerous and include non-craft obligations, cost of materials, lack of public education or exposure to crafts, lack of marketability, lack of training, lack of studio space and lack of recognition. All of these were mentioned by 45 to 18 percent of all crafts-artists (Table IV.13).

Non-craft obligatons were mentioned as the most significant barrier to further satisfaction by crafts-artists in most of the media sub-groups. Metal and glass or other media or media combinations both have the expense of materials mentioned more frequently. Another important difference is in the varying levels of crafts involvement. Crafts-artists who consider crafts their primary activity are much more concerned with extent of public exposure/education, costs of materials and marketability and recognition than are those with a less intense involvement (Table IV.13). For these latter crafts-artists (secondary, leisure or occasional activity), non-craft obligations and costs of materials predominate. When scaled by net income received from crafts, the major difference is the pressure felt by those with net incomes of \$1,000 or more from current production commitments.



TABLE IV.13

BARRIERS TO SATISFACTION FROM CRAFTWORK AND CRAFT-RELATED GOAL 3

SURVEY OF CRAFTS-ARTISTS, 1980

			Prim	ery Cre	ft Hed	ie			Inte	neity of C	refte Invi	<u>Lyam</u> ant	<u> </u>		In	come f	rom Cre	/to_
									Activi	<u>ty</u>	Leteure				Not Lo	of of	Not Go	in of
	All Crofts- Artisto	Fiber	Cley	Motel	Wood		Peper, Leether	Under 40 Hre.	40 Hre. Plue	2nd Activity	10 Hrs. or Less	10 Hre. Plue	Occa- sional Activity	Other	\$500 plus	Under \$500	Under <b>≉1</b> 000	\$1000 <u>plua</u>
Percentage reporting														Ar.				
the following berriers to further estimates:														Ĩ				
Lack of training	23.7	24 5	18.1*	23.5*	27.2*	16.7*	31.9*	16.6	12.9	19.1	36.8	24.5	29.3	22.2	23.4	29.7	22.5	17.9
Lack of atudio apaca Lack of materials	23.4	19.2	33.5	23.3	17.9	27.5	28.4	20.1	30.0	28.4	20.1	18.4	21.8	31.8	27.3*	22.8*	20.8*	24.4
eveilable Lack of materials	9.0	7.9	5.5	8.1	12.0	14,4	14.2	8.6*	10.4*	9.5*	9.5*	12.3*	4.9*	7.6*	6.9*		10.7*	11.0
due to expense	31.7	31.3	26.1	47.8	22.1	35.6	25.9	33.8*	36.5	32.5*	31.0*	35.2*	26.0*	32.9*	37.0*	28.6*	28.9*	
Lack of recognition Lack of exposure/	18.2	14.4	22.5	24.9	14.4	20.2	31.8	33.0	30.2	25.6	13.2	9.7	6.8	15.2	22.2	11.6	22.6	25.4
education of	05.0	25.20	31.1*	27.6*	18.8*	22.4	31.3*	35.4	40.7	38.0	16.2	19.6	13.4	25.3	32.0	15.5	32.6	28.4
public to crefte	25.2		28.1	30.3	16.6	30.4	35.2	37.5	28.4	34.6	18.7	21.3	14.6	20.6	29.9		32.9	26.5
Lock of merketebility Lock of poor communi-	24.5	22.2	20.1	30.3	10.0	30.4	30.2	37.0	E0.4	04.0	,							
cation	9.2	7.8	14.7	13.9	4.7	6.9	10.5	9.6	17.9	12.7	8.1	4.3	6.8	9.3	14.2*	7.7*	8.8*	
Lock of time	4.7	5.2*		4.9*	5.8*			3.8	2.6	3.7	4.8	3.1	9.6	5.1	4.2*	5.2*	4.5*	3.2
Hezerde releted to	٦٠,	0.2	•••		•••	•••				- •								
equipment or																		
meteriele	6.8	2.0	15.7	13.3	2.5	11.1	11.4	12.2	14.1	6.7	3.0	2.7	5.0	10.3	9.4	3.6	7.3	10.6
Non-creft obligations	44.8	51.4		43.8	32.1	31.2	46.1	720.1	17.9	58.4	50.9	42.2	55.0	49.5	50.8	52.2	41 .9	35.2
Pressure from current		_	-	-														
production commit-																		
mente	8.6	5.5	13.0	10.4	8.9	13.1	9.0	22.7	31.5	5.7	2.1	4.3	2.1	4.3	5.9	2.5	6.3	26.
Other	5.4	5.1*	7.5	S.5*	4.9*			7.3	7.1	6.5	2.1	7.1	3.5	4.0	4.2*		4.8*	
Number of respondents	5855	966	434	424	360	203	139	437	255	430	368	305	295	327	458	656	380	476
Percentage with specific goals for the next 5		*																
yeere:										.40.0	00.0	00.4	45.0	07.	20.0	24.4	31.5	43.9
Win recognition/ewerd	31.1	29.9	38.4	34.8	24.1	37.0	32.0	45.9	44.2	•42.3	20.3	33.1	15.8	27.6	39.0	24,1	31.0	40.5
Osystop srtistic competence	55.9	58.0	62.6	56.3	51.2	48.1	44.3	59.4*	54.2*	59.9*	59.4°	59.3*	45.7*	56.8*	<b>63.8</b>	59.6	51.0	52.0
Increses income from								70.4	70.0	00.7	00.4	32.4	20.5	30.9	47.3	23.4	43.0	67.2
increase income from	41.4	34.7	53.3	56.2	36.8	43.4	34.0	70.1	72.2	80.7	23.4	32.4	20.0	30.9	47.3	20.4	40.0	07.2
eele of production works	19.6	16.7	25.1	24.7	17.2	25.7	18.6	39.8	33.6	26.8	10.6	17.3	6.6	12.5	19.3	9.8	27.0	33.4
Osyots more time to				EC E	E2 F	40.0	50 <b>#</b>	22 6	40 4	72.5	75.6	62.5	64.7	58.2	68.5	67.7	58.0	36.3
crefte	58.8	63.9	56.0	56.5	53.5	42.8	59.6	33.6 6.9*	18.1 8.5*		2.8*	5.7°	3.2*	6.3*	6.7*		3.4*	
Other	5.4	5.2*		6.9*	4.5*				255	431	370	306	297	328	460	659	382	476
Number of respondents	2628	968	435	423	363	202	139	437	<b>200</b>	4-J I	3/0	300	E#/	JED	700	~~		7/0

SOURCE: Survey of Crafte-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Besed on calculation of x<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an it larger than can be accounted for by chance except those percentages or percentage distributions followed by an esterisk (\*).

The strength of the intangible commitment to crafts is manifest by the five-year goals set by craft-artists. Two are listed by about equal proportions of crafts-artists: 59 percent wish to devote more time to crafts; 56 percent wish to develop artistic competence.

Secondary goals are to increase income from the sale of crafts and win recognition or an award.

Again the lack of variation in the order of these goals by subcategory of crafts-artists is more notable than the differences in the level of the individual percentages (Table IV.13). Increased income is more important to clay and metal crafts-artists and those who consider crafts their primary activity or earn more than \$1,000 in net income from crafts.

A different perspective is gained from examining the current problems facing crafts-artists. The array of problems is more varied reflecting such concerns as the impact of the general economy on crafts and crafts-artists (Table IV.14). The problem responses were listed by crafts-artists; no response categories were printed in the question-naire. Seven types of problems are mentioned of which lack of time and lack of public appreciation are most frequent (about 20 percent for each). Four economic problems (inflation, difficulty of making a living from crafts, cost of materials, and marketing) are each reported by 14 to 10 percent of crafts-artists.

The dominant problems limit participation or inhibit marketing, giving, for the first time, more emphasis to the commercial aspects over the creative. The degree to which these two motives for crafts involvement co-exist in crafts-artists with the same modia, involvement and income earning characteristics is examined in the following sections.



#### MOST IMPORTANT PROBLEMS SURVEY OF CRAFTS-ARTISTS, 1980

•		-		Primer	y Cref	't Med'i e			1	Intensity	of Crafti	Invo	lvement		<u>I</u>	ncome	from Cr	ofto_
٨.	All Crefte- Artiete	Fiber	Ci∎y	Metel	Wood	Gless,	Papar, Leather	Under 49	Hre.			10 Hrm.	Occa- Dional J Activity		\$500	Under	Not G Under \$1000	\$1000
Percentage of crafts-artists										w.	· A	•	•	f	ă"	*,		
reporting the following es the most important problems										· ·				1		J.		
in crefte:											*		•/,	·				
Luck of time	22.9	27.7	18.3	14.9	22.1	12.3	18.7	8.3	10.7	19.8	29.9	25.1	91.1	27.9	20.5	32.0	18.5	18.5
Economic problems, infla- tion Hard to make a living from	13.8	8.2	18.5	13.1	9.9	15.0	13.8	12.3	22.8	15.5	8.7	5.9	8.9	12.0	9.4	8.0	14.2	18.8
crefts	9.5	7.0	13.5	10.8	8.9	9.2	3.4	13.7	10.3	11.4	2.8	. 7.4	8.4	7.1	8.3	7.2	11.2	10.5
Costs of materials Compatition from other	11.9	8.9		16.1		11 .8	10.8	11.9	5.9	8.4	12.5	15.4	8.2	8.0	14.9	8.2	9.1	7.8
crefts-ertists	4.1	2.8	3.4	1.3	5.5	3.3	2.3	2.0	3.1	, 4.4	2.5	4.1		3.1.		3.4	2.8	
Marksting problems	11.9.	12.9	9.1	8.8	5.1	13.0	9.8	15.4	10.1	12.8	8.1	7.9	₄8 <b>,</b> 8	7,3	10.9	3.2	10.3	10.0
Lack of public appraci-			"														43 ==	40.5
<b>etion</b>	19.0	17.9		18.9		15.7	28.9	22.0		19.5	14.5	14.6	18.8	17.9		15.8	17.5	
Other problems	20.7	14.8	16.5	18.4	23.2	19.8	12.7	14.4	18.3	10.8	21.2	19.4	16.5	15.7	18.1	18.1	18.4	12.8
Number of respondents	2082	745	378	350	270	165	113	387	235	371	264	237	200	251	398	478	312	421

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Based on calculation of x<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esteriak (\*).

196

IV.31

#### E. EXHIBITING CRAFTS

The offering of crafts for exhibition serves two purposes:

- o It is a way of achieving recognition by crafts-artists' peers and the public
- o It may be a direct or indirect marketing activity.

It is difficult to separate these motives; especially because the second motive in traditional market economies is the final arbitrator of acceptance and success. However, in the world of crafts, participation and receipt of awards in juried exhibitions are highly valued surrogates. There are many opportunities to exhibit and crafts-artists have definite preferences. Crafts-artists also were questioned on their specific problems when they exhibit.

Over 82 percent of all crafts-artists exhibited their works within the 12 months prior to completing the questionnaire (Table IV.15). Crafts-artists who consider crafts their primary activity exhibit to a greater extent than other groups. When examined by net income category, crafts-artists with net gains of less than \$1,000 exhibit to a lesser extent than others, including those with net losses.

Among those who exhibit, nearly half (46 percent) did 30 from two to five times in the 12 months prior to completing the questionnaire; thirteen percent exhibited continuously (Table IV.15). The frequency of exhibiting is higher for crafts-artists engaging in crafts as a primary or secondary activity and for those who have net gains in income from crafts. By media type, crafts-artists working in clay and glass or other media or media combinations exhibit more frequently than their counterparts.



TABLE IV.15

EXHIBITION FREQUENCY AND AWARDS SURVEY OF CRAFTS-ARTISTS, 1980

			Prim	ery Cre	ft Hed	10			Inter	eity of C		In	f to					
			*						Activit	<u>.</u>	Leieure /				Net Lo	oo of	Not Go	in of
	All Crefte- Artists	Fiber	Clay	Motol	Wood		Peper, Leether	Under 40 Hre.	40 Hre. Plue	2nd Activity	10 Hrs. or Less	10   Hre. Plue	Occa- microsl Activity	Other	\$500 plum	Under \$500	Under \$1000	\$1000 plue
Percentage of crafts- artists who exhibited their crafts in the													75.0	•	05.4	<b>6</b> 2 3	72.8	85.1
Leet 12 months	82.2	85.5	89.4	78.7	74.2	81 .1	83.0	90.4	89.0	83.1	81.8	82.2	75.2	80.8	85.4	83.3	/2.8	80.1
Number of respondents who exhibit	2176	815	378	340	288	168	107	399	234	369	291	253	225	264	398	534	285	425
Frequency of exhibiting in lest 12 months Less than two times Two to five times Six to ten times More than ten times Continuously	20.7 46.4 12.2 7.7 13.0	24.3 49.2 10.2 5.7 10.8	38.5 16.8 8.3 21.2	19.3 45.9 12.5 9.1 13.2	18.7 56.7 7.9 6.9	10.4 42.5 17.1 18.0	31.8 42.9 16.1 3.7 5.7	11.8 39.9 18.9 9.2 20.4	7.8 31.3 17.5 13.7 29.7	14.4 46.8 13.7 13.0 12.3 100.0	32.9 56.4 5.7 1.2 3.8 100.0	20.9 55.2 8.9 8.3 8.7	45.5 43.5 8.3 0.8 1.9	25.7 52.8 8.9 3.4 9.4	17.8 53.0 14.5 5.1 9.8 100.0	35.8 51.8 8.1 1.7 4.8	20.0 48.3 12.2 8.8 14.9	8.8 38.1 14.2 18.2 25.1 100.0
TOTAL Number of respondents who exhibit	1731	100.0 583	100.0 334	100.0 292	100.0 225	100.0	100.0 83	394	233	359	178	198	90	193	319	282	272	415
Percentage of crafts- artists who won award or Prize in lest 3 years Number of ryspondents who exhibit	42.2 1720	42 <b>.</b> 6° 580	* 41.5* 332	40.9 <b>*</b> 291	45.3° 223	48.7 <sup>4</sup>	39 <b>.4</b> *	48.7°	48.4 <b>*</b> 231	40.5* 356	30.7 <b>*</b> 179	42.0* 197	35.4 <sup>‡</sup>	48 . Զ • 193	<b>46.</b> 0	3 <b>4.4</b> 281	37 <b>.</b> 6 271	47.0 415

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Bessed on calculation of  $x^2$  statistics, in 88 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esteriak (\*).



Excludes respondents who reported they do not show their creft work.

About 42 percent of all crafts-artists who exhibit received at least one award or prize during the 3 years prior to responding. There is no statistically significant variation in receipt of awards among types of media or level of involvement. However, larger shares of crafts-artists with net losses of \$500 or more and net gains of \$1,000 or more received awards compared to those with smaller losses or gains (Table IV.15).

The variety of types of exhibiting or selling establishments where crafts-artists exhibited is quite broad and ranges from art or crafts fairs (46 percent) to cooperatives, mail orders, wholesalers and work group meetings (about 6 percent each). Among the more frequently used outlets are: own shop/studio (35 percent), through commissions (31 percent), art/crafts galleries (29 percent) and crafts shop: (25 percent).

Art/crafts fairs are used by clay and glass or other media or combination of media crafts-artists to a greater extent than those who work in other media. Also respondents working in clay prefer their own shop or studio, crafts shops and art and crafts galleries compared to others. Commissions were more prevalent among metal workers (Table IV.16).

As the level of involvement declines from a primary to a leisure or occasional activity, the percentage showing or selling also declines. However, even among these less than full-time participants, art/crafts fairs, own shops or studio, crafts shops, commissions and art/crafts galleries are still\_the\_most important outlets. The subgroup which shows or sells the least includes those who have net losses of less than \$500,



About 42 percent of all crafts-artists who exhibit received at least one award or prize during the 3 years prior to responding. There is no statistically significant variation in receipt of awards among types of media or level of involvement. However, larger shares of crafts-artists with net losses of \$500 or more and net gains of \$1,000 or more received awards compared to those with smaller losses or gains (Table IV.15).

The variety of types of exhibiting or selling establishments where crafts-artists exhibited is quite broad and ranges from art or crafts fairs (46 percent) to cooperatives, mail orders, wholesalers and work group meetings (about 6 percent each). Among the more frequently used outlets are: own shop/studio (35 percent), through commissions (31 percent), art/crafts galleries (29 percent) and crafts shops (25 percent).

Art/crafts fairs are used by clay and glass or other media or combination of media crafts-artists to a greater extent than those who work in other media. Also respondents working in clay prefer their own shop or studio, crafts shops and art and crafts galleries compared to others. Commissions were more prevalent among metal workers (Table 19.16).

As the level of involvement declines from a primary to a leisure or occasional activity, the percentage showing or selling also declines. However, even among these less than full-time participants, art/crafts fairs, own shops or studio, crafts shops, commissions and art/crafts galleries are still the most important outlets. The subgroup which shows or sells the least includes those who have net losses of less than \$500,



TABLE IV.16
EXHIBITION INVOLVEMENT
SURVEY OF CRAFTS-ARTISTS, 1980

Percantage of crefca- artists shusing or salting their work through:   Art/Creft faire   45.2   41.0   57.7   40.1   47.5   82.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41.0   41	و المحادثي يعلى فسند في وفيد و معاون المحادث المحادث المحادث المحادث و المحادث و المحادث و المحادث و			Pris	nev Cra	ft Mad		<del></del>		Intencity of Crafts Involvement								Income from Crafts				
Percentage of crefts— Artists Pibar Clay Matel Wood Other Lasthar Hrs.   Hrs.   Clay Activity or Lass   Hrs.   Hrs.   Sional   Stop   Under Under Under State    Percentage of crefts— Artists envelop or crefts— artists enveloped or crefts— artists envelop	-				9 7 91			*********	Major				Activit	Y		Net Lo	s of	Net Ge	in of			
Percentage of crefts- artists showing or salting than the clay Mate   Wate   Wa		ALL						_				!!	. –			<b>\$500</b>	linder	linder	\$100			
Percentage of crefte- artists shuwing or selling their work through: ArtVCreft fairs															Other				plu			
articate shusing or selling their work through: through: Art/Creft fairs		Artists	Fiber	CLBY	Metel	Wood	Other	Lesther	Hrs.	Plus	ACTIVITY	or Less	PLUB	VCFIAITA	Utilei	Ptue	<del>4000</del>	<u> </u>	<u> </u>			
## alting their work through: ## comparison and thro	Percentage of crefce-																					
through: Art/Creft fairs Art/C	ertiete showing or														•							
Art/Creft feire 45, 2 41, 0 57, 7 40, 1 47, 5 62, 0 41, 6 57, 2 40, 1 47, 5 62, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52, 0 41, 6 52																						
Art Craft fairs 40.2 43.0 35.6 43.0 35.6 44.8 28.8 28.8 28.6 44.8 28.8 28.8 28.6 47.7 34.8 14.5 16.4 11.5 14.8 20.9 11.4 33.2 Crafts shops 24.7 21.0 38.2 24.4 30.4 10.0 46.7 47.7 34.8 14.5 16.4 11.5 14.8 20.9 11.4 33.2 26.0 11.5 18.8 20.8 11.5 14.8 20.9 11.4 33.2 28.0 25.7 23.4 4.1 5.5 3.9 7.5 10.8 4.0 16.5 Comparatives 8.5 7.2 8.9 3.5 5.8 2.8 7.1 12.8 8.6 8.8 8.8 3.7 3.8 4.1 2.6 5.1 3.9 7.5 10.8 4.0 16.5 Comparatives 8.5 7.2 8.9 3.5 5.8 2.8 14.7 34.2 28.0 25.7 23.4 4.1 12.6 24.7 14.5 24.0 27.9 11.8 41.1 Comparatives 10.5 24.2 45.5 4.7 12.8 26.5 13.4 11.1 19.1 27.1 28.5 13.9 15.2 24.2 28.6 25.7 28.8 4.1 2.6 24.7 14.5 24.0 27.9 11.8 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.7 41.1 28.1 28.1 28.1 28.1 28.1 28.1 28.1 2					40.4	47 E	62.0	41 G	63.0	60.0	59.9	36.5	51.5	23.5	37.2	43.2	33.8	58.0	63			
Our shop/studio												-		13.1	29.1	35.0	17.0	39.1	57			
Certe shops Char retail outlets 12.8 Char retail outlets 12.8 Char retail outlets 12.8 Comparatives 6.5 7.2 9.8 3.5 5.8 2.8 7.1 11.8 21.9 21.1 21.9 38.0 45.1 28.0 28.0 45.1 28.0 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 31.9 38.0 45.1 28.1 38.0 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 45.1 38.0 46.1 48.0 48.0 48.0 48.0 48.0 48.0 48.0 48.0											• -			11.5	14.8	20.9	11.4		41			
Comparatives   Comp													5.5	3.9	7.5	10.8	4.0	16.5	28			
Comperatives 30.3 25.0 38.0 45.1 29.1 31.0 38.0 54.1 52.8 45.4 12.6 24.7 14.6 24.0 27.9 11.8 41.1 Comperations 30.3 25.0 38.0 45.1 29.1 31.0 38.0 54.1 52.8 45.4 12.6 24.7 14.6 24.0 27.9 11.8 41.1 Anti-order 24.0 27.9 14.8 41.1 29.1 31.0 29.1 29.1 29.1 29.1 29.1 29.1 29.1 29.1													3.8	4.1	2.8	5.1	3.0	9.4	13			
Commissions 28.7 24.3 46.5 35.9 14.7 34.2 23.6 52.2 60.7 38.6 11.8 13.5 11.4 29.5 33.9 15.2 24.2 Art/Creft gaiteries 28.7 24.3 46.5 35.9 14.7 34.2 23.6 52.2 60.7 38.6 11.8 13.5 11.4 29.5 33.9 15.2 24.2 Mail orders 5.9 4.4 3.2 8.5 8.4 7.9 5.9 14.4 15.3 8.7 1.1 3.1 0.4 1.7 3.8 1.1 8.7 Moleselars 6.1 4.3 8.9 8.2 4.7 13.4 2.8 18.3 17.3 7.5 0.8 2.0 0.8 2.0 0.8 2.1 5.7 2.0 5.4 Work group eastings 6.0 9.7 2.9 2.4 4.0 2.9 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Work group eastings 8.0 9.7 2.9 2.4 4.0 2.9 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Work group eastings 6.0 9.7 2.9 2.4 4.0 2.9 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Work group eastings 6.0 9.7 2.9 2.4 4.0 2.9 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Work group eastings 6.0 9.7 2.9 2.4 4.0 2.9 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 9.1 7.2 3.2 Work group eastings 6.0 9.7 2.9 8.4 4.0 2.4 10.4 25.0 1.3 1.9 3.4 35.8 22.0 48.1 28.9 20.2 41.9 4.8 Work group eastings 6.0 9.7 2.9 8.6 430 370 305 284 327 459 666 382 Work group eastings 6.0 4.5 1.5 4.4 8.9 5.2 4.0 43.4 13.8 23.5 39.0 22.5 49.9 Work group eastings 6.0 4.5 1.5 1.5 1.5 1.5 4.8 5.6 1.9 0.7 1.9 0.2 1.8 3.5 0.9 1.8 Work group eastings 6.0 4.5 1.5 1.5 1.5 1.5 4.8 5.6 1.9 0.7 1.9 0.2 1.8 3.5 0.9 1.8 Work group eastings 6.0 4.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1	•	- • -							• -			-	24.7	14.5	24.0	27.9	11.8		56			
## Art Unders													13.5	11.4	29.5	33.9	15.2	24.2	51			
Meli orders Molesselars 6.1 4.3 8.5 8.2 4.7 13.4 2.8 19.3 17.3 7.5 0.8 2.0 0.8 2.1 5.7 2.0 5.4 Work group mastings 6.0 9.7 2.9 2.4 4.0 2.8 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Work group mastings 6.0 9.7 2.9 2.4 4.0 2.8 6.0 1.5 1.8 4.6 8.9 11.4 8.8 8.2 8.1 7.2 3.2 Number of Respondents 2625 966 434 423 363 202 139 439 256 430 370 305 284 327 459 656 382  Percentage of crefts-erists participating in following activities in lest 12 months: Art/Creft fairs 38.0 37.8 51.6 36.1 44.9 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 38.0 22.5 49.9 Art/Creft fairs 0.6 4.6 4.6 4.8 1.5 1.5 1.5 1.5 1.5 4.8 5.6 1.8 0.7 1.9 0.2 1.8 3.5 0.9 1.8 Missum 2.2 2.2 6.0 4.5 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9 0.4													3.1	0.4	1.7				14			
Work group mestings 6.0 9.7 2.8 2.4 4.0 2.8 6.0 1.5 1.8 4.6 9.9 11.4 8.8 8.2 8.1 7.2 3.2 Do not show/sell 20.8 26.5 13.4 15.4 20.4 10.4 25.0 1.3 1.9 3.4 35.8 22.0 48.1 28.9 20.2 41.9 4.8 Musber of Respondents 2625 966 434 423 363 202 139 439 256 430 370 305 284 327 459 656 382  Percentage of crefts— ertists perticipating in following activities in lest 12 months: Art/Creft feirs 38.0 37.8 51.6 36.1 44.8 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 38.0 22.5 49.9 1.8 Museum 2.2 2.2 6.0 4.5 1.5 1.5 1.5 1.5 4.8 5.6 1.9 0.7 1.9 0.2 1.8 3.5 0.9 1.8 Sales gellery 7.1 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9 0ther 6.0 4.6 7.1 6.9 6.3 5.5 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8	, ··· · · · · · · · · · · · · · · · · ·								7			0.8	2.0	0.8	2.1	5.7			16			
Work group sesting 20.8 26.5 13.4 15.4 20.4 10.4 25.0 1.3 1.9 3.4 35.8 22.0 46.1 28.9 20.2 41.9 4.8 Number of Respondente 2625 966 434 423 363 202 138 439 256 430 370 305 294 327 459 656 382  Percentage of crefts- srtists perticipating in following activities in lest 12 months:  Art/Creft faire 38.0 37.8 51.6 36.1 44.8 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 39.0 22.5 49.9 41.8 41.8 41.9 41.9 41.9 41.9 41.9 41.9 41.9 41.9					- • -						. • -		11.4	8.8	8.2	9.1			2			
Do not show/sell 26.8 966 434 423 363 202 139 439 256 430 370 305 294 327 459 656 382  Parcentage of crefts- artists perticipating in following activities in last 12 months:  Art/Creft fairs 38.0 37.8 51.6 36.1 44.9 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 39.0 22.5 49.9   One-person exhibits  Hussum 2.2 2.2 6.0 4.5 1.5 1.5 1.5 4.8 5.6 1.8 0.7 1.9 0.2 1.8 3.5 0.9 1.8  Sales gellary 7.1 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9   Other 6.0 4.6 7.1 6.8 6.3 5.5 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8   Group-invited exhibit  Hussum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.8 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8   Group competition  exhibit  Mussum 12.1 11.3 14.7 11.6 12.4 13.8 13.1 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9   Other 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8   Other 7.8 6.9 11.9 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.8   Other 7.8 8.16 6.0 8.3 9.8 9.8 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 24.4 39.9 18.2 30.3   Unjuried event 29.8 28.0 46.6 42.3 28.8 8.9 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 24.4 39.9 18.2 30.3   Unjuried community event 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 1.7 20.5 10.5 23.7 4.8 3.8 4.8 2.8 7.5 3.3 7.2 13.1 4.4 8.8 2.8 7.5 3.3 7.2 13.1 4.4 8.8 2.8 7.5 3.8 3.8 4.8 2.8 6.5 3.3 7.2 13.1 4.4 8.8 2.8 7.5 3.3 7.2 2.3 9.5 10.5 23.7 4.8 3.8 4.8 2.8 3.8 4.8 3.8 3.8 3.8 3.8 3.8 3.8 3.8 3.8 3.8 3							_	-			3.4	35.8	22.0	46.1	28.9		41.9		1			
Parcantage of crafts- artists perticipating in following sctivities in last 12 months: Art/Craft fairs  38.0 37.8 51.6 36.1 44.9 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 38.0 22.5 49.9  One-person exhibits  Museum  5.2 2.2 6.0 4.5 1.5 1.5 1.5 4.8 5.6 1.9 0.7 1.9 0.2 1.8 3.5 0.9 1.8  Sales gallary 7.1 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9  Other 6.0 4.6 7.1 6.8 6.3 5.5 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 6.9 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 6.9 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 6.9 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 6.9 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8 11.3 5.8 13.4 4.4 21.3 23.9 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8 11.3 5.8 13.4 4.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 5.4 11.3 14.7 11.6 12.4 13.9 13.1 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.8 9.7 11.8 11.1 16.	Do not show/sell											370	305	284	327	459	656	382	471			
in following activities in lest 12 months: Art/Creft feire 38.0 37.8 51.6 36.1 44.9 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 39.0 22.5 49.9  One-person exhibite Museum 2.2 2.2 6.0 4.5 1.5 1.5 1.5 1.5 4.8 5.6 1.9 0.7 1.9 0.2 1.8 3.5 0.9 1.8 Sales gallary 7.1 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9 Other 6.0 4.6 7.1 6.9 6.3 5.5 10.2 8.0 10.2 9.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8   Group-invited exhibit Museum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.8 10.3 4.5 8.3 1.8 11.1 18.1 4.7 9.8  Sales gallary 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 0.4	Parcentage of crafts-													-				٠				
in lest 12 months: Art/Creft feire 38.0 37.8 51.6 36.1 44.8 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 39.0 22.5 49.9  Art/Creft feire 38.0 37.8 51.6 36.1 44.9 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.8 23.5 39.0 22.5 49.9  One-person sxhibits Hussum 2.2 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.9  Other 6.0 4.6 7.1 6.8 6.3 5.5 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.3 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 10.2 8.0 10.2 8.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 5.8 14.7 11.6 12.4 13.8 13.1 14.8 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 11.3 14.7 11.6 12.4 13.8 13.1 14.8 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 11.3 14.7 11.6 12.4 13.8 13.1 14.5 12.5 8.0 11.7 3.2 2.3 9.5 13.9 3.8 5.8 13.4 4.5 13.1 14.5 12.5 8.0 11.7 3.2 2.3 9.5 13.9 3.8 5.8 13.8 4.5 13.1 14.5 12.5 8.0 11.7 3.2 2.3 8.7 9.8 3.8 4.5 13.1 11.1 16.8 3.7 9.7 9.7 9.8 3.8 4.5 13.1 12.8 4.1 8.3 14.1 14.1 14.1 14.1 14.1 14.1 14.1 14	ertiete perticipating																					
Art/Creft feirs 38.0 37.8 51.6 36.1 44.8 57.2 36.2 60.6 59.8 52.4 28.0 43.4 13.0 21.1 13.0 21.0 21.8 3.5 0.9 1.8 41.8 420.2 13.8 17.1 20.8 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 44.8 19.3 24.1 12.0 24.3 8.7 25.6 15.0 29.1 18.3 2.0 11.7 20.5 10.5 23.7 44.8 29.8 29.0 46.6 42.3 28.9 14.8 11.8 12.4 12.8 28.0 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 44.8 20.2 13.8 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 4.8 19.3 11.8 2.8 7.2 4.4 39.8 18.2 20.3 11.8 12.1 12.2 24.3 29.9 10.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0 24.3 20.0		ı																				
One-person swhibits  Hussum  2.2  2.6  4.5  1.5  1.5  1.5  4.8  5.6  1.9  0.7  1.9  0.2  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.5  0.9  1.8  3.6  3.7  3.8  3.8  4.8  3.8  3.8  4.8  3.8  3.8	• • • • • • •	20.0	27.0	E4 6	36 1	AA 9	57.2	36.2	60.6	59.8	52.4	28.0	43.4	13.8	23.5	39.0	22.5	49.9	83			
Hussum 2.2 2.2 6.0 4.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1		38.0	3/.0	01.0	30 . 1	77.0	U/ .L	00 12	,-										_			
Sales gellery 7.1 5.4 14.9 10.2 4.7 14.9 5.8 16.8 23.3 8.4 1.3 2.1 1.1 6.8 8.5 1.5 7.8 0ther 6.0 4.6 7.1 6.9 6.3 5.5 10.2 8.0 10.2 9.4 3.8 4.8 2.2 5.8 7.2 4.4 5.8 6roup-invited exhibit  Museum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.8 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8 11.3 Sales gellery 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 0ther 12.1 11.3 14.7 11.6 12.4 13.9 13.1 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 Group competition exhibit  Museum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8 13.9 3.8 5.8 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 0ther 7.8 8.1 6.0 8.3 9.8 7.0 8.0 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 0ther 7.8 8.1 6.0 8.3 9.8 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3 Juried event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3 Unjuried community event 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 13.7 13.8 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 13.7 13.8 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 13.7 13.8 13.8 13.8 13.8 13.8 13.8 13.8 13.8				e 0	4 5	4.5	1.5	1.5	4.8	5.6	1.9	0.7	1.9	0.2	1.8				8			
Sales gallary Other 6.0 4.6* 7.1* 6.9* 6.3* 5.5* 10.2* 8.0 10.2 9.4 3.8 4.8 2.2 5.8 7.2* 4.4* 5.8*  Group-invited exhibit  Mussum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.9 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8 Sales gallary 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 Sales gallary 12.1 11.3* 14.7* 11.6* 12.4* 13.9* 13.1* 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 Group competition exhibit Mussum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8 4.8 3.7 4.3 12.8 4.4 8.8 3.7 3.8 4.8 2.2 5.8 7.2* 4.4* 5.8* 6.8 6.8 11.1 16.1 16.1 16.8 9.7 11.3 11.3 11.3 11.4 11.3 11.4 11.4 11.4									• -			1.3	2.1	1.1					21			
Group—invited exhibit  Museum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.9 10.3 4.5 8.3 1.8 11.1 16.1 4.7 9.8  Museum 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3  Sales gallary 12.1 11.3* 14.7* 11.6* 12.4* 13.9* 13.1* 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9  Group competition  Exhibit  Museum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8  Museum 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5  Sales gallary 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5  Other 7.8 8.1* 6.0* 8.3* 9.9* 7.0* 8.0* 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3  Other 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3  Juried event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3  Juried community  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Juried Community Event 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2										-	9.4	3.8	4.8	2.2	5.8	7.2	4.4*	5.8*	5			
Hussum 10.3 11.6 14.0 18.3 5.8 13.4 4.4 21.3 23.9 10.3 4.5 5.0 0.8 14.1 22.9 7.1 11.3 Sales gallary 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.9 7.1 11.3 0ther 12.1 11.3* 14.7* 11.6* 12.4* 13.9* 13.1* 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 Group compatition exhibit  Hussum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8 10.8 10.8 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 11.1 16.8 9.7 11.8 11.9 11.9 11.9 11.9 11.9 11.9 11.9			₹,0	. /.!	0.5	0.0	0.0				-											
Seles gellery 13.7 13.6 30.7 20.0 7.7 25.4 8.8 32.8 32.4 18.8 2.6 5.0 0.8 14.1 22.8 7.1 11.3 Other 12.1 11.3 14.7 11.6 12.4 13.9 13.1 14.8 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 Group competition exhibit  Museum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8 Museum 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 8.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 Seles gellery 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 8.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 Other 7.8 8.1 6.0 8.3 9.9 7.0 8.0 13.7 8.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3 Other 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3 Unjuried community event 8.8 8.9 14.9 11.9 6.3 17.9 8.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 Juried Community event 8.8 8.9 14.9 11.9 6.3 17.9 8.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2	- · · · · · · · · · · · · · · · · · · ·		44 6	44.0	18.3	5.8	13.4	4.4	21.3	23.9	10.3	4.5	8.3	1.8					19			
Sales gettery 12.1 11.3* 14.7* 11.6* 12.4* 13.9* 13.1* 14.9 15.3 18.7 10.8 9.8 4.5 11.1 16.8 9.7 11.9 Group competition exhibit  Museum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8 Museum 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 Other 7.8 8.1* 6.0* 8.3* 9.9* 7.0* 8.0* 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3 Unjuried event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3 Unjuried community event 8.8 8.9 14.9 11.9 6.3 17.8 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 Juried Community event 8.8 8.9 14.9 11.9 6.3 17.8 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2										32.4	18.8	2.6	5.0	0.8					32			
Group compatition  axhibit  Mussum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8  Mussum 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5  Sales gallary 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5  Other 7.8 8.1 6.0 8.3 9.9 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3  Jurisd event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3  Unjuried community  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7										15.3	18.7	10.8	9.8	4.5	11.1	16.8	9.7	11.9	13			
exhibit         Museum       6.7       6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 9.5 13.9 3.8 5.8         Museum       6.7       6.9 11.5 6.4 3.0 10.9 3.6 13.5 8.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5         Seles gellery       6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 8.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5         Other       7.8 8.1 6.0 8.3 9.9 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3         Juried event       29.8 28.0 46.6 42.3 28.8 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3         Unjuried community       8.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7         event       16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.5 3.3 7.2 13.1 4.4 8.2         Juried Community event 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2		12.1	11.5	17./	11.0	16.	,,,,										1					
Mussum 6.7 6.9 11.9 13.1 2.5 10.4 5.1 14.5 12.5 8.0 1.7 3.2 2.3 3.0 13.0 3.8 4.5 Seles gellery 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 4.3 12.8 4.1 8.3 Other 7.8 8.1 6.0 8.3 9.9 7.0 8.0 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3 Jurisd event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3 Unjuried community event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 event 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2																			4.0			
Seles gellery 6.2 6.9 11.5 6.4 3.0 10.9 3.6 13.5 9.4 9.7 1.7 1.2 0.8 9.7 9.8 3.8 4.5 Other 7.8 8.1* 6.0* 8.3* 9.8* 7.0* 8.0* 13.7 9.7 3.1 6.4 8.8 3.7 4.3 12.8 4.1 8.3 Juried event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3 Unjuried community event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 19.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7 event 9.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2 3.5 3.5 3.5 3.5 3.5 3.5 3.5 3.5 3.5 3.5		6 7	6 0	44 0	13 1	2.5	10.4	5.1	14.5	12.5	8.0								12			
Sales gittery 0.2 0.4.4 12.8 4.1 8.3 0.4 0.4 0.4 0.4 0.4 0.4 0.4 0.4 0.4 0.4			-						13.5	9.4	9.7	1.7	1.2		_ •		_		12			
Unied event 29.8 28.0 46.6 42.3 28.9 43.8 24.8 54.9 60.8 41.4 12.0 24.3 8.7 24.4 39.9 18.2 30.3  Unjuried community  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 18.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Event 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 18.3 26.2 14.0 16.4 7.4 11.7 20.5 10.5 23.7  Event 2.8 7.5 3.3 7.2 13.1 4.4 8.2  Juried Community Event 8.8 8.9 14.9 11.9 6.3 17.9 8.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2										9.7	3.1	6.4							10			
Unjuried community  Sevent  16.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.4  18.5  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18.6  18					-				54.9	60.8	41.4	12.0	24.3	8.7	24.4	39.9	18.2	30.3	58			
sysht 16.4 18.4 20.2 13.8 17.1 20.9 16.1 24.8 18.3 20.2 14.0 10.4 7.5 17.1 14.4 8.2 Jurisd Community sysht 8.8 8.9 14.9 11.9 6.3 17.9 9.5 15.8 19.3 11.8 2.8 7.5 3.3 7.2 13.1 4.4 8.2		20.0	E0.0	·-	72,0				_	•								00 7	-			
Syent 10.4 10.4 10.4 10.4 10.4 10.4 10.4 10.4	•	46 4	18.4	20.2	13.8	17-1	20.9	16.1	24.8	19.3	26.2			•					22			
JUPTED COMMUNICA BABIL D.D. 140 140 140 140 140 140 140 140 140 140							-		15.8	19.3	11.8								19			
Number of Respondents 2196 736 393 371 298 186 110 434 254 419 255 250 188 255 457 458							186	110	434	254	419	255	250	158	256	437	408	380	47			

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Bessd on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted parcentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those parcentages or percentage distributions followed by an exteriak [\*].



about 42 percent. Half this proportion, 20 percent, among those with larger net losses do not show or sell (Table IV.16).

Concentrating more on exhibiting activity, crafts-artists participate in art or crafts fairs and juried events to a much greater extent than one-person exhibits, group-invited exhibits, group competition exhibits, unjuried community exhibits or juried community events. 1/ The importance of some of these are obviously a function of their nature; one-person exhibits are very restrictive while group-invited and competitive exhibits are also limited. However, it is worthwhile to note that larger shares of crafts-artists participated in group-invited exhibits compared to the more open group competition, with no clear preference within either category for museums, sales galleries or other types of exhibits.

There are strong patterns among the subgroups for participation in the various exhibiting activities. Crafts-artists working in clay and, to a lesser extent, glass or other media or combinations of media are much more active in all types of exhibits and both prefer galleries over museums for all types of exhibits. The greater the level of involvement, the higher the percentages of participation in exhibits of all types. Art/crafts fairs and juried events still attract crafts-artists at the leisure and occasional activity level. Among income groups the pattern noted earlier returns: those with large losses or gains are generally more active than crafts-artists with nore moderate results (Table IV.16).



 $<sup>\</sup>frac{1}{2}$  The response pattern results from respondents categorizing their activites into specified, but undefined types of exhibits.

Crafts-artists were asked what they thought were the best places to exhibit. Those who exhibit list equal preference for art/crafts fairs and art/crafts galleries, 34 percent in each case (Table IV.17). The next level of preference is for crafts shops (23 percent) or their own crafts shop (17 percent). Following the pattern of their stronger orientation toward exhibition, clay crafts-artists "voted" for a wider variety of outlets, but ended up ranking fairs and galleries highest. In fact these are the two most highly ranked in each sub-category.

The most important problem mentioned by all crafts-artists regarding exhibits is "too few" (15 percent) followed by "poorly informed communities" (13 percent). Other concerns are no appropriate space to exhibit/sell (9 percent), not enough display space (8 percent) and lack of time to exhibit (6 percent). There is little statistically significant variation among the sub-categories of craft-artists and the order of the reasons mentioned above rarely changes, (Table IV.17). The most noticeable trend is an increase in the desire for more display space for crafts-artists who list crafts as their major activity or derive more than \$1,000 per year in net income from crafts.

#### F. SELLING CRAFTS

As we have seen, one of the important rationales for crafts work, although secondary to intangible benefits such as offering a means of creative expression, is its contribution to present or potential income. However, selling activities are also the manifestations of a desire to achieve recognition and acclaim in the marketplace. These two motives, creative satisfaction and recognition in the marketplace, can result in unexpected patterns. The extent of commercially successful crafts-artists whose production is oriented toward multiples of the

ERIC Full text Provided by ERIC

IV. 37 116

# TABLE IV.17 ATTITUDES TOWARDS PLACES TO EXHIBIT AND PROBLEMS SURVEY OF CRAFTS-ARTISTS, 1980

•			Prim	ery Çre	ft Med	1 0			Inter	neity of C	refte Inv	olvement	<u>t</u>		Income from Crefts				
•									Activi:	<u>ty</u>	Leteure				Net Lo	ss of	Net Gein		
	All Crefts- Artists	Fiber	Clay	Metel	Wood	Gless, Other	Papar, Leather	Under 40 Hre.	40 Hrs. Plus	2nd Activity	10 Hrs. or Less	10 Hrs. Plus	Occa- sions! Activity	_Other	\$500 ptus	Under \$500	Under \$1000	\$1000 plue	
Percentage of exhibiting crefts-ertists who think sech of the following ere the best places to exhibit their work:																			
Art/creft feire	34.5	29.1	43.6	33.3	36.9	49.8	27.9	49.9	50.5	45.2	27 <b>.4</b> 7 <b>.</b> 7	41.9 12.7	15.4 5.3	21.4 13.1	30.9 18.4	19.2 7.9	48.8 19.0	49.0 30.7	
Own craft shop	17.9	12.5 20.6	26.1 32.8	20.5 25.2	18.0 17.3	23.1 27.0	15.9 15.3	34.4 38.0	39.5 31.2	25.8 36.4	14.1	23.0	8.7	14.7		13.1	30.3	33.2	
Creft shope Cooperatives	22.7 6.9	7.9*	8.6*	6.9	4.6*	4.1*	7.9*	10.0	12.7	11.1	3.4	6.1	2.3	5.8	8.6	3.7	9,4	11.2	
Art/creft gelleries	34.4	32.1	45.9	41.1	21.2	37.2	39.1	60.2	59.4	46.7	18.6	27.0	11.9	32.7	41.6	17.8	38.2	56 .	
Art school or	• • • • • • • • • • • • • • • • • • • •					- •	- -											_	
college	6.4	6.3	7.8	9.4	2.3	3.3	12.6	10.6	9.9	6.0	3.0	6.1	1.3	11.9	10.1	4.7	4.2	7.	
College or university	9.8	9.0	13.1	11.5	3.6	6.9	9.3	11.5	15.9	9.8	3.3	6.1	3.4	17.2 5.9	13.9 6.9	5.7 2.3	6.6 4.4	11. 8.	
Other	4.7	5.5*	2.7*	5.4*	4.0*	6.1*	4.1*	7.0	9.3	7.3	2.7	3,1	0.9	0.8	0.8	2.3	7.7	0.	
Number of perpondents who exhibit	1731	583	334	282	225	154	83	434	254	419	255	250	158	256	319	282	272	419	
Parcentage reporting the following main problems in exhibiting																			
Too few exhibits	14.9	14.5	21.0	16.0	7.8	20.7	11.5	23.1	18.2	23.8	11.1	12.6	4.5	14,5	18.9	9.8	19.6	17.	
Not anough display			÷		0.00	0.48	0.48	13.0	17.4	10.0	4.8	5.5	4.1	4.9	8.4	8.2	8.2	13.	
epece	7.9	7.7*	8.7*	6.2*	8.2*	9.4*	9.1*	13.0	17.4	10.0	4.0	0.0	7.1	7.5	0.7	U.L	•		
No appropriate ex- hibit/selling apace	9.0	9.1	11,1	11.2	3.9	10.9	11.9	14.1	19.4	12.9	5.4	6.1	4.2	8.7	10.2*	6.4*	9.6*	12.	
Community poorly	0.0	٥.,	••••		-,-	,,,,,,	****			•-									
informed about		44.00	40.0	12.9*	10.4*	19.1*	13.1*	18.9	21.8	20.1	9.8	7.3	7.5	9.4	14.7	7.3	15.3	15.	
exhibits Lack of time to	12.7	11.3*	16.0*	12.8*	10.4	18.1	13.1	10.5	£1,0	EU.1	3.0	7.0	, ,,,	-,-		-			
exhibit	5.6	5.9*	5.8*	5.3*	5.9*	5.3*	6.2*	6.5*	5.2*	10.6*	2.7*	5.6*	2.7*	8.1*	7.4*	3.4*		7.	
Costs of exhibiting	1.9	1.9*		2.9*	0.9*		3,1*	2.9*	5.2*		1.2*	0.9*		2.0*	3.3*	0.9*	2.1*	2.	
Lack of public			-					_ 4						0.64		4 04	0.4	5.	
understanding	2.7	2.2*		3.3*	2.7			5.9*	5.2*		0.6*	3.6* 4.9*		2.0* 4.4*	2.7* 5.7*			7.	
Other problems	5.8	5.5*	6.7*	6.6*	5.2*	7.6*	4.1*	10.1*	7.1*	9.4*	3.3*	4.8*	1.5*	4,4*	8.7	2.0	· /.J*	/ •	
Number of respondents	0000	959	429	418	363	199	139	439	256	430	370 -	305	294	327	455	655	378 ,	468	
who exhibit	2600	808	468	410	303	100	138	700	200	700	5, U						,		

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Based on calculation of X<sup>2</sup> statistics, in 98 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts-artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an esterisk (\*).

Based on the number of respondents who exhibited in the Lest 12 months.



same work is one example. An understanding of the problems and attitudes involved in selling crafts will augment knowledge of crafts-artists and their work.

Slightly over two-thirds (68 percent) of all crafts-artists produce works for sale. There is a wide fluctuation in this activity when subgroups of crafts-artists are examined. By media, clay, metal, glass or other media or combinations of media are categories where larger shares of crafts-artists produce works for sale (Table IV.18). The same pattern is evident for those whose primary source of income is from crafts sales, although the percentage level is much lower, 23 percent for all crafts-artists and about 30 percent for the clay, metal, glass or other media or combination subgroups.

Major involvement in crafts is reflected in very high proportions (98 percent) of these crafts-artists producing crafts for sale. The proportion whose primary income is from crafts among respondents who have crafts as their major activity is a multiple of 2 or 3 times the same proportion for all crafts-artists. From 35 to 66 percent of the crafts-artists whose participation is for leisure or occasional activity produce crafts for sale; much lower shares (less than 5 percent) have crafts as their primary source of income (Table IV.18).

Most craft works are produced in response to the crafts-artists' own standards of form and expression rather than demands of the market. The averages for all crafts-artists are about 75 and 20 percent respectively, with little variation among the subgroups. Crafts-artists, as a rule, do not engage commercial dealers, agents or other personal representatives to assist in selling their crafts; only 12 percent engaged any of those just mentioned to assist them in marketing their crafts.



TABLE IV.18

INCIDENCE AND METHOD OF SELLING CRAFTS SURVEY OF CRAFTS-ARTISTS, 1980

			Pr	imery C	reft H	edie			Inte	neity of	Income from Crefts							
	All Crefte- Artists	Fiber	Cley	<u> Hetel</u>	Wood		Paper, Leather	Hejor Under 40 Hre.	Activi 40 Hre. Plue	<u>ty</u> 2nd Activity	Laieure 10 Hre. or Leee	10 Hre,	Occa- eionel	Other	Net L \$500 plue	oee of Under \$500	Not Go Under \$1000	#1000 plue
Percentage of crafts—ertiate producing works for eals Number of respondents	67.8 2629	57.4 968	83.4 435	80.7 423	64.9 364	82.5 203	59.6 139	96.0 439	97 .4 256	94.6 431	<b>44.</b> 6 370	<b>66.</b> 3	35.4 296	56.2 328	62. <b>8</b> <b>4</b> 80	38.3 659	91.9 382	96 .4 476
Percentege of crefte-ertists with primery income from eals of crefts Number of respondents who eall crefts	23.1 1910	14.4 602	30.4 377	29.3 351	24.2 248	31.8 177	° 17.5	51.8 424	80.4 252	7.2 409	0.5 183	1.9 215	3.0 110	4.1 209	1 <b>9.4</b> 316	3. <b>8</b> 269	10.0 356	51.7 457
Percenteges producing works for each primerity in response to: Their own stenderds of form end expression Demends of the merket Other fectors TOTAL Number of respondents who	73.5 19.5 7.0 100.0	72.3 21.3 <u>6.4</u> 100.0	78.8 12.8 8.8 100.0	19.3 <u>9.2</u> 0.100.0	23.0 4.9	22.3 7.1 100.0	62.5 29.0 <u>8.5</u> 100.0	65.3 20.8 13.9 100.0	66.6 24.0 7.4 100.0	74.5 18.9 6.6 100.0	76.9 21.5 7.8 100.0	76.3 20.0 3.7 100.0	13.2	81.7 14.6 3.7 100.0	93.0 11.2 5.8 100.0	14.0	73.5 22.0 4.5 100.0	25.1

SOURCE: Survey of Crafte-Artists who are Members of Crafts Organizations, Methematics Policy Research, Inc.

Besed on calculation of  $x^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exteriak  $\{^{+}\}$ .



The problems mentioned by crafts-artists when selling their crafts range from "presenting work to clients" (18 percent), to "too few outlets" (16 percent), "lack of business skills" and "poorly informed of existing outlets" (13 percent each). There is little statistically significant variation in selling problems when crafts-artists are divided into media groups. There is an observable tendency for those devoting more time to crafts (those for whom crafts is a primary or secondary activity) to report a higher level of problems. The same is true for crafts-artists with net income gains from crafts sales (Table IV.19).

Crafts-artists feel the best places to sell their works are art or crafts fairs or crafts galleries (both about 33 percent). A slightly smaller share (23 to 25 percent) reported their own shop or crafts shops in general. About 13 percent mentioned other retail shops. There is some variation by type of media; art and crafts fairs are more heavily favored by clay and glass or other media or combination of media crafts-artists, while galleries are more productive for metal workers.

Galleries are less attractive to those who work in fiber and least attractive for those working in wood. All types of selling outlets attracted more responses from those for whom crafts is a primary or sect dary activity or who have a net gain in income from crafts.

### G. FINANCIAL SUPPORT AND CRAFTS INCOME AND EXPENSES

Crafts is an activity that has only recently been assisted by formal financial support from the National Endowment. Private sources have always been an alternative although not one frequently used.

Cooperative efforts of crafts-artists and the public are also a factor, both from a perspective of cooperative purchasing, but also trading for

			Prim	ery Cre	ft Med	1.			Inter	neity of (	Crefte Inv	lvement	<u> </u>		Income from Crefte				
	4							Major	Activi	ty	Leisure /		L		Net Lo	es of	Net Ge	in of	
	All Crafts- Artiete	Fiber	Cley	Metal	Wood		Peper, Leather	Under 40 Hre.	40 Hre. Plue	2nd Activity	10 Hrs. or less	10 Hre. Plue	Occa- eionel Activity	Other	\$500 ptue	Under \$500	Under \$1000	\$1000 plue	
Percentage of crafts- artists reporting the following main problems												_							
in ealling: Too few outlete	16.4	14.3	23.9	16.8	10.8	24.4	15.3	24.9	20.6	26.8	12.3	11.3	6.8	14.5	20.7	9.8	24.4	19.	
Inaufficient display		r 0\$	0.08	5.4*	4.8*	9.7*	6.4*	9.6	15.5	8.3	4.8	3.8	1.0	4.4	7.5	2.9	7.5	10.	
No appropriate apace Community poorly in-	6.1 8.3	5.6* 7.1*		12.3*	5.4*			10.7	16.4	12.6	4.9	5.7	3.8	10.2	10.7	5.3	10.4	10.6	
formed of existing outlets	12.8	10.7*	16.4*	14.3*	12.7*	16.8*	11.1*	19.8	25.2	18.8	9.3	7.4	5.7	10.8	18.4	6.2	14.4	17.	
Communicating with cliente	9.8	9.8*	11.4*	11.3*	6.B*	9.4*	10.3*	16 .8	21.0	13.8	3.8	7.4	4.5	8.2	9.9	5.1	8.0	16.	
Presenting work to	18.1	17.5*	i9.2*	22.2*	14.4*	19.4*	22.2*	30.0	33.8	26.0	9.6	15.3	7.3	15.2	20.3	10.8	15.9	29.	
Leck of bueiness ekills Leck of time to	13.2	12.6*	15.6*	16.6*	9.4*	11.3*	20.6*	31.2	19.0	18.8	5.8	8.0	3.7	10.2	16.3	8.8	12.2	22. 8.	
produce Other problems	5.4 7.4	3.7* 6.1*		5.1* 10.5*	2.5° 6.7°			4,9 12.8	6.0 12.8	10.9 11.3	2.2 5.2	8.8 <b>4.</b> 1	2.3 1.9	4.6 7.9	4.3 10.3	2.8 4.4	7.3 8.0	12.	
Number of respondents who eall or exhibit crefts	2540	939	418	409	355	196	136	425	246	416	359	585	283	312	<b>44</b> B	841	377	451	
Percentage who think each of the following ere th best ways to sall their	•																		
work: Art/craft fairs	33.7	27.7	45.4	30.5	33.9	50.4	27.9	48.2	53.7	46.8	25.4	38.3 16.7	12.8 8.0	23.4 23.2	30.2 25.5	19.1 10.9	47.0 30.8	51 . 38 .	
Own chop	25.4	17.4	37.5	31.4	23.6	33.8	19.1	43.9	52.5	36.0	11.0	19.9	11.7	17.1	18.8	14.4	32.8	37	
Creft chops	22.8	19.2	36.1	24.2	19.2	27.3	15.3	36.5	34.6	34.2 19.5	16.8 5.7	9.8	3.6	6.8	13.4	8.5	14.5	55	
Other retail outlets	12.B	12.8*		13.2*	9.4			26.3	24.5 13.3	10.3	3.1	4.9	1.3	5.3	8.4	3.5	7.5	11	
Cooperatives	6.7	7.1*		5.0*	4.1			12.1 56.5	56.4	44.8	16.3	23.5	10.8	32.7	39.1	18.8	35.1	53	
Art/creft gelleries	32 .1	27.0	49.1	39.8	19.7	35.8	34.7 5.8*	12.1	10.8	9.0	2.1	3.2	0.8	3.0	4.3	1.1	B.4	8	
Mail ordering	5.4	4.8*		5.9*	5.4° 4.3°			15.4	15.2	8.1	1.4	1.7	0.8	2.8	8.5	2.1	6.2	12	
Whoteseters	5.7	4.6		5.9* 1.1*		• -		1.5*	2.24		4.1*	3.7*		2.4	1.5	3.1			
Work group meetings	2.5	3.2° 8.4°	9.0*	9.8*	4.3			20.1	13.1	15.7	3,1	2.4	3.7	8.1	14.8	2.8	8.2		
Agent/Oseler Other	9.1 3.2	3.5		Ø.4*				5.8	8.8	2.8	5.2	3.4	4.0	1.7	3.5	1.1	8.0	3	
Number of respondents who sell or exhibit crefts	2595	958	432	418	356	505	139	437	255	424	368	297	585	321	455	B47	379	47	

Beend on celculation of X<sup>2</sup> etatistice, in 99 out of 100 cases, the weighted percentages in this table vary from the percentages for all pertiate by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by ERIC seriek (\*).

other crafts or for goods and services. Teaching crafts is also prevalent. The major subsidy of crafts work is income of crafts-artists from non-craft sources or from other family members.

Between 7 and 10 percent of all crafts-artists have received a grant or fellowship (mostly from the federal government) or applied for a loan from a bank for crafts purposes (about 5 percent or three-quarters of those who applied received a loan). About 9 percent purchased materials and/or equipment cooperatively with other crafts-artists or exchanged crafts they produced for living necessities or health care (Table IV.20). The frequency of support from these sources generally increased as crafts involvement approached full-time or crafts-artists earned net income from crafts.

Grants and fellowships were received by 14 percent of craftsartists working in metal and 18 percent who worked in paper or leather.

Metal and glass or other media or combinations of media crafts-artists
applied for loans more often and received them a larger percentage of
the time than crafts-artists in most other media. Fiber and clay workers
were more successful in cooperative purchases of materials and/or
equipment. Clay, metal and glass or other media or combinations of media
crafts-artists exchanged crafts for goods and services to a greater
extent than others. About a quarter of all crafts-artists exchanged
their crafts for those of other crafts-artists.

Two monetary measures were obtained in the survey; one is the gross level of receipts, the other is net income--adjusting receipts for the costs of production and sales. Gross income from crafts sales in the year prior to completing the questionnaire was about \$2,500 for all crafts-artists. This was supplemented by an average of \$300 from crafts-

TABLE IV.20 FINANCIAL SUPPORT RECEIVED AND COOPERATION WITH OTHERS SURVEY OF CRAFTS-ARTISTS, 1980

		Primary Craft Hadia Intensity of Crafts Involvement										Income from Crefte						
								Hajor	Activi	ty	Leieure	Activi t	<b>Y</b> ,		Net Lo	oo of	Not Go	in o
	All Crefte- Artiete	Fiber	Cley	Hetel	<u>Wo</u> od	Glese, Other	Peper, Leether	Under 40 Hre.	40 Hre. Plue	2nd Activity	10 Hre. or Less	10 Hre. P <u>lue</u>	Occa- eionel Activity	Other	\$500 plue	Under \$500	Under \$1800	
Percentege who have received a grent or fallowship Number of respondents	8.3 2629	<b>4.</b> 9 966	9.5 436	14.0 424	7.7 364	6.1 203	17.9 139	13.7 438	14.6 256	10.4 430	3.4 371	1.9 305	2.7 298	13.0 326	11.9 460	3.2 659	7.2 386	16 <b>.</b> 475
Percentage receiving sesistence from: Netional Endowment for the Arts	0.8	0.1	1.4	1.8	0.3	1.2	2.3	1,4	2.1	0.9	-0.2		_	" 1 <b>.</b> 6	1.0		0.9	1.
Other federal agency State agency Local agency Foundation	6.1 4.0 1.5 3.0	3.5 2.0 0.7 1.8	6.4 4.2 2.3 4.0	9.8 6.5 2.0 4.9	6.8 5.0 2.4 3.0	4.6 3.6 1.5 3.0	15.4 10.9 1.0 1.0	11.2 8.5 2.5 5.7	9.3 7.2 2.6 6.4	7.4 4.4 2.5 4.7	2.7 0.9 0.6 0.9	1.9 0.5 0.4	2.2 2.0 	8.5 6.2 2.4 4.4	9.9 6.3 1.8 3.8*		5.6 2.3 2.1 1.7*	12. 8. 2. 5.
Number of respondents	2626	965	436	422	364	203	138	438	255	429	371	305	297	323	<b>46</b> ,0	657	380	475
Percentage applying for a loan for crafts purposes Number of respondents	7 <b>.4</b> 2624	3.5 961	8.3 434	12.8 <b>4</b> 23	7.9 363	17.9 203	<b>4.9</b> 139	18.3 438	23.8 255	7.9 431	0.5 365	1.8 307	0.5 297	6.3 326	10.4 458	1.4 656	3 <b>.9</b> 382	16 . 47 5
Percentage receiving e loan Number of respondents	73.9 238	67.3 47	68.3 48	76.1 59	84.4 29	82.5 41	18.1 4	78.6 83	81 .0 66	75 <b>.</b> 2 42	16.3 4	72.8 6	71.7 2	73.7 25	69.4* 62	55.9* 12	91.5* 18	85 83
Crefts-ertists purchasing equipment or material in cooperation with	9																	
othere: Meteriele Equipment	8.5 1.0	10.2 0.9	13.1 1.7	8.4 1.0	1.7 1.6	5.3	9.8	8.8 1.5	13.2 1.4	11.1 0.8	4.7 1.3	9.0 0.4	7.7	13.3 2.6	9.D 1.7	7.4 0.8	7.8 1.0	11
Soth Number of respondents	3.8 2613	3.6 958	5.3 433	3.2 421	2.6 364	5.0 201	2.9 140	5.7 433	7.0 255	4.7 428	2.5 368	2.0 303	1.5 297	6.9 237	6.2 <b>45</b> 9	2.4 857	2.4 378	47-
Percentage exchanging their crafts for:																		
Living necessities Health core	6.8 5.0 26.3	3.6 2.3 22.5	10.8 9.2 28.1	9.4 6.9 30.6	4.8 2.4 20.0	10.1 10.4 30.3	8.1 4.4 23.1	13.6 13.1 49.7	18.1 17.7 53.1	9.4 5.4 39.3	2.5 0.5 13.7	3.6 0.3 20.3	1.4 0.7 13.2	6.4 3.1 26.1	5.8 3.9 30. <b>8</b>	2.8 1.1 13.2	5.4 3.9 27.2	14 14 47
Crefte end/or ert Number of respondents	2609	95 <b>8</b>	433	417	362	200	140	435	254	428	365	305	298	326	454	654	378	47

SOURCE: Survey of Crafts-Artists who are Hembers of Crafts Organizations, Mathematics Policy Research, Inc.

on celculation of X<sup>2</sup> statistics, in 98 out of 100 cases, the weighted percentages in this table very from the percentages for all in-artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by tarisk (\*).

related sales (materials, tools, etc.) and \$1,100 from teaching crafts. Euring the same period, all crafts-artists earned an average of \$8,000 from non-craft sources. On this basis, crafts income from all sources is about one-third of the gross personal income of crafts-artists (Table IV.21). The percentage becomes even smaller when the average gross crafts and non-craft income of other household members is considered. The share of gross household income attributable to crafts income of crafts-artist respondents is about 13 percent. This drops to about 1 percent when crafts-related expenses, averaging \$2,627, are considered.

From this perspective, crafts activity is much more likely to be a drain on household income than a contribution. However, for those who consider crafts their primary or secondary activity, the share of gross household income is in the 28 to 42 percent range and the contribution to net income is 4 or 5 percent. The contributions to gross and net household income for crafts-artists with high net income from crafts are almost 40 and 20 percent respectively.

Teaching is a larger source of net income than crafts-related sales, on the average \$1,100 compared to \$300 (Table IV.21). About 43 percent of all crafts-artists teach in a variety of settings. The most

<sup>1/</sup>Not all respondents answered all income questions; accordingly, it is difficult to calculate this and other percentages exactly without extensive tabulations. Approximations have been calculated using the average (mean) amounts reported for each income component in the accompanying tables.

TABLE IV.21

### INCOME (GROSS AND NET) FROM CRAFTS AND DTHER SOURCES SURVEY OF CRAFTS-ARTISTS, 1980

			Pri	mery Cr	oft Med	ie .		Intensity of Crefte Involvement					Ir	come fr	om Cref	to	
						b	Major Activity Leieure Activ						Net Less of		Not Goin of		
	All Crofts- Artists	Fiber	Cley	Motel	Wood	Gless, Other	Peper, Leather	Under 40 Hrs,	40 Hrm. Plum	2nd Activity	10 Hrs. or Less	Hrs.	eionel Activity Other	\$500 plus	Under \$500	\$1000	\$1000 pLui
iverege (meen) gross income of crefts- ertists from					,				444500	<b>4</b> 4700	\$ 183	\$ 620	\$ 119 <b>\$</b> 3082	<b>\$</b> 1664	<b>\$</b> 185	<b>\$</b> 906	<b>\$</b> 953
Oun crafts seles lumber of respondents Crafts-related	\$ 2493 2348	\$ 1034 884	\$ 4118 393	\$ 4358 369	\$ 2199 328	\$ 5991 178	\$ 1254 125	<b>\$ 6142</b> <b>40</b> 0	\$11568 229	<b>\$ 1766</b> 377	342	269	292 205	460	660	382	47
sources umber of respondents Teaching crefts umber of respondents	\$ 303 2327 \$ 1109 2352	# 211 889 \$ 544 883 \$ 6078	390 \$ 2493 392	\$ 363 363 \$ 1540 376 \$10483	\$ 397 320 \$ 718 334 \$13781	\$ 904 171 \$ 993 178 \$ 8311	\$ 389 126 \$ 1758 124 \$ 8878	\$ 924 381 \$ 735 386 \$ 3771	\$ 1133 218 \$ 866 220 \$ 2862	\$ 224 372 \$ 1454 380 \$ 8800	\$ 16 344 \$ 198 338 \$13641	\$ 44 265 \$ 149 273 \$12271	\$ 46 \$ 232 271 210 \$ 378 \$ 6060 277 215 \$12034 \$ 4299	460 \$ 1953 449 \$13356	\$ 7 660 \$ 439 644 \$ 9735	382 \$ 899 379 \$ 6335	\$ 116 47 \$ 141 46 \$ .450
Non-craft sources lumber of respondents	\$ 8008 1734	\$ 6078 673	295	277	222	134	98	314	181	- 300	· 246	182	183 162	357	446	280	37
refte income (gross) of other household members lumber of respondents	\$ 547 2252	\$ 274 862	\$ 490 389	\$ 1120 360	\$ 429 291	\$ 514 170	\$ 1075 124	\$ 1350 384	\$ 924 217	\$ 554 369	\$ 113 306	\$ 105 - 262	\$ 63 \$ 342 256 209	_	\$ 116 596	\$ 198 355	\$ 91 44
on-creft gross income of other household members lumber of respondents	\$14323 1739	\$18849 651	\$16017 315	\$11593 298	\$ 6197 208	\$ 9994 132	\$ 7931 96	\$14216 328	\$10703 1 <b>8</b> 2	\$15989 294	\$15659 -230	\$15976 195	\$13749 \$12830 185 169		\$15702 428	\$14041 281	
ium of all household income lumber of respondents	\$28886 1280	\$29437 478	\$301 57 236	\$30718 201	\$25959 162	\$27705 105	\$20273 78	\$27297 242		\$29475 224	\$30890 173	\$31695 136	\$29658 \$29696 131 118		\$26694 335		\$3077 31
verage (meen) gross income from crafts estee and related sources of respondent crafts—artists lumber of respondents	\$ 2736 2247	\$ 1222 862	<b>\$ 4298</b> 381	\$ 4717 349	\$25 <b>8</b> 1 30 <b>8</b>	\$ 6536 165	\$ 1607 120	\$ 6798 379		\$ 1969 - 355	\$ 192 333	\$ 652 258	268 202			<b>\$</b> 998 382	\$1069 47
lyerage (meen) crefts expenses of responds crefts-ertiets lumber of respondents	nt \$ 2627 2169	\$ 1171 793	\$ 2946 368			\$ 5469 164	\$ 1924 119	\$ 8464 362			\$ 613 319	\$ 1123 2 <b>5</b> 2				\$ 767 381	\$ 475 47
Averege (meen) net cre income of respondent crefte-ertiete Number of respondents		\$ 111 730	\$ 1753 343				\$ 798 106	\$ 1167 384		\$ 467 315	\$ -403 2 <b>90</b>	\$ -421 225			\$-170 660	\$ 292 382	\$ 626 47

Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Palicy Research, Inc.

L30

ERIC calculation of X statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crefts-entists by an amount of the can be accounted for by chance except these percentages or persentage distributions followed by an exterisk [\*].

TABLE IV.22,
TEACHING CHARACTERISTICS OF CRAFTS-ARTISTS
SURVEY OF CRAFTS-ARTISTS, 1980

		<del></del>		Primary (	Croft H	edie		<del></del>	Involve	ment				
								Major A	ctivity		Leisure Activity			-
	All Crefts- Artists	Fiber	Clay	Metel	<b>W</b> ood		Paper, Leather	Under 40 Hrs.	40 Hre. Plue	2nd Activity	10 Hre, or Lese	10 Hre. Plue	Occa- eionel Activity	Other
Percentage of crafts- artists teaching crafts in last 3		-		·				٠			A1 &			
yeere	42.8	44.3	53.8	35.8	27.1	46.7	45.7	80,4	57.5	56.7	27.1	34,3	23.4	53.1
lumber of Respondents	2630	986	434	422	365	208	140	438	256	480	371	306	296	326
Percentage teaching in:					_	•			***					
Crefte echool	4.1	4.4*	5.5*	3.6*	2.1	4.8*	2.7*	8,5	11.0	4.0	1.3	3.0	1.1	4.0
Professional workshops	8.0	10.4	8.5	8.2	4.5	10.5	4.8	15.1	15.0	13.3	2.5	4.3	1.2	9.3
Private Lessons	17.2	22.2	15.3	11.8	9.4	22.8	18.3	26.4	25.8	26.9	10.8	18.1	7.1	13.1
Public echoole	11.2	10.6*	14.0*	16.7*	12.4*	7.1*	8.7*	12.3	9.2	13.4	9.0	7.1	7.5	18.2
Adult education courses	10.2	11.5*	11.9*	7.2*	8.4*	10.1*	8.4*	11.9	10.5	19.1	5.8	4.8	5.9	13.4
Community centers	8.8	11.8	11,2	8.3	4.8	13.8	5.1	14.5	10.#	13.3	5.5	10.3	7.5	10.3
Crefte eĥope	6.1	10.0	4.8	0.5	3.9	4.8	1.0	12.2	5.3	10.8	2.5	4.5	1.9	8.2
Art echool	3.5	3.2*	6.1*	4.3*	4.3*	4.2*	4.3*	8.4	9.1	4.8	0.2	0.1	1.2	7.1
College/University	6.2	4.2	12.6	11.1	2.2	5.1	4.4	12.4	10.9	8.1	0.5	1.6	1.3	12.1
Other	3.8	4.1*	3.5	2.1*	2.8*		8.3*	4.0*	4.7*	4.7*	3.2*	3.2*	2.3*	3.2
Number of Respondents	2633	966	435	424	364	203	140	438	256	430	371	306	289	327
lverege (meen) houre per														
week apent teaching	2.6	2.1	3.1	3.1	2.8	2.8	2.5	2.3	3.3	2.4	2.2	1.7	2.5	3.6
lumber of respondents	1079	402	219	167	94	<b>91</b>	55	226	133	220	96	92	88	196
Percentage who teach:														
Beginning students	40.6	42.4	50.5	35.5	25.8	45.8	44.8	55.9	50.2	58.7	26.9	32.0	23.6	52.0
Intermediate atudente	26.3	27.0	38.8	24.5	15.1	26.1	19.0	37.2	45.3	38.5	10.8	18.7	9.2	38.7
Advanced students	15.3	13,5	23.9	17.2	9.9	12.7	14.3	21.7	29.3	19.8	4.5	9.1	4.6	28.1
lumber of respondents	2630	966	435	422	364	203	139	· 428	256	429	<b>3</b> 71	305	297	327

SOURCE: Survey of Crafts-Artists who are Nembers of Crafts Organizations, Mathematics Policy Research, Inc.

Based on calculation of X<sup>2</sup> statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exterisk (\*).

131



there is a net gain as a result of crafts activities of 345.1 Expenses are generally highest for the media with the highest level of gross sales with the exception of wood.

Expenses vary in amount and frequency. For example, 70 to 80 percent of all crafts-artists paid dues, purchased publications and bought materials during the 12 months prior to returning the questionnaire. The first two categories combined averaged \$123 for those making these purchases; materials cost an average of \$1,033. The most was spent on employee salaries, \$6,476, by the 4 percent of crafts-artists who had this type of expense (Table IV.23). Other large average amounts were for apprentice salaries (\$2,881 by 2 percent) and workspace and utilities (\$1,272 by 27 percent). About 54 percent bought tools or equipment during the prior 12 months spending an average of \$528.

The most "expensive" craft is glass or other media or combinations of media at \$5,725 followed by metal, with an average of \$5,034. Fiber and paper or leather were least costly, averaging \$1,194 and \$1,963 respectively. Materials are most expensive for metal workers, least expensive for fiber and paper or leather.

 $<sup>\</sup>frac{1}{}$ The lack of agreement betwen the subtracted difference of \$109 (\$2,736-\$2,627) and \$345 is because only those crafts-artists who reported both gross income and expenses were used to calculate net crafts income while those who did not report expenses are included in gross crafts income.

TABLE IV.23
INCIDENCE AND AVERAGE AMOUNTS OF CRAFTS EXPENSES SURVEY OF CRAFTS-ARTISTS, 1980

		Primary Craft Madia							
	All Crofto- Artioto	Fiber	Clay	Metal	Wood	Gleen, Other	Peper, Leether		
Percentage of crafts-	,					,			
ertiete with following			•						
types of crefts expenses:									
Oues	71.5	75.1	76.0	74.7	88.4	84.4	57.7		
Publications/books	70.1	72.6	78.8	69.6	87.2	63.7	50.5		
Materiale	79.2	<b>8</b> 0.4	90.9	77.7	77.8	78.3	84.5		
Apprentice esteries	2.0	1.2	3.5	3.7	1.8	3.7			
Employee eeleriee	3.9	3.4	2.9	5.3	3.0	11.1			
Equipment/tools	54.1	42.9	63,0	66.0	8.38	60.0	45.4		
Workepace, utilities,									
maintenance	28.5	13.8	48.0	37.2	30.7	32.8	19.8		
Trevel	41.6	35.3	50 <b>.</b> 6	48.2	40.7	57 .8	38.1		
Photography	31.9	25.2	46.5	30.7	25.5	40.0	24.7		
Ineurence	13.0	7.6	18.8	21.5	14.3	17.8	11.3		
Other	15.8	13.4	24.6	19.8	9.7	17.0	18.8		
Wember of respondents	2637	960	435	424	<b>36</b> 5	203	140		
Averege (meen emount									
of crefts expenditures:							_		
Ouee	<b>\$</b> 48	\$ 38	\$ 54	<b>\$</b> 60	\$ 55	<b>\$</b> 71	<b>\$</b> 35		
Publications/books	95	25	81	105	69	90	78		
Meteriele	1033	538	868	2890	- 1138	1971	451		
Apprentice seleries	2881	#	#	#	#	#	#		
Employee eeleriee	8476	#	#	#	#	#	•		
Equipment/toole	<b>528</b>	282	801	804	498	1103	•		
Workepace, utilities,						_	_		
maintenance	1272	257	1455	1241	1191	#			
Traval	570	400	577	967	504	#			
Photography	132	92	131	171	115	270	#		
Ineurence	356	182	457	381	#	#			
Other	957	519	568	•	•	*	*		
Estimated everage of				ý	9				
expenses for crafts-									
ertists incurring									
expenses	\$2828	\$1194	<b>\$294</b> 6	<b>\$5034</b>	\$3179	<b>\$5725</b>	<b>\$19</b> 63		

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematics Policy Research, Inc.

Besed on calculation of  $x^2$  statistics, in 99 out of 100 cases, the weighted percentages in this table very from the percentages for all crafts—artists by an amount larger than can be accounted for by chance except those percentages or percentage distributions followed by an exteriok  $\{*\}$ .

ige not shown because there are fewer than 50 respondents in category.

Beend on crefts-ortists who reported the particular type of expanditure.

#### H. AN OVERVIEW OF CRAFTS CHARACTERISTICS

Before undertaking several empirical investigations of the relationships between involvement in crafts and monetary returns and characteristics of crafts-artists, a review of the major attributes of the population of crafts-artists who are members of crafts organizations is useful. This is a summary profile of the average or typical crafts-artists to which the more specific findings of the regression models reported in Section V can be associated.

Fiber is the most popular primary media among crafts-artists, but as a secondary, leisure, or occasional activity rather than a fulltime or major activity. Clay and metal are the most prevalent primary media for crafts-artists who spend more than 40 hours per week producing crafts or, if less hours are spent, who consider crafts their primary activity. Glass is also prevalent as a primary activity. On the other hand, wood is most frequently mentioned as a leisure-occasional primary media. Paper, leather and other or combinations of media are much less popular as primary media.

Current trends in crafts design and work are reported as the most important influence on crafts-artists' work. American folk themes and ethnic/cultural traditions are also important sources of inspiration.

Most crafts-artists produce one-of-a-kind objects and those who engage in crafts as a primary activity accept commissions at an average of 22 per year.

The most popular form of crafts training is "self-taught"; the most frequently reported formal avenues are workshops at crafts schools and college or university courses in crafts. Least prevalent are apprenticeships. The relationship between amount of training on the one

hand and intensity of involvement on the other is not straightforward, with some media appearing to "require" more training than others. Fiber and wood are more self-taught than other media.

Satisfaction with past training is high and about 15 percent of all crafts-artists are presently receiving some form of training for an average of one hour every week. Two-thirds plan to acquire additional training in crafts. Lack of time and money are the most important barriers for more training.

Training in business skills has been acquired by about 30 percent of crafts-artists; but nearly 60 percent of those who have not received such training indicate a desire for it.

The "average" crafts-artist has been involved in crafts for 16 years and works almost 20 hours per week designing and producing crafts.

An additional 2 hours per week are spent in marketing. About 90 percent of crafts-artists visited a crafts gallery or museum in the 12 months prior to the survey and read an average of 2.4 crafts publications. Two-thirds of crafts-artists also collected crafts.

Crafts-artists tend to work alone and in their home. About a third report that other family members work in crafts, but often in a different media. One-third also replied that their parents were involved in crafts.

Creative or intangible aspects of crafts as a "means of creative expression," "sense of accomplishment," and "life's work" outrank more pecuniary aspects as important satisfactions. However, the barriers to achieving these satisfactions are non-craft obligations, money, time, lack of public appreciation, marketing opportunities and training. The most frequent goals set by crafts-artists are to devote more time to



crafts and to develop artistic competence; increasing income and winning an award or recognition are less important.

Over 82 percent of all crafts-artists exhibited their crafts within the 12 months prior to the survey, with nearly half doing so from two to five times. About 42 percent of repondents who exhibited received an award or prize within 3 years of the survey. Art/crafts fairs and galleries were each listed by 35 percent as the best places to exhibit. Crafts shops (either their own or someone else's) were next most popular. The lack of exhibits and poorly informed communities were listed as the major problems by crafts-artists who exhibit.

About two-thirds of all crafts-artists produced goods for sale; of these, 23 percent report receiving their primary income from crafts sales. Art/crafts fairs and galleries are again listed as the best places to sell. Too few outlets, dealing with clients, lack of business skills and a poorly informed public are listed as the major problems.

Relatively small proportions, each less than 10 percent, have received a grant or fellowship, applied for a crafts-related bank loan, purchased equipment or materials in cooperation with other crafts-ertists or exchanged crafts for other goods and services.

Crafts provides a relatively small contribution to net household income and is about as likely to be a drain on household income as a contribution. The estimated average crafts-related income of crafts-artists is \$2,/93. Nearly 42 percent of crafts-artists teach crafts and earn an average of \$1,109 from teaching. Crafts expenses are estimated on average at \$2,627, leaving a net gain of \$345 (based on data for those who reported both income and expense categories).

Average non-crafts personal income is about \$8,000 per year while other family members earn about \$14,900 (\$14,300 of it from non-crafts sources). Overall, average annual income for crafts-artists' households is about \$28,900.

Materials are the most frequent crafts-related expense, amounting to about \$1,000 per year for those who buy them. Dues and publications are the next most prevalent expense at about 70 percent each and costing an average of \$48 and \$85 per year repectively. The most expensive item--employee salaries at \$6,500--is among the most infrequent cost at 4 percent.

The next section presents and interprets the empirical estimates of a causal model that attempts to explain the factors leading to a greater number of hours per week spent in crafts production and marketing and also to higher gross hourly earnings from the sale of crafts products.

## V. EMPIRICAL ANALYSIS OF CRAFTS-ARTISTS ACTIVITY AND FINANCIAL REWARDS

#### A. INTRODUCTION

رهين

The descriptive tabulations presented in the previous chapters provide information that has not been previously available on a wide range of characteristics of crafts-artists who are members of crafts organizations. The cross-tabulations that show the joint distribution of two characteristics, such as media by years of involvement or level of involvement by net crafts income, in many instances reveal patterns suggestive of underlying causal relationships that warrant further investigation. In this section, the focus shifts from a broad-based descriptive analysis to an indepth analysis of two specific aspects of the activity of crafts-artists: (1) their level of involvement in terms of weekly hours spent producing and selling crafts, and (2) the financial returns to crafts activity measured by gross hourly crafts earnings. A multivariate model of crafts-artists' behavior is specified and estimated with the survey data using multiple regression techniques. The results of the analysis indicate the extent to which various characteristics of crafts-artists affect the two outcome measures-hours of involvement and financial returns to crafts activity-noted above.

Crafts-artists are an interesting, but difficult, group to study because the pursuit of crafts is influenced by noneconomic as well as economic factors. Furthermore, crafts activity may or may not be intended to generate sales income, which implies that the distinction between leisure-related and work-related activites is considerably blurred. Despite these inherent difficulties, the empirical findings



V.1

reveal a number of interesting insights. By and large, crafts-artists display much of the same economic behavior that has been found in other studies of workers in the private sector. In general, they benefit financially from the acquisition of skills and spend more time producing crafts in response to greater potential remuneration. Since crafts-artists have much in common with self-employed business persons, it is not surprising that product quality and managerial skills are relevant determinants of economic success.

As noted earlier in the report, the survey on which the analysis is based was designed to represent the U.S. population of crafts organization members. Hence, caution should be exercised in generalizing these results to nonmember crafts-artists. In addition, this analysis does not focus on the decision to engage in crafts because only active organization members are sampled. However, some indirect insight into this issue is possible by comparing various characteristics of the crafts-artists sample with corresponding characteristics of the overall population using a national data file such as the Current Population Survey (see Section III).

In the remainder of this section, a conceptual framework of crafts activity and pecuniary returns is developed, then the empirical results are presented and interpreted, and, finally the section concludes with a summary of the findings, recommendations for future research, and key policy implications of the study.

#### B. CONCEPTUAL OVERVIEW

The analysis focuses on the level of crafts involvement and the income earned by crafts-artists. Operationally, these concepts are



defined as (1) the hours per week spent in crafts production and marketing, and (2) the gross hourly earnings generated by the sale of crafts products. The rationale behind the definition of these dependent variables is described below. The general model for each analysis is then summarized, based on a conceptual framework that is elaborated in a technical discussion contained in Appendix D.

### 1. <u>Dependent Variables</u>

The notion of an individual's involvement in crafts does not lend itself to a simple unidimensional measure. Rather, it suggests several facets of behavior that include hours devoted to producing and marketing crafts, capital expenditures, involvement with organizations, efforts to improve skills and interaction with the public and other crafts-artists through shows, exhibits and teaching. For the purpose of this research, it was decided to concentrate on a single, though probably the most important, aspect of involvement, the time spent in producing and selling crafts. Other aspects of crafts involvement are more difficult to interpret (e.g., number and type of exhibits, teaching, capital expenditures), because they are influenced by such factors as an individual's skill and artistic ability, financial resources and local opportunities. An hours measure, on the other hand, is a more direct barometer of crafts activity. 1/2

Hours of craft activity in the analysis is based on a respondent's answer to how many hours are spent producing and marketing craft products. These responses are interpreted as hours spent in a typical or average week.

The financial rewards associated with crafts activity can be measured in terms of hourly or annual earnings. The former is more attractive from an analytical point of view because it conveys the pecuniary return to a small unit of labor. The latter, on the other hand, reflects not only the return per unit of labor input, but also the total number of yearly hours spent in crafts. The primary interest in this analysis is in the financial return to crafts activity that is standardized across different levels of involvement. The hourly earnings variable is constructed by dividing an individual's gross annual crafts income by the total number of hours spent in producing and marketing crafts. $\frac{1}{4}$  A preferred hourly earnings measure would deduct for workrelated expenses and income taxes. Unfortunately, this is not easily done. First, as with virtually all surveys, the members' data set was not designed to collect tax information, nor were there sufficient project resources to impute tax liabilities. Second, while the survey did gather information on production costs, this item had a large number of missing questionnaire responses. Eliminating survey observations with incomplete cost information for the multivariate analysis would have seriously reduced the size of the analysis sample. Also, the questionnaire was not designed to distinguish between capital and noncapital expenses (e.g., the nonrecurrent cost of a kiln versus the recurrent expenses of clay), which makes it difficult to create a net earnings measure.

ERIC®

142

 $<sup>\</sup>frac{1}{}$  The survey did not collect information on weeks worked in the last year. To create an annual hours measure consistent with the annual earnings, the hours per week variable was arbitrarily multiplied by 50 weeks.

#### 2. Hours of Crafts Activity

A useful context within which to conduct the empirical analysis is the labor-leisure model of consumer behavior developed in the economic literature. This framework, which is described in Appendix D, provides an insightful and consistent guide in specifying what factors influence an individual's decision to participate in crafts activity. Modeling this type of behavior, however, is more complicated than is the case for most workers in the labor market. This is because nonpecuniary elements of crafts activity, on average, represent a much more important motivation for involvement than is true for most workers. Moreover, crafts-artists have considerable discretion in adjusting both the level of their involvement and the mix of pecuniary and nonpecuniary remuneration that is derived from crafts activity. Similarly, the distinction between work and leisure is less clear for many crafts persons.

These complications are taken into account in the model of individual behavior presented in Appendix D. Both financial and nonfinancial rewards to crafts activity are explicitly addressed by the model, in that the incentive to engage in crafts is shown to be a function of total or potential remuneration received from crafts including wage and nonwage elements. The model is also useful for identifying the inherent difficulties in investigating the pecuniary returns to crafts activity because of: (1) the unobserved nature of its nonpecuniary returns and (2) the unobserved degree of individual choice between financial and nonfinancial returns.

Based on the conceptual model presented in Appendix D, the supply of work hours (H) spent on crafts activities can be specified as a general function:

v.5

ERIC"

143

 $H = f(W*, P, Y_n; X)$ 

where W\* is hourly remuneration, consisting of monetary and non-monetary benefits, P is the rate at which income is foregone as crafts activity shifts away from sales toward an avocational focus, Y is other family income, and X is a vector of personal and environmental characteristics. Work effort (H) is hypothesized to be a positive function of the incentive variable (W\*), i.e., more hours of work are expected the greater the financial returns, and a negative function of P, the rate at which nonpecuniary job benefits are sacrificed to increase income. The latter relationship, however, is not observable in the survey data base and is not included in the analysis. The link between other household income (Y and H is ambiguous. On the one hand, a high Y suggests less of a need to generate sales income from crafts work; while, on the other hand, a high Y may imply greater crafts involvement if the latter is viewed as a leisure activity.

The empirical model that is estimated in the following section is able to test for the effects of W\* and Yn, together with a number of personal and environmental variables represented in X. These variables are grouped into the following categories: general and specific training (education, training in crafts, training in business skills), artistic ability (kinds of exhibitions, receipt of awards or financial support), production factors (type of workspace, presence of other crafts-artists in the household), location (urban versus rural and region of residence), and demographic characteristics such as age, sex, and marital status.

#### 3. Financial Returns to Crafts Activity

The empirical analysis is also concerned with the determinents of a crafts-artists's hourly earnings or wage rate. In general, the analysis assumes that an individual's potential level of renumeration is a function of crafts skills, production facilities (PF), opportunities to market craft products (MKT), marketing skills (MS), and unobserved preferences for monetary versus nonmonetary returns. Skills are presumably acquired through experience (EXP) and training (T). In addition, a person's artistic ability, (A), is germane and must be taken into account. This suggests a general functional form determining a craft-artist's wage rate (W):

$$W = g(EXP, T, A, PF, MKT, MS, P, Z)$$
 (2)

when Z is a vector of personal characteristics, including leisure-income preferences, and P is the price of nonmonetary job benefits. Equation (2) implies that, for a given set of personal and taste characteristics, a crafts-artists's hourly earnings will be higher, the greater is the level of skills and artistics ability (EXP, T, A); the better are the opportunities to exhibit and sell crafts (MKT); and the more adept is the individual in marketing crafts products (MS). The empirical analysis also controlled for the crafts medium in which an individual is primarily engaged. This was done to ascertain whether or not some media are associated with higher returns than others while holding the effects of our factors constant.

The wage rate analysis is useful because it potentially permits estimating the average financial returns to investments in crafts

training, crafts experience and marketing ability. The specification also permits estimation of the financial implications of artistic ability (within the confines of the data). However, this analysis is limited to an unknown degree by the crafts-artist's perference for wage versus nonwage remuneration. If an individual chooses not to sell his or her crafts (an avocational orientation), then, regardless of the accumulated training and experience, artistic ability and so forth, the corresponding wage rate observed for that person will be very small, if not zero. In fact, in the analysis sample, almost one third of the crafts-artists do not report any crafts-related earnings (W = 0). As implied by the above, the individual's choice between wage and nonwage renumeration must be explicitly recognized and resolved if one is to obtain reliable estimates of the determinants of craft-artists' hourly earnings.

#### C. EMPIRICAL FINDINGS

This section discusses the empirical results for the relationships developed in Section B. The analysis focuses first on the correlates of weekly hours of crafts activity, and then turns to the determinants of hourly earnings of crafts-artists.

#### 1. Weekly Hours of Crafts Activity

The dependent variable used in the multivariate regression analysis is the sum of weekly hours spent in producing and marketing craft products. The second component is a relevant constituent in that



 $<sup>\</sup>frac{1}{0}$  Of course, avocational crafts activity need not imply W=0 because many hobbyists sell their products at crafts fairs on an occasional basis.

crafts sales require some marketing effort. The analysis sample for the hours analysis consists of 1363 observations reporting a positive number of hours spent in crafts activity.  $\frac{1}{2}$ 

Table V.1 presents unweighted descriptive statistics for the analysis sample. The mean number of weekly crafts hours is 23.

Tabulations not shown indicate that 38.6 percent of the sample work less than 10 hours per week at their craft; 57 percent work less than 20 hours per week; 70 percent less than 30 hours per week; and 80 percent less than 40 hours per week.

In general, the sample is predominantly female (68 percent), white (97 percent), married (74 percent), and well educated (58 percent with at least a BA, 97 percent with at least a high school education). The sample members are evenly dispersed by degree of urbanicity and by the geographic location of their respective crafts organization; 81 percent are homeowners.

As noted earlier in this report, the surveyed crafts-artists have accumulated substantial training in crafts through a variety of methods. In the analysis sample, 16 percent have earned an MFA; 26 percent have received an award or prize in the last three years; and 7 percent have been the recipient of some form of financial support. Tabulations not shown indicate that 51 percent exhibited their work at least twice during the past twelve months; and about one-third are currently teaching



 $<sup>\</sup>frac{1}{T}$  The analysis sample is smaller than the number of survey respondents because cases with missing data on variables used in the estimation were excluded.

TABLE V.1

UNWEIGHTED DESCRIPTIVE STATISTICS FOR THE WORK HOURS ANALYSIS (Sample proportions or means with standard deviations in parentheses)

*	4
Hours/Week Producing & Selling Crafts 23.12 Imputed Wage Rate Private Workshop .07 Other Household Member in Crafts - Same Medium .13	(19.47) ( .55)
Other Household Member in Crafts - Different Medium .19 Less than High School Education .03 High School + Vocational Degree .05 Some College .24 B.A. Degree .25 Graduate School .33 Crafts Lessons from Friend/Relative (years) .158 School Art Class (years) .120 Community Center Training (years) .62 College/University (courses) .18.23 Apprenticeship (months) .1.43 Adult Education (courses) .66 Crafts School Workshops (number) .41 Professional Lessons (years) .41 Professional Workshops (number) .285 Earned an M.F.A.	( 5.46) ( 3.88) ( 4.78) (34.67) ( 7.32) ( 2.02) (15.30) ( 1.43) ( 13.79)
Currently Receiving Training Received Training in Business Skills Parents Produced Crafts for Sale Parents Produced Crafts for Leisure Teach Crafts - College Teach Crafts - Workshops Teach Crafts - Public Schools Teach Crafts - Private Lessons Exhibited in Museum Exhibited in Gallery Exhibited in Other Context Exhibited at Community Event Exhibited at Crafts Festival Received Crafts Award or Prize Received Financial Support Exchange Crafts with Other Artists Reside in Urban Area Reside in Suburban Area Male  114 Received Crafts for Necessities Received Financial Support Reside in Suburban Area Reside in Suburban Area Reside in Suburban Area Reside in Suburban Area	
Age Household Size Never Married Separated/Divorced/Widowed Homeowner  45.49 2.67 13 13 84 14 15 16 17 18 18 18 18 18 18 18 18 18 18 18 18 18	(14.81) (1.33)
Pelong to Small Crafts Organization .71 Belong to Large Crafts Organization .21 Organization in Northeast .25 Organization in North Central .08 Organization in South .17 Oganization in West .09 Frimary Medium is Clay .16 Primary Medium is Leather .01 Primary Medium is Paper .05 Primary Medium is Glass .05 Primary Medium is Metal .14 Primary Medium is Wood .17 Primary Medium is "Other" .02	
SAMPLE SIZE 1,363	

crafts. Approximately two-thirds of the analysis sample sold crafts products during the past twelve months, and this subset has an average wage rate of \$2.64 per hour.

As suggested by the foregoing, there is considerable variation in the level of crafts involvement in the sample. In particular, the weekly hours in which individuals are engaged in crafts production and marketing varies widely. There are several factors which may influence an individual's hours decision, thereby explaining the variation observed in the sample. The explanatory factors can be grouped into six general categories: (potential) monetary incentives to produce crafts, other household income, production costs, general and specific training, artistic ability, location, and demographic characteristics. Also included in the estimating equation are the primary media with respect to each crafts-artist and the type of organization to which he or she belongs.

Monetary Incentives. A theoretically important determinant of crafts hours is the potential returns to participation. For someone interested in producing crafts for sale, such a concept is measured by the expected dollar return to an hour in crafts production and marketing (i.e., an expected wage rate). However, for someone who participates in crafts as an avocational or leisure activity, such a wage rate is difficult to measure (there may be zero observed sales income) and tricky to interpret. A useful approach to solving this problem is to "impute" a wage rate to all crafts-artists, regardless of whether or not they sell their products. The imputed hourly earnings variable is then interpreted as the potential gain to an hour of crafts activity, irrespective of the

decision to sell. A negative decision would suggest that other, nonpecuniary job satisfactions are realized from crafts participation, which have a monetized value equal to the imputed wage.

The estimation is thus based on a two-step procedure. First an hourly earnings regression is estimated for the subsample of crafts-artists who sold crafts. Second, the estimated coefficients obtained from the first step are then used to predict a wage rate for all sample members. The latter presumes that sellers and nonsellers differ behaviorally only in their decision to sell or not. That is, a typical nonseller is assumed to have been able to earn the same pecuniary return for an hour of work as an otherwise similar crafts-artist who did report selling his or her product. The results from this estimation are discussed in the following subsection.

Table V.2 presents the empirical estimates for two regression equations—with and without the imputed wage rate. The discussion focuses primarily on the second equation, although it is important to examine the stability of the estimated coefficients when including an imputed variable that is a function of several of the other explanatory variables. In general, the results are similar for both equations. It is quite apparent that the potential rourly return to crafts activity is a significant determinant of weekly hours. For a one dollar increase in the potential wage rate, it is estimated that a crafts—artist will engage in almost 14 additional weekly hours of production and/or marketing activity. It should be noted, however, that a one dollar increase in the

TABLE V.2

REGRESSION COEFFICIENTS ON WEEKLY HOURS OF CRAFTS ACTIVITY (Absolute value of t-ratios in parentheses)

į.	Equation (1)		ion (2)
mputed Wage Rate		13.87	(8.36)***
rivate Workshop (320) <sup>a</sup>	7.53 (6.46)***	2.45	(1.91)*
ublic Workshop (95)	15 ( .08)	.71	( .41)
ther Household Member in			
Crafts - Same Medium (183)	3.23 (2.39)**	2.23	(1.69)*
ther Household Member in			
Crafts - Different Medium (261)	50 ( .43)	20	( .17)
ess than High School Education (36)	- 1.07 ( .35)	- 1.80	(.61)
igh School + Vocational Degree (71)	.23 ( .10)	.18	( .08)
Some College (325)	.68 ( .40)	.33	( .20)
3.A. Degree (336)	- 2.11 (1.22)	- 2.13	
raduate School (449)	- 3.67 (1.93)		(2.19)**
rafts Lessons from Friend/	3.07 (2.7,07		(
Relatives (years)	04 (1.00)	07	( .80)
chool Art Class (years)			(.71)
Community Center Training (years)	11 (1.24)		(1.34)
college/University (courses)	.02 (1.24)	.01	( .96)
pprenticeship (months)	.23 (3.69)***		(4.06)***
dult Education (courses)	61 (2.73)***		(2.48)
rafts School Workshops (number)	04 (1.24)		(1.36)
rivate Professional Lessons (years)	- ,32 (1.00)		( .64)
rofessional Workshops (number)	.03 ( .74)	.03	( .77)
arned an M.F.A. (217))	.67 ( .48)		( .58)
Surrently Receiving Training (191)	- 1.18 ( .84)	1.87	(1.32)
Received Training in Business	1110 ( 104)	2.07	(210-)
Skills (282)	3.77 (3.36)***	2.18	(1.97)***
Parents Produced Crafts for Sale (62)	_	2.36	(1.12)
arents Produced Crafts for Leisure	1.01 ( 1.73)	2.50	(2022)
(369)	- 1.51 (1.43)	- 1.41	(1.38)
Ceach Crafts - College (32)	4.16 (1.33)		(2.33)**
each Crafts - Workshops (25)	71 ( .21)		(2.25)**
each Crafts - Public School (144)	- 1.32 ( .86)		( .55)
Ceach Crafts - Private Lessons (239)	3.54 (2.90)***		
Exhibited in Museum (203)	14.73 (8.98)***		(7.57)***
Exhibited in Gallery (183)	10.26 (6.48)***		(4.99)***
Exhibited in Other Context (320)	9.60 (7.29)***	8.07	(6.62)***
Exhibited at Community Event (65)	5.11 (2.35)**	3.82	(1.80)*
Exhibited at Crafts Festival (116)	6.52 (3.81)***	5.41	(3.23)***
Received Crafts Award/Prize (358)	1.94 (1.66)*	.08	( .07)
deceived Grant's Award/Frize (350)	57 ( .31)	- 1.95	(1.10)

continued

TABLE V.2 (continued)

# REGRESSION COEFFICIENTS ON WEEKLY HOURS OF CRAFTS ACTIVITY (Absolute value of t-ratios in parentheses)

Independent Variables	Equation (1)	Equat	ica (2)
Exchange Crafts for Necessities (37)	2.79 ( .97)	- 8.39	(2.69)***
Exchange Crafts with Other Artists			
(392)	5.30 (4.79)***	3.09	
Reside in Urban Area (386)	- 3.54 (3.25)***		(3.00)***
Reside in Suburban Area (502)	- 3.18 (2.90)***		(2.81)***
Male (440)	.07 ( .06)		(2.21)**
Age	08 (2.01)**	.08	(2.08)**
dousehold Size	52 (1.38)	69	(1.91)*
Never Married (174)	1.33 ( .88)	1.33	
Separated/Divorced/Widowed (184)	1.37 ( .97)	.80	(.58)
Homeowner (1105)	- 3.94 (2.83)***	- 3.74	(2.76)***
Belong to Small Crafts Organiza-			
tion (965)	3.68 (2.22)**	3.32	(7.06)**
Belong to Large Crafts Organiza-			
tion (282)	2.54 (1.35)	2.63	(1.43)
Organization in Northeast (342)	53 ( .38)	99	( .72)
Organization in North Central (111)	- 1.52 ( .89)	- 4.08	(2.17)**
Organization in South (241)	- 2.94 (2.03)**	- 2.86	(2.02)**
Organization in West (128)	35 ( .19)	.46	( .26)
Primary Medium is Clay (223)	3.42 (2.38)**	.97	(.69)
Primary Medium is Leather (15)	.84 ( .20)	- 2.72	(.65)
Primary Medium is Paper (64)	- 3.86 (1.76)*	- 3.82	(1.80)*
Primary Medium is Glass (66)	3.54 (1.58)	- 1.71	(.76)
Primary Medium is Metal (197)	28 ( .18)	- 5.01	(3.11)***
Primary Medium is Wood (237)	1.65 (1.03)	.40	( .25)
Primary Medium is "Other" (33)	5.27 (1.79)*	1.56	( .54)
Constant Term	10.94	19.21	
R <sup>2</sup>	.36	.39	
R Ratio for the Equation	12.68	14.53	
SAMPLE SIZE	1,363	1,363	

<sup>\*</sup>Significant at the 10% level of confidence, two-tailed test.

\*\*Significant at the 5% level of confidence, two-tailed test.

\*\*\*Significant at the 1% level of confidence, two-tailed test.

 $<sup>^{\</sup>rm a}$  For dummay variables, the number of observations are noted in parentheses.

imputed wage is a relatively large amount since its mean value for the sample is \$.61 per hour.  $\frac{1}{}$ 

Other Household Income. Some unreported regression equations were tested for the influence of an income effect on crafts involvement. The explanatory variable used was income earned by other members of the household (excluding the crafts-artist in question). The estimated coefficient for other household income, however, was statistically insignificant. Therefore, it was dropped from the estimating equation. Another reason for excluding it was that 30 percent of the sample observations had missing household income data. The insignificant coefficient for other household income may reflect offsetting effects due to: (1) a positive correlation between other income and crafts as a leisure activity, and (2) a negative association for individuals who treat crafts as an income producing activity. Future analysis should attempt to interact other household income with an imputed measure of sales orientation in order to disentangle these conflicting effects. An individual's noncraft labor earnings were not included in the estimation.

<sup>1/</sup>As a point of comparison, equation (2) was re-estimated using the observed gross wage rate (including zero values for nonsellers). The mean wage was \$1.74; its estimated coefficient was .27 (t=1.98); R = .36. Clearly, the conceptually more appropriate variable performed better statistically as well as providing a more satisfactory interpretation of behavior. Note that the average imputed wage rate is smaller than the average observed wage rate. This is because of the method used to determine imputed wage rates, which eliminates random and unexplained variation. In technical terms, the imputed value is calculated by taking the antilog of the predicted log-wage, based on the regression results discussed below. The value of the antilog is determined by the mean and variance of the log-wage. Since the predicted log-wage had a smaller sample variance than the observed log-wage because random and unexplained variation are omitted in the former, the resultant wage rate obtained from the antilog is smaller for the imputed wage.

As noted earlier, there are theoretical reasons to berieve that crafts and noncrafts employment decisions are made jointly. If noncraft earnings were included in the regression equation without explicitly modeling the underlying decision-making process, the estimated results would suffer from simultaneity bias.

Production Costs. The level of crafts activity may also be sensitive to the cost incurred. Unfortunately, the survey data do not contain cost measures suitable for the analysis. If resources permitted, an attractive solution would be to collect secondary data on production costs (recurrent and nonrecurrent) that are standardized by scale of operation and by media. While the inclusion of a crafts-artists primary medium may control for this cost variation, it may also be capturing the effects of other unobservable factors.

One indirect measure of crafts production costs may be the extent to which an individual shares the fixed costs of producing crafts with another. Such an example is the case of an individual who resides in a household with another crafts-artist. In this situation, some of the fixed, noncurrent costs of producing costs could be shared, thereby reducing the costs of activity. Since the production techniques vary substantially by crafts media, one would expect this effect to be most pronounced in households with two or more crafts-artists who work in the same as opposed to different media. The data confirm this expectation in that crafts-artists in the first instance are estimated to spend two additional hours per week in crafts activity over that for otherwise similar persons.

A second indirect measure of crafts production costs is related to the type of workspace used in producing crafts. This measure,

V.16

ERIC

154

however, is very imprecise for it consists of two dummy variables indicating whether or not a crafts-artist uses a private studio or workshop, or a public facility (school, community, cooperative), versus the excluded alternative, an in-home workshop. On the one hand, one might expect that rental costs would influence the work hours decision, although the effect should vary considerably by the nature of the activity. On the other hand, the measure does not reflect the level of production efficiency associated with different workspaces that may offset the negative cost effects. The regression estimates suggest that crafts-artists using a private work space, on average, spend over two more hours than those using public or in-home facilities. This supports the work-efficiency contention. It may also reflect the fact that persons producing at a relatively large scale tend to use private workspaces that may offer greater flexibility and cost advantages. However, given the general nature of the measure, it is difficult to draw clear inferences.

Education and Training. The level and type of training received does not exhibit any systematic pattern of influence. Years of formal education, with one exception, is not significantly correlated with the level of craits involvement. That persons with advanced degrees significantly engage in about four fewer hours than other crafts-artists may be indicative of the presence of other activities that compete with crafts for their time.

Specific training in crafts is generally uncorrelated with crafts activity except for individuals who have received apprenticeship training, who are estimated to participate in almost three additional hours per week for each year of apprenticeship. An explanation for why

V.17

ERIC

155

this type of training stands out among the other may be that it connotes a greater commitment to the field. The negative relationship between crafts hours and adult education courses may to picking up the reverse effect—those who treat crafts more as a part-time activity than a job (other types of training that tend to be more occasional in nature also have negative signs, although their effects are statistically insignificant). The receipt of an MFA does not appear to be statistically associated with hours devoted to crafts.

Teaching crafts may serve as a better indicator of skill attainment than past enrollment in crafts training courses. Presumably, a teacher possesses sufficient skills to be able to instruct others in the field. Whether or not a crafts-artist is a teacher may also depend upon teaching opportunities and a willingness to do so, which makes it difficult to classify persons in the sample by skill levels according to their teaching experience (measured over the past three years in the data set). The regression estimates reveal that crafts-artists teaching at the college level and those giving private lessons are significantly more active in producing crafts than nonteachers. On the other hand, those teaching at crafts schools or professional workshops spend significantly fewer hours in producing crafts than nonteachers. The latter finding is puzzling and it may be idiosyncratic, especially in light of the small number observed in that category. 1

Note that because survey respondents were encouraged to report all teaching experiences, it was necessary to select only one response in order to construct mutually exclusive dummy variable categories. The method used for selecting a teaching measure was to rank all reported teaching posts for each respondent and choose that which was believed to require the most skill. The ranking of teaching posts is in the same order as that presented in Table V.2

Prior training in business skills (e.g., marketing, accounting, contracting) is estimated to have a positive effect on work hours. In one sense, this training opens a new area of crafts involvement, thereby leading to a greater likelihood of engaging in marketing activities. But, this relationship may be more indicative of the tendency for serious crafts-artists to acquire the necessary business skills to promote the sale of their product. Hence, the estimated effect may be one of association rather than causal.

The lack of a systematic tie between crafts involvement and prior training is not surprising in light of the nature of this form of activity. Many crafts media do not require substantial training, and, furthermore, activity levels are not subject to institutional constraints that might vary according to technical ability. From an economic perspective, the a priori expectation is that crafts training should affect work hours indirectly, principally through the influence on the potential returns to crafts production.

Artistic Quality. The regression estimates suggest that artistic quality may be related to the level of crafts involvement. Those who have exhibited their work sometime during the last year spend significantly more time producing crafts than those who have not. This relationship is most pronounced for those who have exhibited in a museum, and slightly less for other types of exhibits (e.g., galleries). These exhibits, one-person shows, and invited or group competitions, imply a relatively high standard of artistic excellence. Those who have participated in juried community events and art and crafts festivals are also more likely to spend greater amounts of time in producing than those who have no such experience.

ERIC Full Text Provided by ERIC

v.19 · 157

Also included in the estimating equation are other possible measures of artistic quality: the receipt of an award or prize in the past three years, and ever having received a fellowship, grant or other support from a government agency or private foundation. Neither of these variables, however, display any systematic relationship to work hours that is statistically reliable. Another potential indicator of artistic quality is whether or not a crafts-artist exchanges his or her work for that of other crafts-artists. The occurrence of such exchanges could be interpreted as a rough measure of peer acceptance or judgement. This variable is estimated to be significantly and positively related to work hours. However, one should not read too much into this association because exchange among comparable crafts-artists may occur at several levels of artistic quality. It may also occur more frequently in areas with relatively large groups of crafts-artists, implying some type of group dynamics effect.

The fact that artistic quality varies substantially among craftsartists is a unique factor of this field. Since crafts products are not
homogeneous items, quality as well as specific stylistic and design
factors should be important elements in distinguishing the work of one
artist from another. And, if the buying public recognizes and is willing
to pay for higher quality, one should expect a positive relationship
between quality and sales income. Part of this phenomenon should
manifest itself indirectly through a higher imputed wage rate, thereby
eliciting a greater work effort.

Artistic quality may also have a direct effect on work effort to the extent that higher quality may imply greater quantities sold rather than higher unit prices (obviously, both may happen). Thus, craftsartists of unequal abilities may have the same observed (average) hourly wage rate, but the better endowed individual is able to work more hours because of the greater demand for his/her product. This is particularly germane for the crafts field because those who sell are similar to selfemployed workers who are not paid for hours worked, but by how much they sell. Hence, crafts-artists are in a better position to adjust both their work hours as well as their unit prices in response to demand conditions. Finally, it should be stressed that the measures of craftsartist quality used in this analysis may not adequately reflect true artistic quality. Even under the best circumstances, quality is difficult to gauge and is often subject to personal tastes. To some extent, a measure of recent exhibits may reflect popularity rather than intrinsic quality. Furthermore, the local availability of exhibits and shows will influence a crafts-artist's ability to show his or her work. Even so, one's reputation and the marketing advantages derived from exhibitions and shows are estimated to have a positive influence on the level of crafts involvement for the above reasons.

Demographic Characteristics. The variation in weekly hours of crafts activity is also correlated with several demographic characteristics. Controlling for other factors, males work almost three fewer hours per week than females. It is interesting to note that when the imputed wage is excluded from the regression (equation 1), there is no significant difference between males and females. In equation 1, the

ERIC Full Text Provided by ERIC

 $<sup>\</sup>frac{1}{A}$  Although a lower quality artist might accumulate a greater inventory than a higher quality artist, there are limits to the extendibute that unsold products can be accumulated.

male dummy variable is capturing the positive hours effect due to higher wage rates for males, which is offset by the negative hours effect independent of the financial returns to crafts.

Crafts activity is estimated to be significantly related to age; older persons spend more hours in crafts than younger persons. An explanation for this may be that crafts became an increasingly important leisure-time activity as an individual matures. The negative relationship between work hours and household size is suggestive of the demands placed on household adults' time by other members, thus leaving fewer hours available to pursue crafts. Marital status, however, does-notexhibit a statistically significant association with hours. Single and married individuals tend to engage in the same level of activity, other things being equal.

Location. Crafts-artists residing in urban and suburban locations are found to work about three hours per week less than otherwise similar workers living in rural areas. One speculation regarding this difference is that, on average, crafts production may be a more important source of livelihood and recreation for residents in more sparsely populated regions. That is, there may be fewer employment or leisure opportunities in rural areas.

Craft Organization Characteristics. The estimating equation also included two characteristics of craft organizations to which members belonged. There are no a priori expectations for doing so other than investigating any inforseen empirical links associated with organizations. The empirical results indicate that crafts-artists of small organizations, less than 2,500 members, worked three more hours per week than similar artists belonging to medium and large organizations (the

latter was the excluded group in the regression equation). Craftsartists belonging to organizations in the South and North Central regions
of the country were found to work significantly fewer hours than members
of national organizations (the excluded group) or those of organizations
in other regions.

Media. Although the results suggest that crafts-artists who report that paper and metal are their primary media tend to work fewer hours, these estimates do not suggest a straightforward explanation. The major purposes for including the primary medium were to control for personal differences associated with crafts media and identify significant patterns of crafts involvement by medium.

In summary, the level of crafts involvement, as measured by weekly hours of production and marketing, appears to be influenced by several factors. The two most powerful predictors of work hours are (1) the imputed wage rate, and (2) where and if an individual exhibited his or her products. Prior education or crafts training, other than apprenticeship, generally were not influential determinants of work hours. On the other hand, teaching at a college and giving private lessons were significantly related to work hours and demonstrate a positive effect of acknowleged crafts expertise. In general, males, younger persons, those owning a home, and those with larger families tend to spend less time in crafts activities, most likely because of noncraft commitments. The receipt of financial or honorary rewards does not seem to bear any systematic relationship to crafts activity levels. Finally, there does not appear to be a pronounced difference in work hours by the primary medium of crafts-artists that lends itself to any meaningful interpretation.

ERIC Foundation by ERIC

16i

#### 2. Hourly Returns to Crafts Activity

Two-thirds of the crafts-artists in the sample report that they sell their products. This raises the interesting question of what are the financial returns to crafts activity? To answer this question, the annual sales income was divided by annual hours spent producing and marketing crafts products. The hourly rate that is obtained is thus standardized across differing levels of involvement. The average hourly return for the analysis sample of sellers is \$2.64; expenses and taxes have not been deducted. Among the subset who have positive gross wage rates, 27 percent earn less than \$.50 per hour. 43 percent earn less than \$1.00 per hour, 68 percent earn less than \$2.50 per hour, and 12 percent earn over \$5.00 per hour.

The skewed distribution of wage rates indicates that this sample of crafts-artists derives fairly low hourly return from crafts activity. As discussed earlier, however, the hourly return to producing and selling crafts understates the potential return from this activity because: (1) crafts artists do not necessarily attempt to sell all of their work, and (2) the satisfaction and artistic fulfillment associated with crafts represents a large, if not dominant, share of the total returns from crafts activity (pecuniary plus nonpecuniary rewards). Since the extent to which crafts-artists attempt to sell their products, or what fraction of their activity is oriented toward sales, cannot be determined, estimates of the determinants of the financial returns to crafts activity must necessarily reflect an average of these two unobserved components. However, by excluding crafts-artists who choose not to sell, the most

162

extreme cases of producing crafts for sale versus producing for other reasons are eliminated.  $\frac{1}{}$ 

A statistically appropriate method to estimate the determinants of hourly earnings is with a Tobit function. This procedure combines both sellers and nonsellers and statistically controls for the clustering of a large number of observations at a limit value (a zero wage rate in this case). Unfortunately, resource constraints did not permit such an estimating approach. Therefore, ordinary-least-squares regression analysis was used. This simpler and less expensive estimation has been applied to the two-thirds of the sample who reported crafts sales. Thus, the regression findings should be interpreted carefully; that is, the results apply to crafts-artists conditional on their choosing to sell. The analysis does not focus on what factors influence the probability of selling, but the estimation could be easily extended to include this decision-making behavior.

Unweighted descriptive statistics for the wage rate analysis sample are presented in Table V.3. Because of the similarities with the full sample, they will not be discussed (note that the explanatory variables used in the wage rate analysis do not overlap entirely with those used previously). The <u>dependent variable</u> for the regression equation is the natural logarithm of the hourly earnings variables. The reason for using this transformed wage variable is to correct for the skewed distribution of the hourly wage variable. The log-wage rate more

ERIC

v.25 . 163

 $<sup>7^{\</sup>frac{1}{2}}$  Of course, it is possible that crafts-artists who report no sales may have desired to sell their products but were unable to do so, although it is difficult to imagine that nothing could have been sold.

TABLE V.3

UNWEIGHTED DESCRIPTIVE STATISTICS FOR THE HOURLY EARNING ANALYSIS (Sample proportions or means with standard deviations in parentheses)

		( - (()
Observed Wage Rate	\$ 2.64	( 3.66)
Log Wage Rate	.12	(1.43)
At Least Some College	.81	
Received Training in Business Skills	.29	
Currently Receiving Crafts Training	.11	( 11 22)
School Art Class Training (years)	1.22	(4.33)
Community Center Training (years)	.58	(4.67)
College/University Courses (courses)	18.84	(34.83)
Apprenticeship (months)	2.08	(8.88)
Adult Education (courses)	.59	( 2.08)
Crafts School Workshops (number)	4.36	(15.04)
Private Professional Lessons (years)	.39	(1.30)
Professional Workshops (number)	2.77	(12.22)
2 to 9 Exhibits	.48	
10 or More Exhibits	.25	
Received an Award	34	
Received Financial Support	.09	•
Exchange Crafts for Necessities	.04	Σ.
Exchange Crafts with Other Artists	.39	
Teach Crafts Workshops	.03	(411, 29)
Years of Age	43.70	(14.28)
Years of Age - Squared	2113.00	(1385.00)
Male	.34	
Produce in Private Workshop	.31	
Produce in Public Workshop	.05	
Primary Medium is Clay	. 19	
Primary Medium is Leather	.01	a.
Primary Medium is Paper	.04	
Primary Medium is Glass	.06	
Primary Medium is Metal	. 17	
Primary Medium is Wood	.16	*
Primary Medium is "Other"	.03	
SAMPLE SIZE	959	

ment of regression analysis. By using the log-wage as the dependent variable, the coefficients obtained from the regression estimation should be interpreted as the percentage change in the wage rate in response to a unit change in a given independent variable, other things held constant. In discussing the empirical results that are reported in Table V.4, the independent variables have been grouped according to their commonality to the theoretical variables discussed in Section B.

Training and Work Experience. The production of crafts products can generally be thought of as the result of combining labor and physical capital inputs according to some technological production process. Within this context, the skills of a crafts-artist acquired through training and experience may have several effects on the production process, sales and costs. First, the technological procedures are not necessarily fixed or known to individuals, and training or experience may lead to more effective and efficient methods to produce more and better products. New techniques can be learned or discovered, new styles and colors experimented with and innovative marketing strategies developed. In this sense, the crafts-artist may act in the capacity of the production engineer, plant manager and marketing director that is analogous to any economic enterprise producing a consumer good. Second, a crafts-artist also serves as a worker whose talents are reflected in the quantity, cost, and quality of goods produced. For workers employed by firms, the value of their productivity is rewarded by a higher rate. For a self-employed crafts-artist, greater productivity is rewarded in higher (net) income because of the better quality, lower cost and/or greater quantity produced per hour by labor input.

TABLE V.4

REGRESSION COEFFICIENTS ON THE LOG-W/GE RATE (Absolute value of t-ratios in parentheses)

At Least Some College (774) <sup>a</sup>	. 32	( 2.96)***
Received Training in Business Skills (276)	.22	( 2.51)**
Currently Receiving Crafts Training (102)	<b></b> 66	( 4.76)***
School Art Class Training (years)	<b></b> 02	( 2.16)**
Community Center Training (years)	01	( 1.48)
College/University Courses (courses)	.003	( 2.30)**
Apprenticeship (months)	.01	( 3.03)**
Adult Education (courses)	005	( .24)
Crafts Schoo: Workshops (number)	008	( 2.52)**
Private Professional Lessons (years)	.003	( .11)
Professional Workshops (number)	002	( .64)
2 to 9 Exhibits	.40	( 3.93)***
10 or More Exhibits	.82	( 6.92)***
Received an Award	.08	( .89)
Received Financial Support	.08	( .56)
Exchange Crafts for Necessities	.39	( 1.82)*
Exchange Crafts with Other Artists	.11	( 1.17)
Teach Crafts Workshops	. 58	( 2.28)**
Years of Age	.05	( 2.56)**
Years of Age - Squared	0006	( 2.95)***
Male	.26	( 2.55)**
Produce in Private Workshop	.41	( 4.47)***
Produce in Public Workshop	54	( 2.82)***
Primary Medium is Clay	. 36	( 2.97)***
Primary Medium is Leather	.58	( 1.69)*
Primary Medium is Paper	.005	( .03)
Primary Medium is Glass	.62	( 3.43)***
Primary Medium is Metal	.70	( 5.60)***
Primary Medium is Wood	. 38	( 2.67)***
Primary Medium is "Other"	.74	( 3.02)***
Constant Term	<b>-</b> 2.05	( 4.52)***
R <sup>2</sup>	.31	
F Ratio for the Equation	13.59	
SAMPLE SIZE	959	

<sup>\*</sup>Significant at the 10% level of confidence, two-tailed test.



<sup>\*\*</sup>Significant at the 5% level of confidence, two-tailed test.

<sup>\*\*\*</sup>Significant at the 1% level of confidence, two-tailed test.

 $<sup>^{\</sup>rm a} {\mbox{For dummy variables}}, \mbox{ the number of sample observations are noted in parentheses.}$ 

According to economic theory, an individual acquires skills through training (formal education, training courses). The decision to invest resources to obtain such training (direct outlays and foregone income or leisure) depends on the perceived benefits vis-a-vis the costs. To the extent that training leads to higher skills, which in turn yield pecuniary returns, one anticipates a positive financial return to such investments in human capital. Of course, the return to crafts training estimated in the data do not capture the nonmonetary returns resulting from the acquisition of new or better skills.

Included in the estimating equation are several types of crafts training (note that they are measured in different units--months, years, courses, number of workshops). From the financial perspective, it appears that taking college courses and serving as an apprentice have a positive and significant impact on hourly earnings. Taking four college courses in crafts is estimated to increase wage rates by approximately one percent. The return to an apprenticeship program has a much larger impact; each month of apprenticeship leads to over a one percent gain in wages, or a 16.5 percent increase for a year's apprenticeship. On the other hand, individuals <u>currently</u> serving as an apprentice forego a subtantial portion of their earnings; on average, they receive a wage rate that is two-thirds lower than otherwise similar crafts-artists not actively participating as an apprentice.

Other types of crafts training are estimated to have negligible effects on hourly earnings. In fact, participation in workshops held by crafts schools or organizations is associated with significantly negative wage repercussions. While it is possible that some types of training may have little impact on hourly earnings, it is difficult to

ERIC Full Text Provided by ERIC

157

explanation, although it must remain conjectural at this point, is that crafts-artists who enroll in workshops may be less inclined to sell their products than those who invest in a college or apprenticeship program.

An analogous argument could be made for some of the other training modes.

Several levels of formal educational attainment were included, but the only significant finding was for crafts-artists who have earned at least a BA degree. An interpretation for this positive relationship between higher education and crafts earnings is that better educated ... individuals may possess general abilities that are useful in organizing a personal business and establishing an effective marketing stategy. Communication skills may also be relevant in talking to customers about their products. Training in business skills is expected to increase sales income, thereby leading to a greater hourly return to crafts activity. The results indicate that individuals who have acquired such training receive a 22 percent higher wage rate than those who have not. As noted above, one should be careful in inferring causality from this empirical relationship. It is possible that crafts-artists with a greater commitment to sales tend to be the ones who are likely to obtain such training, and may have received a higher observed wage in the absence of training because of their greater underlying sales motivation.

Another source by which crafts-artists acquire production and marketing skills is through their work experience (e.g., learning-by-doing). Most analyses of wage rates for employees have found a curvilinear relationship between years of work experience (or age) and wage rates. That is, wage rates generally increase with experience up to

some maximum and thereafter begin to decline. To test for this relationship, a quadratic years of experience variable (experience and experience squared) was included in the log-wage equation. The equation was reestimated replacing a years of age quadratic expression for the experience terms. For this sample, experience has an insignificant effect, while age demonstrated a significant curvilinear relationship that is analogous to what has been found in other studies: that is, the estimated coefficient for age is positive, while that for age-squared is much smaller and negative in value. This implies that wage rates for craftsartists, on average, increase with age, and based on the empirical evidence, the growth in wages continues until an individual is 42.5 years of age, and then it begins to decline.

Artistic Quality. The positive link between crafts training and experience) and hourly earnings is explained by the underlying assumption that training leads to greater skills which in turn raise the value of a crafts-artists's hourly labor input. For any given level or type of training, however, there undoubtedly remains considerable variation in artistic quality and even technical skills. The data provide several measures of the latter which are referred to generally as "quality" indicators. By regressing the log-wage on both the training and quality variables, as well as other factors, one can estimate the separate influence of each set of measures.

183

 $<sup>\</sup>frac{1}{2}$  For private employees, it has been found that wage rates increase with age up to 40-45 years and thereafter begin to decline.

Conceptually, there is a distinction between technical skills and artistic quality since excellence in one may not necessarily imply excellence in the other. Similarly, it is not clear which component is rewarded more highly in the market place through greater sales or higher wage rates. The assumption is that both artistic quality and techni al competence are related to such factors as exhibitions and shows, receipt of financial support and recognition through awards. Whether or not a crafts-artist is a teacher and whether other crafts-artists are willing to exchange products are ancillary measures of skill and quality.

The regression results offer strong evidence in support of the expected association between recent exhibits and hourly earnings.

Crafts-artists who have exhibited their work between 2 and 10 times during the past year are estimated to receive a wage rate that is 40 percent higher than comparable persons with mone or one exhibit. Those with more than 10 exhibits earn an hourly wage over 30 percent greater than those with less than two exhibits. The financial implications of exhibiting one's work are very substantial and confirm the popular wisdom that exhibits are an important if not prerequisite step toward relative financial success.

The way in which exhibition experience leads to greater earnings probably consist of three related phenomena. First, to the extent that skill or quality is positively correlated with the number of exhibits, crafts-artists with greater exhibition experience will sell more of their work, perhaps at higher prices, than less able crafts-artists because of quality differences. Second, the manner in which potential customers learn of and choose among crafts-artists is likely to be influenced by what is observed at exhibitions. Shows serve as both a marketing outlet

and a form of credentializing for crafts-artists. The former affects customer accesss, while the latter provides an institutional quality label. Also related to this phenomenon is the appeal of owning works produced by well-known and recognized crafts-artists.

Third, there may be an element of self-selection in that craftsartists desiring to sell their works may make a greater attempt to
exhibit their products than similarly skilled persons without such a
(strong) disposition. Finally, there is a Catch-22 element to
exhibitions, particularly for museums and one-person shows. Craftsartists with excellent skills and great artistic quality may be thwarted
in their efforts to show their work unless they have had previous shows.
While high quality and skill may ultimately lead to exhibits, such a
self-perpetuating pattern suggests that the measure of past exhibits
should be considered an imperfect barometer of artistic ability.

It is interesting that neither the receipt of financial support to produce crafts nor the receipt of an award or prize for crafts is significantly related to hourly earnings. Presumably these forms of artistic recognition are reflective of skills and quality. One could conjecture that the difference in effect between these variables and the impact of exhibitions is because <u>public</u> recognition is a more important determinant of a crafts-artist's financial success than the receipt of financial support or awards.

Teaching crafts connotes both skill attainment and some degree of recognition. The only statistically significant finding in the regression analysis is for teaching at a crafts school or professional workshop.

Many crafts-artists exchange their work, with others and trade their products with nonartists for living necessities and health care. The latter was found to be significantly correlated with hourly earnings. On one hand, the ability to exchange crafts may be associated with crafts skills (i.e., a desirable product). But on the other hand, individuals who trade their work for living necessities may be those who rely more heavily on crafts for their livelihood which implies a greater orientation toward sales than with other crafts-artists. This variable may be primarily picking up the latter effect.

Production Facilities. The data contain little information about the physical capital inputs that go into producing crafts. Nor is there information concerning the type or efficiency of the production process itself. An indirect measure that is reported by the survey respondents is where they produce their crafts. The estimation, distinguishes between thee general categories: private facilities (studio or workshop), public facilities (cooperative, community or school workshop), and in-home facilities (this category is the excluded variable in the regression).

Relative to producing crafts at home, crafts-artists using a private workshop receive higher hourly earnings, while those using public workshops earn less. An explanation for these statistically significant correlations is subject to speculation. Those who use private facilities are more likely to pay for the space and equipment (rent or mortgage payment if owned), which suggests that such an arrangement requires greater hourly earnings to compensate for the added costs of the facilities. A similar argument can be made for in-home workshops versus public studios in that the former implies foregoing space that could be used for other

ERIC Fred House Boy Eric

172

activities. However, it is unclear the extent to which there are user charges for public workshops, although the above argument holds as long as relative price differences remain in effect.

Demographic Factors. In addition to age which is viewed as a proxy for work experience, the estimation tested for male-female differences. The results suggest that males with otherwise comparable characteristics earn 26 percent higher wage rates than females. This significant difference may reflect: (1) sexual discrimination, (2) unobserved skill advantages for males over females, or (3) a greater tendency for males to engage in selling activities than females. Without more precise information, the three effects cannot be disentangled from one another.

Crafts Media. While statistically controlling for the influence of various forms of personal heterogeneity discussed above, it is interesting to see if there are significant differences in financial returns across crafts media. What is immediately apparent is that crafts-artists whose primary media consist of clay, leather, glass, metal, wood, and "other media" (a residual category) are estimated to earn significantly higher hourly returns than individuals engaged in the comparison group (the omitted fiber medium). 1/Put differently, crafts-artists working in paper and fiber earn somewhere between 35 and 74 percent less per hour than comparable persons involved in other media.

 $<sup>\</sup>frac{1}{2}$  The differences among the more financially attractive media, while large, are not estimated to be significantly different from one another.

Since 39 percent of the sample consists of crafts-artists whose primary medium is fiber, and to the extent that the sample is representative of the general population of crafts-artists who sell their products, then it is possible that the crafts market for fiber products is characterized by a relatively large supply of goods compared with other crafts markets, thus driving down sales and reducing hourly earnings. Although this is a possibility, there are several reasons why this must remain conjecture. One reason relates to the possible differences in sales efforts among crafts-artists in different media. It is possible that crafts-artists engaged primarily in fiber and paper are not as strongly motivated toward sales as with other crafts-artists. Another reason relates to gross versus net income. Media other than fiber and paper may incur greater production costs (space, equipment, materials) which would reduce the net financial return if these expenses were deducted from gross income.

In summary, the regression analysis has yielded a number of interesting insights into what factors are associated with crafts-artists' hourly earnings. Consideration of these empirical relationships may assist in helping crafts-artists improve their financial status. In general, the hourly earnings of crafts-artists are determined by the same set of factors that have been found to influence the wage rates of private employees. Higher skill levels, denoted by past training, educational attainment, experience, and artistic quality have been estimated to have positive earnings implications.

An inherent problem with the analysis is the inability to measure the degree of sales orientation of the sample members. Two similar crafts-artists may spend equal amounts of time producing crafts,

ERIC

Since 39 percent of the sample consists of crafts-artists whose primary medium is fiber, and to the extent that the sample is representative of the general population of crafts-artists who sell their products, then it is possible that the crafts market for fiber products is characterized by a relatively large supply of goods compared with other crafts markets, thus driving down sales and reducing hourly earnings. Although this is a possibility, there are several reasons why this must remain conjecture. One reason relates to the possible differences in sales efforts among crafts-artists in different media. It is possible that crafts-artists engaged primarily in fiber and paper are not as strongly motivated toward sales as with other crafts-artists. Another reason relates to gross versus net income. Media other than fiber and paper may incur greater production costs (space, equipment, materials) which would reduce the net financial return if these expenses were deducted from gross income.

(j)

In summary, the regression analysis has yielded a number of interesting insights into what factors are associated with crafts-artists' hourly earnings. Consideration of these empirical relationships may assist in helping crafts-artists improve their financial status. In general, the hourly earnings of crafts-artists are determined by the same set of factors that have been found to influence the wage rates of private employees. Higher skill levels, denoted by past training, educational attainment, experience, and artistic quality have been estimated to have positive earnings implications.

An inherent problem with the analysis is the inability to measure the degree of sales orientation of the sample members. Two similar crafts-artists may spend equal amounts of time producing crafts,

but a difference in effort to sell will be reflected in the observed hourly returns. It is possible that variation in sales effort may explain the estimated male-female differences and the different \*teturns associated with crafts media.

#### D. CONCLUSIONS AND RECOMMENDATION

Two facets of crafts-artists activity have been investigated in this analysis: the level of involvement as measured by weekly hours of work, and the financial returns to crafts as measured by gross hourly earnings. With regard to the empirical examination, crafts-artists represent an interesting, albeit difficult, group to study because the pursuit of crafts is strongly influenced by both-economic and noneconomic factors. The regression analysis finds, that while craftsartists may be motivated by artistic and other intangible goals, their behavior is also significantly responsive to economic stimuli. In the latter sense, crafts-artists display much of the same economic behavior that has been found in studies of the economic behavior of workers employed in the private sector. For example, workers, tend to benefit from the acquisition of skills and work effort is induced by greater remuneration (pecuniary and nonpecuniary); relationships that seem to hold regardless of the type of work or circumstances of employment. While there are undoubtedly idiosyncracies with respect to classes of workers and individuals within classes, there also appear to be many areas of commonality.

### 1. Primary Findings

With regard to the weekly hours of crafts involvement, the empirical analysis has yielded several interesting findings. There is strong evidence that crafts-artists respond positively to a greater potential wage rate. Whether or not they convert the potential into actual earnings may be largely governed by personal preference regarding sales. For example, two identical workers producing the same type and quantity of products/may differ with respect to their sales effort; hence, the respective hourly earnings will vary. The analysis attempted to circumvent this problem by inputing an hourly wage rate to each crafts-artist based on the regression estimates for the subsample of individuals who reported positive sales income. In the example above, both crafts-artists would have the same value for the imputed wage rate. Presumably, the person choosing to sell less of his or her product derives nonmonetary satisfaction equals the dollar gains to the one with a greater sales orientation. The fact that the estimates using the imputed wage are statistically more powerful than when using the observed wage (including zero for nonsellers) implies that the potential returns to crafts activity is a more appropriate incentive measure.

This finding suggests that a better understanding of what determines the hourly earnings of crafts-artists will also assist in predicting which persons are more likely to spend greater time in producing crafts. The estimation finds a positive financial return to certain types of training in crafts production as well as in business-related skills. The quality and/or recognition of a crafts-artists is found to be a significant factor in explaining hourly wage rates.

The general work experience that crafts-artists accumulate over time also bears a significant monetary impact. While not all of the skill, training, and quality measures are estimated to have a significant impact, the collective evicace is quite conclusive in support of the financial importance of acquiring technical and business skills. Artistic quality, however important, is not considered to be as amenable to training as technical skills and business know-how. Finally, individuals specializing in paper and fiber (the latter being the most popular medium) are estimated to have significiantly lower hourly earnings than otherwise similar crafts-artists. While this may partially reflect a difference in sales orientation, it may, especially for fiber-related crafts, suggest some overcrowding in that sector.

The lack of statistically significant patterns of association are often of interest. In particular, the receipt of financial support and receipt of an award or prize appear to have no systematic relationship with either hourly earnings or weekly hours. However, this is not necessarily a disappointing result, since one rationale for awarding grants to artists is to provide assistance to those with artistic quality but who may be experiencing financial difficulties. The findings suggest that this form of help or recognition is not associated with greater pecuniary returns or greater involvement. However, a test of the impact of such assistance on crafts-artists activity and financial success requires a longitudinal sample in order to estimate any cause-effect relationships that may exist.

Most measures of training or skills do not bear much relationship to crafts-artists' work hours, although a few stand out-exhibition experience, apprenticeship training, and college teaching or giving

private lessons. Given the avocational motive for many crafts-artists, it is not surprising that the acquisition and possession of skills lack a systematic association with involvement.

Crafts do compete with other activities, and the data offer some indirect evidence of this competition for one's time. Individuals from larger families and those who own their own home tend to work less than those in the opposite situation. Rural area residents appear to spend more time in producing crafts than otherwise similar persons living in more populated areas, possibly due to lack of alternative opportunities and/or local crafts traditions. Males are estimated to receive higher hourly earnings than females, although, they tend to spend fewer hours in crafts for a given potential wage rate. This implies that crafts may be a more important leisure activity for females than males and that the latter may be more oriented toward sales. Finally, crafts involvement was not found to be significantly related to the level of income from other household members.

## 2. Problems and Recommendations for Future Research

There are two characteristics of crafts-artists that create analytical difficulties. First, the returns to crafts activity consist of both monetary and nonmentary elements. Hence, an analysis of the financial returns to crafts will understate the true returns to the extent that nonpecuniary remuneration is present. In the present data set, it is not possible to discern the relative importance of these remunerative elements. Second, individuals may engage in crafts as an avocation, as an income producing activity, or both. Persons with similar features may be strongly oriented toward one but not the other

which, in turn, is reflected in different levels of observed hourly earnings. While it is difficult to identify where individuals position themselves on the leisure-work continuum of crafts activity, some of the explanatory variables used in the empirical analysis undoubtedly reflect such a phenomenon. To the extent to which individuals have selected themselves into a sales or avocation position, the interpretation of some of the estimated relationships should be viewed with caution. For example, apprenticeship training may have a positive effect on earnings and work hours, but those who acquire such training may have a strong disposition toward crafts activity and sales, so that they would have had relatively high hours and wages even in the absence of such training. Hence, one should be somewhat guarded in drawing policy inferences from the empirical results.

Another weakness is the use of a gross rather than net hourly earnings measure. Production techniques and costs vary both within and especially across crafts media. Furthermore, variable costs are obviously related to the level of output. The estimates for the wage equation, however, are consistent with many a priori expectations and are similar to the findings for privately employed workers, which lends some credence to the general pattern of relationships that were estimated.

Further analysis with this data set would be a useful undertaking. First, the wage rate analysis should be reestimated with the full sample, using a Tobit model to control statistically for the presence of many observations with a zero wage rate (nonsellers). This procedure would provide a means to confront directly the uncoserved decision to sell or not. Second, tests for inveractions among variables should be made. For example, it is possible that if media and training

were interacted in the regression estimation, the negative association with earnings found for some training modes would persist only for some media types and not for others. Third, it would be desirable to try to exploit the data collected on production expenses to obtain a more accurate picture of net earnings.

Other phenomena could be examined with this data set as well. For example, it would be most interesting to investigate what factors prompt an individual to pursue crafts as a primary occupation, or to engage in crafts as the primary source of income. It would also be useful to explore the relationship between (lack of) financial success and the responses of crafts-artists regarding the most significant obsectles to exhibiting and selling crafts.

Finally, the link between crafts activity and other employment could be investigated with the survey data. For many crafts-artists, the level of involvement is intimately related to other employment.

Fluctuations in the latter's work requirements may significantly constrain the available time for crafts activity. Alternatively, some crafts-artists may choose positions of employment that accommodate their extracurricular interests, especially if crafts serve as a secondary job or as a major source of income.

#### 3. Financial Rewards from Crafts Activity: Some Policy Implications

The preliminary nature of this analysis limits the degree to which policy implications should be drawn from the results. Nevertheless, there are several findings which seem to offer some guidance to policy-makers and crafts organizations concerned with assisting crafts-artists attain greater financial rewards from their endeavors.

Given the relatively low earnings derived from crafts activity, such an outcome would surely be welcomed by crafts-artists.

First, not all types of training appear to increase the hourly earnings of crafts-artists. More specialized or intensive methods, such as college courses and apprenticeships, seem to have a larger financial impact that other programs. In addition, training in business-related skills is significantly linked to earnings which makes good sense in light of the fact that many crafts-artists are in essence self-employed business persons. Greater encouragement and direction along these lines may be warranted, as well as the provision of more of these programs.

Second, the exposure received by crafts-artists through exhibiting their products is a well-known ingredient to making sales. While competitive exhibits and shows are undoubtedly tied to artistic excellence, crafts festivals may have less stringent standards. Parcicipation in, and availability of, crafts shows should also be encouraged to increase exposure to potential buyers and provide a better means by which the public can judge and choose among various artists.

Third, there is no perceptible effect of crafts awards or financial support on either the level of activity or the hourly earnings. This finding suggests that perhaps the resources of the Endowment and other public and private agencies might be more effective if directed toward other strategies to assist crafts-artists. This, of course, is within the context of crafts involvement and monetary returns. There are other objectives of financial support and crafts awards that may, in themselves, justify their continuation.

Finally, there is some indirect evidence that fiber-related crafts may be encountering a relative oversupply of products, as

reflected in the lower returns earned by these crafts-artists. However, this implication must remain tentative until analysis can be carried out that better controls for the sales orientation of crafts-artists and deducts production costs from gross earnings. Fiber crafts-artists may very well be less oriented to sales and have lower production costs than crafts-artists working in other media.



#### VI. CRAFTS PROFESSIONALS AND SUBSCRIBERS

This section of the report is devoted to testing the extent to which the survey of crafts-artists who are members of crafts organizations include three subsets of crafts-artists. The groups are crafts-artists who:

- Sell their crafts at nationally recognized shops and galleries
- o Exhibit at prominent fairs and crafts exhibitions
- Subscribe to crafts periodicals not published by crafts organizations.

Each of these groups may be different from "member" crafts-artists in some crucial way. If this is so, the generalizability of the latter data would be more difficult. There are actually two steps to this testing process. The first is to check on the extent of membership among each of the subgroups; the second is to examine the characteristics of each subgroup to be sure that a similar type of crafts-artist is included in the "member" survey. The higher the proportion of membership among each of the subgroups and the fewer major differences in characteristics (demographic or crafts-related) appear, the more exact will be estimates of the number of crafts-artists and conclusions regarding their characteristics based on the members survey.



vI.1 154

The Methodology Report for this study includes details on selection of outlets and individual crafts-artists. The sampling procedures used do not produce representative samples of their respective sample frames.

#### A. EXTENT OF MEMBERSHIP

Virtually all of the sellers and exhibitors are members of one or more crafts organization (Table VI.1). About three-quarters of subscribers are members. These findings are based on tabulations of the responses to questions asked during a telephone screening interview to sellers and exhibitors and from responses to questionnaires mailed. No telephone calls were made to subscribers; all responses from subscribers are tabulations of questionnaire responses (Table VI.1).

A complicating element is present which must be recognized in making the estimates of membership. Not all respondents to the survey of members reported their membership. In fact, all respondents to that survey were members of organizations that met our definition; however, 18 percent did not report they were a member of any crafts organization when that question was asked in the questionnaire. (As indicated earlier in the report, it was necessary to know the number of organizations each crafts-artist belonged to so that each response would be properly weighted.) Given that less than 100 percent of crafts-artist members reported their membership, it is reasonable to make a corresponding adjustment to the reported membership figures for sellers, exhibitors and subscribers. This adjustment, which is reflected in the membership estimates in Table VI.1, takes the reported percentages of membership in crafts organizations among sellers, exhibitors and subscribers and inflates each of these figures by 1.22 (derived as the ratio of actual membership among member crafts-artists to reported membership or 100/82). As a result all sellers are estimated to be members, as are 95 percent of exhibitors. Membership is less prevalent among subscribers, but at an estimated 73 percent is still substantial.

TABLE VI.1

EXTENT OF MEMBERSHIP IN CRAFTS ORGANIZATIONS IN VARIOUS

CRAFTS-ARTIST SAMPLE FRAMES, 1980-1981

		Surveys of	Crafts-Artists	· · · · · · · · · · · · · · · · · · ·
	Members	Sellers	Exhibitors	Subscribers
Percentage of crafts- artists reporting				v
membership. 2	82%	87%	79%	60%
During telephone contact	NA	89%	78 <b>%</b>	N <b>A</b>
In mail response	NA	85%	81%	60%
Actual (or estimated) membership	100%	100 <b>%</b> <u>b</u> /	95 <b>%</b> * <u>b</u> /	73 <b>%</b> <u>b</u> /
Number of respondents	2,610	172	565	168

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

#### NA = Not Applicable

a/The Survey of Member Crafts-Artists was based on membership in a crafts organization so the percentage reporting such membership should be 100 percent. See Section IV.C for a more detailed description. Percentages for member crafts-artists are weighted; the others are not.

 $\frac{b}{}$ These estimated percentages are calculated by multiplying the proportion 100/82, which is the proportion of member crafts-artists reporting membership, by the reported membership percentage. For sellers of crafts, the result is 104 percent which is shown as 100 percent.

ERIC

consequently, the estimate of 150,000 to 180,000 crafts-artists, developed in Section II based on the data from the members survey, is quite satisfactory as an estimate of all crafts-artists as regards, sellers and exhibitors but may be a slight understatement because not all crafts-artist subscribers are members. In order to judge the extent of the underestimate, reliable information on the size of the unduplicated, subscription list for all crafts-related publications not affiliated with a crafts membership organization is needed. In addition, an assumption that the estimate of 73 percent membership is representative of the subscribers to all such periodicals would be necessary. The estimate of 73 percent membership may not apply because a nonrandom sample of periodicals was used precisely because no reliable estimate of the size of the universe of subscribers was available to permit drawing a representative sample.

## B. COMPARISONS OF CRAFTS-ARTISTS IN DIFFERENT SAMPLES

The second aspect of the test is to determine the extent to which the information gathered from crafts-artists who are members of crafts organizations represents all crafts-artists. In order to facilitate comparisons to the sellers and exhibitors who are members of one or more crafts organization, the information about member crafts-artists whose primary activity is crafts and who spend more than 40 hours per week producing their work is used. On the other hand, tabulations for all member crafts-artists are compared to all subscribers. By using these comparisons, similarities or differences of characteristics can be more readily observed. To the extent that there are similarities in the characteristics reported, there is reason to be ieve that the data from

VI.4

the Survey of Member Crafts-Artists adequately represents the particular subgroup (sellers, exhibitors, or subscribers). Equally important, the information from the members survey can be weighted to properly reflect the significance of components of the crafts-artist population.

Accordingly, in this section, the most important aspects are the extent to which sellers, exhibitors and subscribers characteristics match those of the groups of member crafts-artists to which they are compared.

## 1. Demographic Profile

A glance at the statistical profiles indicates a remarkable similarity between the groups of crafts-artists selected for the comparisons. Sellers and exhibitors more closely resemble those crafts-artists who spend more than 40 hours per week on crafts than they do all member crafts-artists. Also, subscribers are much closer in characteristics to all member crafts-artists than to the subgroup who work at crafts for more than 40 hours per week (Table VI.2).

In examining the individual characteristics, it is tempting to think that sellers and exhibitors are components of the members' survey. If the value reported by sellers, for example, is often above the reported value from the members' data, then the value for exhibitors is often below it or vice versa. Actually, the three sets of information are independent samples drawn in completely different ways from different populations. What is depicted is an indication that, at least for demographic characteristics, there is a consistent pattern of similarities. 1/2

 $<sup>\</sup>frac{1}{}$  There are statistical tests which can be used to determine the probability that two (or more) sets of information describe similar groups in a population. However, among the assumptions or pre-conditions

TABLE VI.2

COMPARISON OF SUMMARY DEMOGRAPHIC CHARACTERISTICS
SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wk	Sellers	Exhibi- tors	Member Crafts- Artists: All Membersa/	Súbscri- bers
Percentage of females	47.8	54.4	48.6	70.2	87.5
Median age	37.9	37.3	35.3	45.2	39.7
Percentage married	65.8	56.4	74.6	71.6	77.4
Percentage white	99.1	95.6	100.0	97.9	99.4
Percentage of Hispanic origin or background	2.4		1.4	1.3	1.2
Percentage completing 4 or more years of college	54.5	79.4	73.6	55.9	59.5
Percentage employed full- or part-time	82.9	95.6	79.2	58.9	64.1
Percentage with crafts as main occupation	96.6	86.2	91.4	23.6	42.1
Percentage who own their home	67.4	66.2	79.2	81.7	82.6
Average (mean) persons in household	2.6	2.5	2.6	2.7	2.9
Number of respondents b/	433-439	65-68	69-72	2589-2632	165-168
Average (mean) household income	\$30,421	\$36,554	\$34,821	\$28,886	\$27 <sup>6</sup> ,058
Number of respondents	140	56	63	1,280	122

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

 $<sup>\</sup>frac{a}{}$  Percentages are calculated using weighted responses; the number of respondents is unweighted.

 $<sup>\</sup>frac{b}{R}$  Range shown is the lowest and highest number of respondents reporting the characteristics shown.

The only major differences involve the percentage of females which is larger for subscribers than for all members and the greater percentages of sellers and exhibitors who have completed 4 cr more years of college compared to member crafts-artists.

# 2. Craft's Characteristics

Although the populations examined appear to have the same personal characteristics, there is more variation when certain aspects of their crafts involvement are studied. For example, the extent of females in the population of subscribers is responsible for the high share reported for fiber as a primary media (Table VI.3). This can be confirmed by noting the low percentages for metal and especially wood, which are primary media activities that attract more males among all member crafts—artists. There are somewhat more potters among sellers and fiber crafts—artists among exhibitors than are found among member crafts—artists who spend more than 40 hours per week producing. Also, a larger share of sellers tend to have worked in crafts for more than 10 years. Otherwise, the incidence of commissioned work, number of years in crafts and hours per week are quite close for sellers and exhibitors and subscribers and their respective comparison groups (Table VI.3).

for this is that each set of information be representative of the population from which it is drawn. There is no such assurance regarding the extent to which sellers, exhibitors or subscribers included in this data set represent all sellers, exhibitors, or subscribers. However, the great similarities observed make it unlikely that there are major differences in demographic characteristics among the components.

TABLE VI.3

MEDIA AND INVOLVEMENT IN CRAFTS
SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wk	Sellers	Exhibi- tors	Member Crafts- Artists: All Members 4	Subscri- bers
Primary Media:					
Fiber	16.4	11.8	21.7	41.8	67.0
Clay	31.6	44.0	34.9	17.4	20.4
Leather or paper	3.4	4.4	4.4	4.9	w
Glass or other	11.4	13.2	11.5	6.8	6.6
Metal	16.7	-11-7	14.5	12.5	3.6
Wood	20.5	14.7	13.0	16.6	2.4
 Totai	100.0	100.0	100.0	100.0	100.0
Number of respondents	249	68	<sup>1</sup> 69	2536	167
Percentage working in primary media for more than 10 years	47.9	69.1	44.4	44.6	45.2
Number of respondents	256	68	72	26 27	<sup>°</sup> 168
Average number of years spent working in crafts	13.2	13.3	11.4	16.3	16.8
Number of respondents	254	68	72	2608	167
Average number of hours per week spent:				<i></i>	
Producing crafts Marketing crafts	54.9 7.2	44.3 7.5	42.7 9.1	19.5 2.3	22.6 3.2
Number of respondents	228	62	66	2375	155
Percentage doing commis- sioned crafts work	77.6	72.0	77.8	45.8	45.2
Number of respondents	256	68	72	2628	168

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

ú

 $<sup>\</sup>frac{a}{}$ Percentages are calculated using weighted responses; the number of respondents is unweighted.

In regard to training received, about 90 percent of craftsartists in all subgroups agree they are satisfied or very satisfied with
their past crafts training. There is a little more variation regarding
the share now receiving training and those planning to acquire additional
training. Sellers and exhibitors each are undergoing less training and
desire less additional training than do member crafts-artists (Table
VI.4). Subscribers also are taking less training, but desire more in the
future. Training in business skills is more highly valued by exhibitors
than sellers; subscribers are also interested in acquiring more business
skill training.

The pattern of satisfactions reported by crafts-artists, whether they are sellers, exhibitors, subscribers or in one of the comparison groups, is the same. Creative expression is most highly ranked by all (Table VI.5). The second most important satisfaction for sellers and exhibitors and their counterparts among the comparison groups is "life's work," followed by a sense of accomplishment and their ability to derive sufficient income from crafts. All of these are reflected in the satisfactions reported by member crafts-artists working more than 40 hours per week at crafts. "A diversion from daily routine" replaces sufficiency of income as an important satisfaction for subscribers.

There is nearly the same unanimity for goals to be obtained in the next 5 years. Increased income from the sale of unique works was mentioned by about three-quarters of the "professional" crafts artists.

Developing artistic competence and win recognition award are also very prevalent; each was reported by 45 to 55 percent of sellers of exhibitors and their comparison group. Subscribers along with all member crafts-artists ranked devoting more time to crafts and winning recognition/award about equally—56 to 61 percent (Table VI.5).

TABLE VI.4

COMPARISON OF TRAINING CHARACTERISTICS SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wka	Sellers	Exhibi- tors	Member Crafts- Artists.: All Members	Subscri- bers
Percentage satisfied or very satisfied with					
past training in crafts	92.8	89.7	92.9	89.6	88.0
Number of respondents	254	68	70	2593	167
Percentage now receiving crafts training	3.5	1.5	1.4	14.5	8.3
Number of respondents	256	68	72	2586	168
Average hours per week of training	1.4	<b>#</b>	*	1.0	<b>*</b>
Number of respondents	256			2586	4.
Percentage with training in business skills	30.2	20.9	31.0	29.2	29.9
Number of respondents	251	67	71	1919	107
Percentage with no business training with a desire for training in business		51.9	72.9	57.4	75.3
skills	58.8 176	52	48	1304	73
Number of respondents	170		<b>→</b>		
Percentage planning to acquire additional crafts skills	57.5	42.6	54.9	67.2	76.5
Number of respondents	254	68	71	2603	166

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

 $<sup>\</sup>underline{\mathtt{a}}/\mathtt{Percentages}$  are calculated using weighted responses; the number of respondents is unweighted.

<sup>#</sup> Average not shown because there are fewer than 50 respondents in category.

TABLE VI.5

COMPARISON OF SATISFACTION FROM CRAFTS AND CRAFTS-RELATED GOALS SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wk	Sellers	Exhibi- tors	Member Crafts- Artists: All Members	Subscri- bers
Percentage of crafts-		<del></del>			
artists with most				V-	
important satisfactions			<del>-</del> -		
from crafts work:					
Means of creative				•	•
expression	39.7	41.2	37.5	42.4	55.4
Life's work	25.0	38.2	25.0	6.0	8.3
Sense of accomplishment		8.8	8.3	17.1	19.6
Produce works for sale	7.1	1.5	5.6	3.5	2.4
Derive sufficient incom	e 10.9	7.4	16.7	2.0	3.0
Diversion from daily					
routine	1.3	1.4	1.4	8.0	10.7
Other	1.3	2.9	5.6	0.6	1.8
				العلات العلات العلاق العل	
Number of respondents	256	68	72	2587	168
Percentage with the					
following specific					
goals for the next 5					
years:					
Win recognition/award	44.2	55.8	44.4	31.5	35.7
Develop artistic compe-	•				
tence	54.2	51.5	50.0	55.9	61.3
Increase income from					
sale of:					
Unique works	72.2	77.9	75.0	41.4	50.0
Production works	33.6	28.0	33.3	19.6	22.0
Devote more time to					
crafts	18.1	30.9	22.2	58.8	58.9
Other	8.5	16.2	23.6	5.4	9.5
Nu 'er of respondents	255	68	72	26 28	168

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

 $\underline{a}/Percentages$  are calculated using weighted responses; the number of respondents is unweighted.

By definition, sellers and exhibitors participate in crafts exhibits; also, the share of subscribers who exhibit matches the percentage of all crafts-artists who exhibit (Table VI.6). A little more surprising is that sellers exhibit more frequently than those labeled "exhibitors." Sellers have also won more awards in the past 3 years than exhibitors. Subscribers are again indistinguishable from all crafts-artists.

In addition to art/crafts fairs (92 percent), exhibitors participated in juried events (88 percent) and one-person invited exhibits (68 percent). Sellers were more attracted to group-invited exhibits (100 percent), one-person exhibits (79 percent) and group competition exhibits (78 percent). The most popular exhibiting activities for member crafts-artists who work at crafts 40 or more hours per week were group-invited exhibits (72 percent) and juried events and art/crafts fairs (each at 60 percent). Among subscribers and all crafts-artists, group-invited exhibits and art/crafts fairs were most popular.

Nearly all respondents among members who worked at crafts for more than 40 hours per week, sellers or exhibitors produced crafts for sale; about two-thirds of all member crafts-artists and subscribers did the same (Table VI.7). Large proportions (76 to 87 percent) of sellers, exhibitors and crafts-artists in their comparison group report their primary income is from the sale of crafts, while about a quarter of subscribers and all crafts-artists do. More sellers (93 percent) list art/crafts galleries as the place they sell their work than exhibitors who give just preference to art and crafts fairs (96 percent). Art/crafts fairs are the first choice of those subscribers who sell, as is also true for all member crafts-artists (42 and 46 percent respectively).

ERIC PROVIDENCE PROVIDENCE

TABLE VI.6

COMPARISON OF EXHIBITING CHARACTERISTICS SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wk	Sellers	Exhibi- tors	Member Crafts- Artists: All Members	Subscri- bers
Percentage exhibiting					
their crafts in the	ومعارات المعدورات المتدورة				<u></u> -
last 12 months	89.0	100.0	98.6	82.2	82.0
Number of respondents who exhibit	234	68	72	2176	167
Percentage exhibiting more than 5 times or continu- ously in past 12 months	60.9	77.9	70.4	32.9	33.0
Number of respondents	233	68	71	1731	94
Percentage who won an award or prize in last 3 years	48.4	61.8	46.7	42.2 1720	44.1 93
Númber of respondents	231	68	71	1720	93
Percentage participating in the following activities in the last 12 months:					
Art/crafts fairs	59.6	55.9	91.7	38.0	36.5
One-person exhibits	39.1	79.4	29.2	15.3	17.4
Group-invited exhibit	71.6	100.0	68.1	36.1	44.9
Group competition exhibi		77.9	38.8	20.7	24.6
Juried events	60.6	57.4	87.5	29.8	31.7
Unjuried community event	19.3	17.6	19.4	16.4	21.0
Juried community event	19.3	13.2	26.4	8.8	12.0
Number of respondents	254	68	72	2196	167

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.



 $<sup>\</sup>frac{a}{}$  Percentages are calculated using weighted responses; the number of respondents is unweighted.

 $<sup>\</sup>frac{b}{}$ Excludes respondents who reported they never exhibit their crafts work.

TABLE VI.7

COMPARISON OF SELLING CHARACTERISTICS
SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Ų.	Member Crafts- Artists: Primary Activity		E <b>x</b> hibi-	Member Crafts- Artists: All	Sub <b>s</b> cri-
Characteristics	40 + hr/wk	Sellers	tors	Members **	bers
Percentage producing -					
works for sale	97.4	98.5	98.6	67.6	63.7
Number of respondents	256	68	72	2629	168
Percentage with primary income from sale of craft	s 80.4	76.1	87.3	23.1	25.2
Number of respondents	252	68	71	1910	107
Percentage who sell their work through:					
Art/crafts fairs	60.0	57.4	95.8	46.2	41./
Own shop/studio	69.4	64.7	68.1	35.3	39.9
Crafts, shops	47.7	60.3	65.3	24.7	31.5
Other retail outlets	25.7	36.8	50.0	12.8	14.3
Cooperatives	8.6	4.4	11.1	6.5	3.6
Commissions	52.8	47.1	61.1	30.8	33.3
Art/crafts galleries	60.7	92.6	73.6	28.7	32.7
Mail orders	15.3	16.2	22.2	5.6	4.2
Wholesalers	17.3	14.7	30.6	6.1	5.4
Work group meetings	1.8			6.0	7.7
Number of respondents	256	68	72	2625	168

SOURCE: Surveys of Crafts-Artists, Mathematica Policy Research, Inc.

 $\frac{a}{}$  Percentages are calculated using weighted responses; the number of respondents is unweighted.

The crafts-related income of sellers, exhibitors and subscribers is greater than that of their counterparts in the two comparison groups (Table VI.8). This pattern is also evident in the pattern of lower non-craft sources and higher crafts expenses. Exhibitors have a higher net income from crafts, even though both sellers and exhibitors reported nearly the same average gross income from crafts. Also, subscribers do not net as much as sellers or exhibitors.

The income characteristics of sellers, exhibitors and subscribers demonstrate the most variation from their comparison groups of any of the categories investigated. In large measure this could result from the specific focus of the sellers and exhibitors samples on crafts-artists who are especially active marketers of their crafts. For subscribers, there is not as obvious an explanation.

Nonetheless, considering all the demographic and crafts-related characteristics reviewed in this section, there are remarkable similarities among the seller and exhibitor crafts-artists and respondents to the members survey who are highly involved in crafts, and also between subscriber crafts-artists and all respondents to the survey of members. Because of the similarities, it appears the survey of member crafts-artists includes many crafts-artists who closely resemble those who sell or exhibit and those who subscribe. Moreover, because the subscribers information includes characteristics of both members and nonmembers, it is unlikely, given the similarities shown here, that the universe of subscribers contains crafts-artists who are radically different from member crafts-artists.

198

TABLE VI.8

COMPARISON OF INCOME CHARACTERISTICS
SURVEYS OF CRAFTS-ARTISTS, 1980-1981

Characteristics	Member Crafts- Artists: Primary Activity 40 + hr/wk	Sellers	Exhibi- tors	Member Crafts- Artists: All Members	Subscri- bers
Average gross income of crafts-artists from:		= - · ·	-		<u>-</u> ·
Own crafts sales	\$11,568	\$20,778	\$20,303	\$ 2,493	\$ 5,362
Number of respondents	229	61	70	2348	150
Crafts-related sources	\$ 1,133	\$ 481	\$ 23	\$ 303	\$ 119
Number of respondents	219	62	69	2327	155
Estimated sales and crafts related gross income of respondents  Average gross income of crafts-artists from	\$12,701	\$21,259	\$20,326	\$ 2,796	\$ 5,481
teaching crafts	\$ 886	\$ 2,884	\$ 1,333	\$ 1,109	\$ 2,322
Number of respondents	220	59	65	2352	155
Average non-crafts income of respondents	\$ 2,823	\$ 1,290	\$ 1,466	\$ 8,008	\$ 5,231
Number of respondents	181	58	67	1734	135
Total estimated average gross income of respondents	\$16,410	\$25,433	\$23,125	\$11,913	\$13,034
Average crafts expenses of respondents	\$10,842	\$17,193	\$12,299	\$ 2,627	\$ <b>2,638</b>
Number of respondents	210	68	72	2169	168
Estimated net crafts income of respondents c/	\$ 1,859	\$ 4,066	\$ 8,027	\$ 169	\$ 2,843

SOURCE: Surveys of Crafts-Artists, Mathematica Poli. Research, Inc.

 $<sup>\</sup>frac{a}{r}$  recentages are calculated using weighted responses; the number of respondents is unweighted.

 $<sup>\</sup>frac{b}{}$ Estimated based on sum of averages shown.

c/Estimated by subtracting average crafts expenses from estimated sales and crafts-related gross income of respondents.

frequent are private lessons (17 percent), public schools (11 percent) and adult education courses, community centers and professional workshops (from 8 to 10 percent each). An average of 2.6 hours per week is spent teaching crafts. About 15 percent teach advanced students with larger percentages instructing intermediate and beginning students (Table IV.22).

Gross income from crafts sales is highest for glass or other media or combinations of media (\$6,000), metal (\$4,350) and clay (\$4,100). Clay and glass or other media also have the highest net incomes (\$1,750 and \$1,300 respectively), while metal has a slight net loss (\$-76). Gross income from teaching is highest, on the average, for potters, nearly \$2,500, more than twice the average for all crafts-artists. Crafts-related sources contribute about \$800 toward gross income of glass or other media and combinations of media crafts-artists (Table IV.21).

The amounts of gress and net income increase as the extent of involvement rises. Crafts-artists with crafts as a primary activity and working more than 40 hours a week report gross income from crafts sales of \$11,600 and net incomes of \$1,500. Only slightly less net income is earned by those working less than 40 hours per week. Crafts-artists with leisure or occasional involvement report less than \$600 per year in gross income from crafts and net losses ranging from \$100 to \$400 (Table IV.21).

The difference between crafts gross income and net income is crafts expenses which for all crafts-artists tote up to \$2,627, almost equal to the average crafts-related income of \$2,736. On the average,

Mathematica Policy Research, Inc.



New Jersey 08540 609-799-2600 September 1980

Dear Crafts-Artist:

The National Endowment for the Arts has contracted with Mathematica Policy Research to conduct a national survey of crafts-artists. We have drawn our sample from people who belong to crafts organizations, and/or sell through major sales outlets and/or subscribe to crafts periodicals.

The Endowment is attempting to assist American craftspersons on both the local and national levels. In 1978, it funded a Census of Crafts Organizations which represented a major step in obtaining information needed to carry out this important project. The present survey is equally important because it will add significantly to the presently available data on such an important aspect of American life.

The enclosed questionnaire has been designed to require a minimum of your time. Most questions can be answered by simply circling a code number. Although your participation is voluntary, we hope you will work with us to ensure that craftspersons like yourself are fully represented in this survey. To protect confidentiality, the survey results will be reported in aggregate form only and individual craftspersons will not be identified.

A prepaid addressed envelope is enclosed. We are eager to complete the study and a prompt reply would be most appreciated. If you are not a craftsperson or you do not collect or read about crafts, visit crafts shows or do not have a crafts-related occupation, please complete the enclosed postcard but leave the questionnaire blank and return them both in the envelope.

Sincerely,

Audrey McDonald Project Director

AM:ha Enclosures

P.S. If you have any questions about the study or this questionnaire, please feel free to call me at (609) 799-2600, extension 2542.



. Which	of the following best describes your current involvement with crafts
	(CIRCLE THE ONE CODE NUMBER THAT BEST APPLIES)
	CREATING/PRODUCING CRAFTS IS MY PRIMARY OCCUPATION 1
	CREATING/PRODUCING CRAFTS IS MY SECONDARY OCCUPATION 2
	I AM A FULL-TIME TEACHER OF CRAFTS
	I AM A FULL-TIME STUDENT OF CRAFTS 4
	MY CRAFT IS MY MAIN LEISURE ACTIVITY 5
	I DO OCCASIONAL WORK IN CRAFTS6
	OTHER (SPECIFY):
	<u> </u>
	any years altogether have you been working in crafts?
2. How ma	
. How ma	YEARS
	cind of training have you had throughout your involvement in crafts?  HOW MANY?
	cind of training have you had throughout your involvement in crafts?  HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS . YEARS
	tind of training have you had throughout your involvement in crafts?  HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS
	LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS
	HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS  COMMUNITY CENTER OR RECREATION PROGRAM COURSES
	LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS  COMMUNITY CENTER OR RECREATION PROGRAM COURSES  COLLEGE/UNIVERSITY COURSE IN ART COURSES
	HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS  COMMUNITY CENTER OR RECREATION PROGRAM COURSES  COLLEGE/UNIVERSITY COURSE IN ART
• What 1	LESSONS FROM FAMILY MEMBERS OR FRIENDS
	HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS  COMMUNITY CENTER OR RECREATION PROGRAM COURSES  COLLEGE/UNIVERSITY COURSE IN ART
3. What 1	HOW MANY?  LESSONS FROM FAMILY MEMBERS OR FRIENDS YEARS  SELF-TAUGHT YEARS  SCHOOL-ART CLASS, TEACHER DEMONSTRATION YEARS  COMMUNITY CENTER OR RECREATION PROGRAM COURSES  COLLEGE/UNIVERSITY COURSE IN ART COURSES  APPRENTICESHIP

. 0



rja.			YE	s	.I	WAS		(SK	IP 1	0 1	0.)							•	1
							-												
			NO																
	,	*	NO	• •	• •	•	• •	(01)	-;		••,								
,	9A.	How long	have	you 1	beet	n an	apı	ren	tice	?		f .		YEAI	RS				
,	9 <b>B</b> •	On the av		, ho	w ma	an y	ho <b>u</b> :	rs p	er (	æek	do	you	wor	k a	s an	l.			
			·						Type:					HOU	RS_				_
	9C.	Including	the	time	yσ	u, ha	ve	alre	ady	spe	n <b>t</b> a	ıs a	n ap	pre	ntic	e,	ho <b>w</b>		
		long is y	our a	ppre	nti	cesh	ip?												
	, .	•	v	.2										YEA	RS			—	_
		rm	3			-	h1	ino	082	fra	m V	1112	work	28	an				
	9D•	What is y		vera Ç∂	ge i	mont	nr. C	Tuc	ome	LTC	om Ac	Jul	WOLF	. as	au				
			- 2										MONT	HLY	\$	_			
			111	IPAID					•					•					C
•	<b>.</b> ,	ou have a	ÝI	ES .	•				•						• •	•			1
			N(		•				•		·• •	. •	•		• •			. •	C
			4								J								
· ~	(PLF	thich of the ASE CIRCLE	ne foi	THE	MUM YO	BERS U C	S TH ONSI	AT A	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE 1.)	R TO	) T	HIE
• • •	(PLF	ASE CIRCLE	ne foi	THE	MUM YO	BERS U C	S TH ONSI	AT A	TO	Y. BE !	THE	PRI	MARY	ME.	DIU	ETTE 1.)	R TO	) T:	HŒ
• • •	(PLF	ASE CIRCLE	ne fo E ALL EDIUM	THE	MUM YO	BERS U CO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	) T.	HE
* *.	(PLF	ASE CIRCLE OF THE MI	ne foi E ALL EDIUM F	THE	YO YO	BERS U CO CIRO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	) T	HE
• • •	(PLF	EASE CIRCLE FOR THE ME	ne fo E ALL EDIUM F	THE THAT	YO YO	BERS U CO CIRO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO		HE
	(PLE	EASE CIRCLE F OF THE MI A B	ne fo E ALL EDIUM F C	THE THAT IBER •	YO YO	BERS U CO CIRO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	) T	HE
4.	(PLE	CASE CIRCLE T OF THE MI A B	ne fol E ALL EDIUM F C LL	THE THAT IBER. LAY.	NUM YO (	BERS U CO CIRO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	• •	HE
4.	(PLE	EASE CIRCLE T OF THE ME A B C	ne fo: E ALL EDIUM F C; Li P	THE THAT  IBER  LAY  EATHE	NUM YO (	BERS U CO CIRO	s th onsi	AT A DER ALL	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	• •	1
4.	(PLE	EASE CIRCLE T OF THE ME A B C D	ne foi E ALL EDIUM F C Li P	THE THAT  IBER  LAY  EATHE  APER  LASS	NUM YO (	BERS U CO	S THONS I	AT ADER	TO	Y. BE !	THE! OUR	PRI	MARY	ME.	DIU	ETTE	R TO	• •	1

	1 1 1 1 1 1 1 1 1 1 2 1 2 2 2 2 2 2 2 2
15B. How man	ny commissioned works did you do in the last twelve months?
	NONE
15C. What wa	as the approximate dollar amount of your lowest and tommissioned sale in the last twelve months?
	LOWEST
	HIGHEST
. What are the craft? PLEA	chief satisfactions that you derive from working in your SE RANK THE FOLLOWING MOST IMPORTANT SATISFACTIONS:
	MEANS OF CREATIVE EXPRESSION
	LIFE'S WORK
	SENSE OF ACCOMPLISHMENT
	PRODUCING WORK THAT OTHERS WILL PURCHASE
₹	DERIVING A SIGNIFICANT PORTION OF INCOME FROM YOUR CRAFT
	DIVERSION FROM DAILY ROUTINE
•	OTHER (SPECIFY)
	··

Is your work primarily influenced by:
(CIRCLE THE ONE CODE NUMBER THAT BEST APPLIES)
AMERICAN FOLK THEMES
ETHNIC/CULTURAL TRADITIONS
CURRENT TRENDS IN YOUR CRAFT
OTHER ART FORMS
A MOVEMENT OR SCHOOL IN THE ART WORLD 5
OTHER (SPECIFY)
· [[]
19A. Please describe this influence:
[[]
Do/did your parents work in crafts?
YES CRAFT WORK IS/WAS THEIR PRIMARY SOURCE OF INCOME
CRAFT WORK IS/WAS THEIR SECONDARY SOURCE OF INCOME
CRAFT WORK IS/WAS A LEISURE TIME ACTIVITY 3
NO
Do other members of your household work in crafts?
YES SAME CRAFT MEDIUM AS MINE 1
ANOTHER CRAFT MEDIUM (SPECIFY)
NO
*

		<i>M</i>
Where	do you sp	end most of your time working in your craft?
		(SURCLE THE ONE CODE NUMBER THAT BEST APPLIES)
		IN YOUR HOME
		PRIVATE STUDIO OR WORKSHOP
		COOPERATIVE STUDIO OR WORKSHO?
		COMMUNITY STUDIO OR WORKSHOP
		SPACE PROVIDED BY SCHOOL
		OTHER (SPECIFY)
		* Committee * Committee
25 <b>A.</b>	Do you ow free of c	on or rent your primary workspace or is it provided
		own
		RENT
		FREE OF COST
25B.	Do you sh	are your workspace with other crafts-artists?
	•	
	·	YES
	·	YES
	·	
		NO
Do yo	ou prod <b>u</b> ce	
Do yo	ou prod <b>uce</b>	NO
Do yo	ou prod <b>u</b> ce	NO
Do yo	ou produce	NO
		NO
		works for sale?  YES
		works for sale?  YES
		works for sale?  YES
	Is your p	works for sale?  YES
26A.	Is your p	works for sale?  YES
26A.	Is your p	Works for sale?  YES
26A.	Is your p	works for sale?  YES



30. Have you taught crafts in the last three years?
YES
NO (SKIP TO 31.) 0
30A. Where do/did you teach crafts?
(CIRCLE ALL THAT APPLY)
CRAFTS SCHOOL
PROFESSIONAL WORKSHOPS
PRIVATE LESSONS
PUBLIC SCHOOLS
ADULT EDUCATION COURSES
COMMUNITY CENTER/ARTS LEAGUE
CRAFTS SHOP
ART SCHOOL OR COLLEGE
COLLEGE OR UNIVERSITY
OTHER, (SPECIFY)
30B. Approximately how many hours per week do/did you teach?
HOURS PER WEEK:
HOURD FEX WEEKI
30C. Are/were your students:
(CIRCLE ALL THAT APPLY)
BEGINNING
INTERMEDIATE
ADVANCED

33. Now often did you show or exhibit your work in the last tw	melve months?
(CIRCLE THE ONE CODE NUMBER THAT BE	
NOT AT ALL	1
ONCE	
TWO TO FIVE TIMES	3
SIX TO TEN TIMES	4
MORE THAN TEN TIMES	5
CONTINUOUSLY	4
34. In the 1 st three years have you won any awards or prizes	for your work?
YES	1
No	0
	whihi+i~~
35 What are the main problems that you face in showing or ex your work?	
(CIRCLE ALL THAT APPLY)	)
TOO FEW OUTLETS	1
INSUFFICIENT DISPLAY SPACE	
NO APPROPRIATE EXHIBITION/SELLING SPACE.	
COMMUNITY POORLY INFORMED ABOUT EXISTING EXHIBITS, SHOPS	•
OTHER (SPECIFY)	<del></del> .
	lll
NONE	0
36. Where do you think are the best places to show or exhibi	t your work?
(CIRCLE ALL THAT APPLY	
AT ART/CRAFTS FAIRS	
YOUR OWN SHOP	
// 5.42 15 54.55	1
1	1
ARTS OR CRAFTS GALLERIES	
ART SCHOOL OR COLLEGE	
COLLEGE OR UNIVERSITY	1
OTHER (SPECIFY)	
DO NOT SELL WORKS (SKIP TO 40.) .	, 0
20 101 2000 101 1010	

	id you and other persons living in your house the from the following sources before taxes ARE ACCEPTABLE—REMEMBER, THIS INFORMATION ORM.)	and other deductions?
	YOUR OWN INCOME FROM CRAFTS SALES	\$
	FROM CRAFTS- RELATED SOURCES	\$
	FROM TEACHING CRAFTS	\$
	FROM NON-CRAFTS INCOME	\$
	CRAFTS INCOME OF OTHER HOUSEHOLD MEMBERS	\$
·	OTHER INCOME OF ALL OTHER HOUSEHOLD MEMBERS	\$
	DUES TO CRAFTS ORGANIZATIONS	\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$ \$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$\$ \$\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$\$ \$\$
	PUBLICATIONS/BOOKS RELATED TO YOUR CRAFT	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$

	YES HOW MANY CRAFTS ORGANIZATIONS?
e .	NO (SKIP TO 45E.) 0
	list the crafts organizations to which you belong that are portant to you:
	[[]
	[[]
45E. collect	
	YES
	NO
. Are you	
yee s s s	FEMALE
	MALE
* ****	THAN ON DIDITI
. When were you	
. What is your	
	NEVER MARRIED
	MARRIED
	widowed/divorced/separated
). What is vour	current employment status:
10 your	(CIRCLE THE ONE CODE NUMBER THAT BEST APPLIES)
	WORK FULL-TIME (30 HOURS OR MORE PER WEEK) (GO TO 49A.) 1
	WORK PART-TIME (GO TO 49A.) 2
	FULL-TIME HOMEMAKER OR HOUSEWIFE . (SKIP TO 50.) 3
	FULL-TIME STUDENT (SKIP TO 50.) 4
	FULL-TIME STUDENT (SKIP TO 50.) 4  RETIRED (SKIP TO 50.) 5  UNEMPLOYED (SKIP TO 50.) 6

49D. If crafts production is your main occupation, did you have another occupation before coming to crafts?
YES WHICH? SELECT A CATEGORY FROM THE LIST IN QUESTION 49C. AND PLACE THE NUMBER OF THE CATEGORY HERE:
NO
DO NOT PRODUCE CRAFTS
50. What is the highest level of school you have completed?
SOME HIGH SCHOOL
HIGH SCHOOL DIPLOMA
SOME COLLEGE
TWO YEAR COLLEGE DEGREE 4
FOUR YEAR COLLEGE DEGREE 5
TECHNICAL/VOCATIONAL DEGREE OR CERTIFICATE 6
GRADUATE WORK
MASTERS DEGREE
DOCTORATE, PROFESSIONAL DEGREE
OTHER (SPECIFY)
50A. What was your major or field?
[[_]
50B. Did you receive G.I. Bill benefits to enable you to attend school or to finish your degree work?
YES
NO
51. Do you own or rent your home?
OWN
RENT



Are there any general commen position of the crafts-artis		oout the
	<u></u>	 [[]

THANK YOU VERY MUCH FOR FILLING OUT THIS QUESTIONNAIRE. PLEASE RETURN IT IN THE ENCLOSED POSTPAID ENVELOPE.

OMB # 128580004

Expiration Date: 12/31/80

## APPENDIX B

# COMPARISON OF SURVEY RESULTS BY DATA COLLECTION METHOD

Because both bias and sample variability are at the minimum levels provided for in the sample design when the response rate is 100 percent, any substantial deviations from high response levels call for investigation. We agreed with the National Endowment we would conduct such an investigation if the response rate to the Survey of Crafts-Artists who are Members of Craft Organizations did not reach 75 percent. As a result of the mail procedures, a response rate of 69.3 percent was obtained.

The hypothesis to be examined is that different methods of data collection obtain responses from different components of the sample; those who respond without need for follow-up activities are different in personal and craft-related characteristics from those who respond after being "followed up." 1/The stated hypothesis is actually a surrogate for one which would compare respondents to non-respondents. However, because examination of the latter hypothesis is impossible, if differences between those who respond prior to follow-up activities and those who respond as a result of follow-up activities are different, non-respondents are assumed to be more like those who required follow-up to achieve a response.



 $<sup>\</sup>frac{1}{\ln}$  this instance the follow-up consisted of an attempted telephone contact with a sample of all non-respondents to the mail procedures.

On the basis of chi square statistics and their related probabilities, we find there are statistically significant differences between those respondents who completed a questionnaire as a result of a telephone follow-up and those who completed a questionnaire without such follow-up (Tables B.1 and 2).

crafts-artists who responded to the telephone follow-up tended to engage in crafts more as a leisure activity compared with crafts-artists who were mail respondents. Telephone respondents were more oriented toward wood as their primary media, as opposed to metal and clay. Also, telephone respondent crafts-artists worked slightly more hours per week (21 compared to 19.4) than mail respondent crafts-artists. The former have upent one less year in crafts compared with mail respondents (Table B.1).

Smaller proportions of crafts-artists who responded as a result of the telephone follow-up compared with those who responded to the mail procedures:

- o Produce crafts for sale
- o Derive their primary income from crafts
- o Teach crafts
- o Visited crafts museums in the 12 months prior to completing the questionnaire
- o Collect crafts.

In addition, the number of crafts publications read in the last 12 months and the number of memberships in crafts organizations are fewer for those who responded as a result of the telephone follow-up compared with those who responded to the mailed request.

ERIC Fred House Boy Eric

 $\frac{211}{1}$ 

TABLE B.1

COMPARISON OF CRAFTS-RELATED CHARACTERISTICS BY MODE OF DATA COLLECTION: SURVEY OF MEMBER CRAFTS-ARTISTS, 1980

	Telephone Response	Mail Response	X <sup>2</sup> Statistic Significance Level
Percentage of crafts-artists by level of intensity of			
craft activity			
Major activity			
Under 40 hours per week	12.2	17.1	
40 hours or more per week	8.3	8.4	
Secondary activity	19.6	17.9	
Leisure Activity			
Under 10 hours per week	27.4	17.7	
10 hours or more per week	17.3	13.6	
Occasional Activity	7.1	14.4	
Other	8.1	10.9	
Total	100.0	100.0	p < 0.01
Primary media			
Fiber	39.9	40.1	
Clay	14.3	16.8	
Leather	0.1	1.0	
Paper	4.1	3.8	
Glass	2.7	3.2	
Metal	8.6	12.1	
Wood	23.5	15.6	
Other	5.8	7.4	
Total	100.0	100.0	p < 0.01
Average hours per week spent	01.0	10 /	N.A
producing crafts	21.0	19.4	A.A
Average number of years		16.	37 A
involved in crafts	15.4	16.4	N.A
Percentage producing crafts		(0.0	
for sale	60.3	68.0	p < 0.01

Table B.1 (Continued)

	Telephone Response	Mail Response	X <sup>2</sup> Statistic Significance Level
Percentage with primary income from sale of crafts	21.7	23.1	p < 0.05
Percentage who taught crafts in last three years	29.4	43.3	p < 0.01
Percentage visiting art/ crafts ruseums in last 12 months	78.0	91 <b>.</b> 3	p < 0.01
Average number of crafts magazines read in last 12 months	1.6	2.4	N.A.
Average number of crafts organizations	. 1.2	1.5	N.A.
Percentage collecting crafts	59.4	68.4	p < 0.01
Number of respondents	90	2543	

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

Personal characteristics of crafts-artists also reveal some differences although generally not as large as those already presented. A lower proportion of crafts-artists who responded to the telephone component are female, slightly older, and to a much smaller extent, more likely to be married. Telephone respondents, compared with their mail counterparts, are more likely to be retired, less likely to have a college education, and are less concentrated among whites and those of Hispanic heritage or descent (Table B.2).

In spite of these statistically significant differences which demonstrate that those respondents who required a telephone call to achieve a completed questionnaire and those who did not are from diverse groups, the following should also be recognized:

- o The number of telephone respondents (90 out of a total of 2,637) was quite small. 96.6 percent of the response rate was achieved by mail and 3.4 percent by telephone follow-up.
- o The weighting procedures used recognized differences in response levels for media type and size of organization. This type of weighting tends to compensate for some of the accuracy lost through non-response because respondents from various strata are assigned weights, not because they responded or not, but based on the estimated number of member crafts-artists in each stratum.

Because of the weighting scheme used, the differences by media type in the telephone and non-telephone components are entirely corrected; on the other hand, to the extent differences that arise for other reasons—type of involvement in crafts, for example—these differences will not be adjusted.

TABLE B.2

COMPARISON OF PERSONAL CHARACTERISTICS BY MODE OF DATA COLLECTION: SURVEY OF MEMBER CRAFTS-ARTISTS, 1980

	Telephone Response	Mail Response	X <sup>2</sup> Statistic Significance Leve
· ·			<del></del>
Percentage Female	67.1	70.3	p < 0.01
Age Distribution	4		
Under 15	2.6	3.4	
25 to 34	19.9	25.0	
35 to 44	24.0	20.4	
45 to 54	15.3	18.0	
55 to 64	16.3	18.5	
65 and over	21.9	14.7	
Total	100.0	100.0	p < 0.01
Median age	47.3	45 • 7	
Percentage Married	72.9	71.7	
Labor Force status:	·		
Employed full time	44.9	45.6	
Employed part time	11.2	13.3	
Homemaker	14.0	19.1	
Student	0.9	2.9	
Retired	21.8	13.6	*
Unemployed	3.8	2.8	
Other	<u>3.4</u>	2.7	<b>₩</b> **
Total	100.0	100.0	p < 0.01
Percentage reporting crafts			0.40
as main occupation	34.0	33.4	p = 0.43
Percentage with college or			× 0. 01
higher education	36.6	56.6	p < 0.01
Percentage White	93.9	98.1	p < 0.01
Percentage H_spanic	0.2	1.3	p < 0.01
Number of respondents	90	2543	

SOURCE: Survey of Crafts-Artists, who are Members of Crafts Organizations, Mathematica Policy Research, Inc.

ERIC

Full Text Provided by ERIC

APPENDIX C

# COMPARISON OF AMERICAN AND GAMADIAN CRAFTS-ARTISTS

North Americans share many interests and characteristics, one of these interests is crafts. During the 1970's, the Canadian government sponsored several studies of crafts-artists in Canada. A large mail survey was conducted in 1976 of a sample of approximately 10,000 persons, drawn from mailing lists of all large crafts associations in all provinces except Quebec. The results, based on 2,400 mail responses, were published in Crafts and Craftsmen in Canada. 1 Because techniques were not used in the Canadian survey to achieve a high level of response (the response rate was 24 percent compared to 74 percent in the Survey of Crafts-Artists who are Members of Crafts Organizations in the United States), the results may not be representative of all Canadian member crafts-artists. Nevertheless, a comparison is made between the Canadian and U.S. survey results to see the extent of apparent similarities and differences between these two groups of crafts-artists who share the characteristic of belonging to crafts membership organizations. Many of the questions used in the two surveys are similar in wording, although there are often minor differences in the response categories.

Based on the two surveys, American crafts-artists are older than their Canadian counterparts, 45 years compared to 39.5 on the basis of median ages calculated from abbreviated age distributions (Table C.1).



Barry de Ville and Canadian Crafts Council, <u>Crafts and Crafts</u>
men in <u>Canada</u> (Ottawa, Canada: Research and Statistics Directorate,
Arts and Crafts Branch, Secretary of State, 1978).

Americans are also more highly educated, perhaps partly reflecting their age, but not entirely, since the major differences are the much larger share in the 45 and over age group most of whom would have finished their education 15 to 20 years ago. Another reflection of the 6-year difference in median ages is the 8-year difference in median years of involvement in crafts (Table C.2).

Canadian crafts-artists are more evenly distributed between crafts as a full-time occupation, part-time occupation and full-time crafts teacher than are American. About a fifth of respondents in both countries reported full-time involvement. A much larger proportion of U.S. respondents engage in crafts as a leisure or occasional activity compared to those in Canada. Students account for 2 and 7 percent of respondents, but while 4 percent of the U.S. respondents list their involvement as full-time crafts teachers, 25 percent of the respondents to the Canadian questionnaire are estimated by the report authors to be exclusively teachers of crafts (Table C.1).

As mentioned above, American crafts-artists have typically been working in their craft longer than Canadian crafts-artists—on average for over 16 years compared to under 9 years. About equal proportions of American and Canadian crafts-artists—68 percent and 70 percent—produce works for sale. Deflating gross sales figures for American crafts—

In both questionnaires, "leisure" is respondent-defined as is crafts.

## TABLE C.1

# COMPARISON OF U.S. AND CANADIAN CRAFTS-ARTISTS DEMOGRAPHIC CHARACTERISTICS, 1980 AND 1976

U.S. Characteristics			Canadian Characteristics
	Age Dis	tributions	
- AF	2 •	A .	16 42 25 2222
Under 25 years	3 %	9 %	16 to 25 years
25 to 34 years	25	34	26 to 35 years
35 to 44 years	20	20	36 to 45 years
45 and older	51	$\frac{37}{100}$	46 and older Total
Total	99 % 45.2	39.5	Median Age
Median age	43.4		median age
	Educa	tion	
Some high school	5 <b>%</b>	3 <b>%</b>	Primary or less
Completed high school	12	13	High school or less
Some college, or technical			
or vocational degree	21	27	Some post secondary
Two or four year			Completed university
college degree	31	41	or community college
Graduate work, masters			
or doctorate degree	31	15	Post graduate
Other	1		
Total	101 %	99 %	Total
<u> </u>	nvolvement	in Crafts	
	01 <b>T</b>	21 %	Full-time occupational
Major (primary) activity	24 %		
Secondary activity	16	36	Part-time occupation
Leisure or occasional	54	12	Leisure
activity	J4	14	WEYBOTE
Full-time student (of crafts) and "other"	2	7	Student
Full-time teacher	4	•	Exclusively teacher of
of crafts	4	25	crafts teacher or
Total	100 %	101 7	45 87 CB
TOLET	100 *	101 %	

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations, Mathematica Policy Research, Princeton, N.J., 1980, and Crafts and Craftsmen in Canada, Research and Statistics Directorate, Arts and Culture Branch, Secretary of State Department, Ottawa, Canada, 1978.

The text in <u>Crafts and Craftsmen in Canada</u> accompanying the distribution of type of involvement in crafts (Table 2.6, page 29) includes the following: "The full-time occupational group [28% in Table 2.6] includes roughly equal proportions of craftsmen and teacher-craftsmen. About 7 percent of this group are exclusively teachers of crafts. On the other hand, approximately 50 percent of the students [13 percent in Table 2.6] and leisure [23 percent in Table 2.6] groups also noted they were exclusively teachers of crafts . . . "

C.3



# TABLE C.2

# COMPARISON OF U.S. AND CANADIAN CRAFTS-ARTISTS CRAFT-RELATED CHARACTERISTICS, 1980 AND 1976

U.S. Chara	cteristics			Canadia	n Characteristics
		Experience i	n Crafts		
Years work	ing in crafts			Years b	een a craftsman
Less than	n 5 years	8 %	32 <b>%</b>	1 to	5 years
5 to 9 y		25	34	5 to	10 years
10 to 14	years	24	34	More	than 10 years
15 or mo		43			-
Total	•	100 %	100 %	Tot	al
Median	years	16.3	8.6	Med	ian years
	of crafts-	roduction of (	rafts for		age producing cra
	of crafts- producing works		rafts for		age producing cra
artists	of crafts- producing works		70 <b>%</b>	Percent	
artists	of crafts- producing works Gro ge (mean)	68 %	70 % aft Media n 1976	Percent for s	
artists for sale	of crafts- producing works Gro ge (mean)	68 % 	70 % aft Media n 1976	Percent for s	ale 
artists for sale  1980 avera gross cra	of crafts- producing works	68 % ss Sales by Cr U.S. Sales i Canadian doll	70 % aft Media n 1976	Percent for s	ess sales Clay
artists for sale  1980 avera gross cra  Clay Fiber	of crafts- producing works	68 % ss Sales by Cr U.S. Sales i Canadian doll \$3,025 760	70 % aft Media n 1976	Percent for s	ss sales Clay Fibre
artists for sale  1980 avera gross cra  Clay Fiber	of crafts- producing works	68 % ss Sales by Cr U.S. Sales i Canadian doll \$3,025 760 1,616	70 % aft Media n 1976	Percent for s 1976 gro \$4546 2524 4464	ss sales Clay Fibre
artists for sale	of crafts- producing works	68 % ss Sales by Cr U.S. Sales i Canadian doll \$3,025 760	70 % aft Media n 1976	Percent for s 1976 gro \$4546 2524 4464	ss sales  Clay Fibre Wood

SOURCE: Survey of Crafts-Artists who are Members of Crafts Organizations,
Mathematica Policy Research, Princeton, N.J., 1980 and <u>Crafts and</u>
<u>Craftsmen in Canada</u>, Research and Statistics Directorate, Arts and
Culture Branch, Secretary of State Department, Ottawa, Canada, 1978.

4,402

Based on a change in the all commodities consumer price index in the U.S. from 1976 to 1980 and 1976 exchange rates as reported in: U.S. Bureau of the Census, Statistical Abstract of the United States: 1980 (Washington, D.C., 1980) Tables 808 and 1602.

paper Glass or

other

5991

9532

Glass

artists from 1980 to 1976 dollars and accounting for the exchange rate, it appears that Canadian crafts-artists generate higher revenue from the sale of their work than do American crafts-artists. This appears true for every media category except metal (Table C.2).



### APPENDIX D

#### MODEL OF CRAFTS INVOLVEMENT

The fundamental premise of the behavioral model is that individuals attempt to maximize their utility or overall satisfaction, which is posited to be a function of leisure time and purchased goods and services. Crafts involvement is defined as hours of work effort which generates purchasing power, but implies foregoing leisure time. Because an extra hour of grafts work permits greater consumption of purchased goods and services, but reduces the time for leisure activities, individuals are faced with a negative trade-off between labor and leisure. Determination of the optimal level of work (or leisure) essentially balances these offsetting effects.

Theoretically, the optimal level of crafts work effort is influenced by the total remuneration offered to an individual—both pecuniary and nonpecuniary payments. But in practice, the money wage rate is used in most analyses of labor supply. Such a simplification, however, is a serious limitation for a study of crafts-artists since the latter are clearly motivated by nonpecuniary factors. In fact, there is some question as to whether or not a labor-leisure model is an appropriate paradigm for studying crafts-artists. Many such persons do not consider themselves as workers (self-employed or otherwise). Rather they consider their involvement as a leisure rather than as a work activity. Any income from sales might be considered incidental and use to defray the costs of pursuing their avocation. Furthermore, even for those who are serious about selling their products, the major satisfaction from

D.1

ERIC TO TENT PROVIDENCE OF THE PROVIDE OF THE PROVIDENCE OF THE PROVIDENCE OF THE PROVIDENCE OF THE PR

224

crafts appears to be artistic in nature rather than financial.  $\frac{1}{2}$ 

The above arguments are correct as they go, but there are several reasons to justify using a labor-leisure framework for the analysis.

First, despite the artistic motivation behind crafts involvement, 73 percent of the sample indicated that they produced works for sale (19 percent reported that crafts sales were their major income source). This suggests that a more accurate observation is that crafts activity yields both pecuniary and nonpecuniary returns, and that the mix between the two is partially subject to the discretion or preferences of the individual.

The above ideas can be highlighted more clearly with the development of a simple model that incorporates the notion of full remuneration (see Black, 1978; Borjas, 1981; and Robins, 1977). It is assumed that an individual's utility (U) is an indirect function of leisure time (L) and market goods purchased out of income (i.e., purchasing power, Y).

Leisure time and purchased goods are combined inputs into producing leisure-time-comsumption (LTC) which directly affects utility. An individual may also enjoy on-the-job consumption (i.e., job satisfaction, OJC) which affects utility directly. OJC is assumed to be a product of work hours (H) and an index of nonpecuniary job attributes (C). Thus, total utility is defined as:

$$\mathbf{U} = \mathbf{U}(\mathbf{LTC}) + \mathbf{U}(\mathbf{OJC}) \tag{1}$$

$$U = U(L,Y) + U(CH)$$
 (1)

Less than 10 percent of the surveyed crafts-artists reported that the chief satisfaction gained from crafts was financial. The satisfactions most often cited as primary were crafts "as a means of creative expression" (52 percent), "as a sense of accomplishment" (21 percent), "as one's life's work" (9 percent), and "as a diversion from the daily routine" (9 percent).

where one assumes that utility can be split into two additive components.

In the context of crafts, H represents the number of weekly hours spent producing and marketing crafts and C is the hourly flow of non-pecuniary benefits gained from crafts involvement. For simplicity, L denotes all noncraft time, leisure as well as noncraft employment. The amount of goods that can be purchased is limited by a budget constraint that defines as individual's income:

$$Y = WH + Yn \tag{2}$$

where W is an hourly money wage rate and Yn denotes other noncraft income available to the individual. Work hours is limited by total time (T) that must be split between work and leisure, T = L + H. An increase in work hours thus affects utility by decreasing leisure time, L = T - H, by increasing purchasing power, WH, and by increasing on-the-job consumption, CH.

Another concept to introduce is that, to some extent, a craftsartist is likely to reduce the nonpecuniary returns to producing crafts
as his or her activity becomes more oriented toward sales, thereby
increasing W at the expense of CH (e.g., the work may become more repetitious, more intense, and marketing efforts may displace time spent in
producing crafts). This suggests a relationship between total hourly
renumeration (W\*) and its two components:

$$W^* = W + PC. \tag{3}$$

 $<sup>\</sup>frac{1}{\text{We}}$  abstract from the problem of simultaneity that exists between hours supplied to crafts versus other labor activities. A complete model would focus on the joint determination of crafts-related work and other employment decisions.

The price of nonpecuniary job benefits (P) is assumed to be fixed and it represents the income foregone as crafts activity shifts away from a sales orientation toward an avocational focus. Equation (3) is consistent with a positive association between W\* and human capital. It canalso be used to estimate the market value of various nonwage job characteristics. Note that multiplication of (3) by work hours yields the income constraint.

Given that Y, C, H are controllable by the individual and W\*, P, Yn are exogenous, the first-order conditions are maximizing utility under the budget constraint (Y = W\*H - PCH + Yn) are:

$$\frac{\partial U}{Y} - \lambda = 0 \tag{4}$$

$$\frac{\partial U}{\partial Y} \cdot W - \frac{\partial U}{\partial L} \frac{\partial U}{\partial OJC} \cdot C + (W-W* + PC) g = 0$$

$$\frac{\partial U}{\partial V} \cdot H - (PH) \lambda = 0.$$
(6)

$$\frac{\partial U}{\partial OJC} \cdot H - (PH) \lambda = 0. \tag{6}$$

These conditions imply (assuming second-order conditions are fulfilled) that H and C should be adjusted until:

$$MU_{Y} = \frac{MU_{OJC}}{P} = \frac{MU_{L}}{W*}$$
 (7)

where the marginal utilities (MU's) are the respective partial derivatives in (4-6). The optimizing condition in (7) can be re-expressed as:

$$\frac{MU_{L} - CMU_{OJC}}{MU_{Y}} = W.$$
 (8)

The conditions stated in (7) and (8) can be contrasted with those when OJC is ignored in the labor-leisure model:

$$\frac{MU_{L}}{MU_{Y}} = W. \tag{8}$$



227

The relevance of OJC in labor supply decisions is evident in (7) where both C and H must be adjusted until the ratios of their marginal utilities to their respective prices are in equality with MU<sub>y</sub>. The shadow price of leisure is now W\* which supports the general notion that total labor compensation is what theoretically affects behavior, not W by itself. It is also apparent that W is an endogenous variable that is jointly determined with C. If work produces no on-the-job consumption (MU<sub>OJC</sub> = 0), then equation (8) reduces to (8)°. If, on the other hand, work time affects utility because of nonpecuniary factors (rather than only indirectly through earnings), then the wage rate that is acceptable to a worker will be less to the extent that a job is pleasurable. Alternatively, a pleasurable job will induce greater work effort for a given wage rate than a similarly paid job with less attractive nonwage elements. 1/

Based on the first-order conditions, the supply of work hours can be specified as a general function:

$$H = f(W*, P, Y_n; X)$$

(9)

where X is a vector of personal and environmental characteristics. For a given set of X attributes, hours of crafts activity is a positive function of the potential returns of participation (i.e., the incentive effect, W\*), and a negative function of P, the rate at which nonpecuniary job benefits must be foregone as monetary benefits are increased (see equation 3). The link between hours and other individual or household income, Yn, is ambiguous. On the one hand, a relatively high Yn would be associated with less crafts activity if it were viewed as a work activity (i.e., an income effect). On the other hand, a relatively high Yn might induce greater crafts involvement if it were viewed as a leisure activity (i.e., a "normal" consumption good).

 $<sup>\</sup>frac{1}{A}$ s noted by Borjas (1981), observed wage rates will understate the value of leisure time when there is nonpecuniary remuneration.

## REFERENCES

- Black, Matther. "Job Satisfaction of Urban Blacks." Faper delivered at the Southern Economic Association Meetings. November 1978.
- Borjas, George J. "Nonpecuniary Job Benefits: A Theoretical and Emprical Survey." Report prepared for the Department of Commerce. February 1981.
- Duncan, Greg J. "Earnings Functions and Nonpecuniary Benefits." <u>Journal</u> of <u>Human Resources</u>. 1976
- Robins, Philip K. "Job Satisfaction, Labor Market Mobility, and Income Maintenance: Evidence from the Seattle and Denver Income Maintenance Experiments." SIME/DIME Research Memorandum Draft. SRI, 1977.